

City of Chicago



O2021-3265

Office of the City Clerk

Document Tracking Sheet

Meeting Date: 7/21/2021

Sponsor(s): Lightfoot (Mayor)

Type: Ordinance

Title: Sale of City-owned properties to and execution of

redevelopment agreement using multi-family program funds and tax incremental financing (TIF) with 43rd and Calumet Phase I LP, 43rd and Calumet Phase I LLC, Habitat 43 Green LLC, and P3 Markets 43 Green LLC for construction of mixed income, mixed use residential complex at 312 E 43rd St, 4310 S Calumet Ave, 4312 S Calumet Ave and

4314 S Calumet Ave

Committee(s) Assignment: Committee on Finance



OFFICE OF THE MAYOR

CITY OF CHICAGO

LORI E. LIGHTFOOT
MAYOR

July 21, 2021

TO THE HONORABLE, THE CITY COUNCIL OF THE CITY OF CHICAGO

Ladies and Gentlemen:

At the request of the Commissioner of Housing, I transmit herewith an ordinance authorizing the issuance of financial assistance to Calumet Phase I LP for affordable housing development.

Your favorable consideration of this ordinance will be appreciated.

Very truly yours

Mavor

ORDINANCE

WHEREAS, the City of Chicago (the "City"), a home rule unit of government under Section 6(a), Article VII of the 1970 Constitution of the State of Illinois, has heretofore found and does hereby find that there exists within the City a serious shortage of decent, safe and sanitary rental housing available to persons of low and moderate income; and

WHEREAS, the City has determined that the continuance of a shortage of affordable rental housing is harmful to the health, prosperity, economic stability and general welfare of the City; and

WHEREAS, the City has certain funds available from a variety of funding sources ("Multi-Family Program Funds") to make loans and grants for the development of multi-family residential housing to increase the number of families served with decent, safe, sanitary and affordable housing and to expand the long-term supply of affordable housing, and such Multi-Family Program Funds are administered by the City's Department of Housing ("DOH"); and

WHEREAS, DOH has preliminarily reviewed and approved the making of a loan to the 43rd and Calumet Phase I LP, a Delaware limited partnership (the "Borrower" or "43 Green"), whose sole general partner is 43rd and Calumet Phase I LLC, an Illinois limited liability company (the "General Partner"), the members of which will be (i) Habitat 43 Green LLC, an Illinois limited liability company (the "Managing Member" or "Habitat 43") and (ii) P3 Markets 43 Green LLC, an Illinois limited liability company ("GP Member or "P3 Markets"), in an amount not to exceed \$3,000,000 (the "Loan"), to be funded from Multi-Family Program Funds pursuant to the terms and conditions set forth in Exhibit A attached hereto and made a part hereof; and

WHEREAS, the members of Managing Member will be Habitat Acquisitions Company LLC, an Illinois limited liability company ("Acquisitions") and Generations Housing Initiatives, an Illinois not-for-profit corporation ("GHI"); and

WHEREAS, upon the selection of other parties to be the limited partners of Borrower upon the closing of the Loan, pursuant to the terms of Borrower's partnership agreement, said limited partners may have the right to replace General Partner as the sole general partner of Borrower pursuant to certain defaults or other events as may be approved by the City; and

WHEREAS, the City has established the Community Development Commission ("CDC") to, among other things, designate redevelopment areas, approve redevelopment plans, recommend the sale of parcels located in redevelopment areas, and designate the developers of certain redevelopment projects, all subject to the approval of the City Council of the City ("City Council"); and

WHEREAS, under ordinances adopted on May 29, 2002, and published in the Journal of the Proceedings of the City Council (the "Journal") for such date at pages 85676 to 85904 (as amended by ordinances adopted on: (A) November 13, 2013 and published on pages 63293 to 63297 of the Journal for such date and (B) May 28, 2014 and published on pages 80952 to 80956 of the Journal for such date) and under provisions of the Tax Increment Allocation Redevelopment Act, 65 ILCS 5/11-74.4.1 et. seq (as amended, the "Act"), the City Council (i) approved a certain redevelopment plan and project (the "Redevelopment Plan") for a portion of

the City known as the "47th and King Drive Redevelopment Project Area" (the "**Redevelopment Area**"); (ii) designated the Redevelopment Area as a "redevelopment project area" within the requirements of the Act and (iii) adopted tax increment financing for the Redevelopment Area (the "**Adoption Ordinance**"); and

WHEREAS, the City is the owner of four vacant parcels of land (collectively, the "Property"), located at the SW corner of 43rd Street and Calumet Avenue, in Chicago, Illinois 60624, as legally described on <u>Exhibit B</u> attached hereto, which includes the common addresses of 4310—14 Calumet Avenue and 321 E. 43rd Street; and

WHEREAS, the Property is located within the Redevelopment Area; and

WHEREAS, the initial appraised value of the Property, as cited in the CDC Resolution (as defined below), was approximately \$566,698, based upon an opinion of value stated in a market study dated May 10, 2021; and

WHEREAS, the Borrower has submitted a proposal to DOH to acquire the Property for \$1 per parcel in order to construct the Project (as hereinafter defined); and

WHEREAS, the Borrower intends to redevelop the Property by constructing a building that will contain ninety-nine (99) residential units that will include studio units, one-bedroom units and two—bedroom units and associated amenities such as on-site laundry, fitness center and outdoor terraces, along with approximately 5,000 square feet of ground floor commercial space (as more particularly described in Exhibit A hereto, the "Project"); and

WHEREAS, the Project is consistent with the purposes and objectives of the Redevelopment Plan; and

WHEREAS, the Borrower and 43 Green JV LLC, a Delaware limited liability company ("**JV**") in association with GHI and P3 Markets (collectively the "**Developer**") have agreed to undertake the Project and redevelop the Property in accordance with the Redevelopment Plan and pursuant to the terms and conditions of a redevelopment agreement in substantially the form attached hereto as **Exhibit C** (the "**Redevelopment Agreement**"); and

WHEREAS, the redevelopment of the Property, including but not limited to the construction of the Project's facilities, will be financed in part by incremental taxes deposited in the 47th/King Drive Redevelopment Project Area Special Tax Allocation Fund (as defined in the Adoption Ordinance) pursuant to Section 5/11-74.4-8(b) of the Act; and

WHEREAS, Borrower and JV are each affiliates of Habitat Company, an Illinois limited liability company ("Habitat") and P3 Markets; and

WHEREAS, pursuant to Resolution 21-CDC-21 adopted on June 8, 2021 (the "CDC Resolution"), the CDC authorized DOH to advertise the City's intent to negotiate the sale and redevelopment of the Property with Habitat, together with its affiliates, and P3 Markets together with its affiliates (collectively, the "**Joint Venture**") and to request alternative proposals, and recommended the sale of the Property to the Joint Venture and the designation of the Joint Venture as the developer of the Property if no responsive alternative proposals were received at the conclusion of the advertising period, or, if alternative proposals were received, if DOH determined in its sole discretion that the Joint Venture's project was the best proposal; and

WHEREAS, public notices advertising DOH's intent to enter into a negotiated sale of the Property with the Developer and requesting alternative proposals appeared in the *Chicago Tribune* on June 11, June 18, June 25, and July 2, 2021; and

WHEREAS, upon completion of a final appraisal dated May 21, 2021, the Property was determined to have an appraised value of approximately \$815,565; and

WHEREAS, no other responsive proposals were received by the deadline set forth in the aforesaid notices; *now, therefore,*

BE IT ORDAINED BY THE CITY COUNCIL OF THE CITY OF CHICAGO:

SECTION 1. The above recitals are expressly incorporated in and made a part of this ordinance as though fully set forth herein.

SECTION 2. Upon the approval and availability of the Additional Financing as shown in **Exhibit A** hereto, the Commissioner of DOH (the "**Commissioner**") and a designee of the Commissioner (collectively, the "**Authorized Officer**") are each hereby authorized, subject to approval by the Corporation Counsel, to enter into and execute such agreements and instruments, and perform any and all acts as shall be necessary or advisable in connection with the implementation of the Loan. The Authorized Officer is hereby authorized, subject to the approval of the Corporation Counsel, to negotiate any and all terms and provisions in connection with the Loan which do not substantially modify the terms described in **Exhibit A** hereto. Upon the execution and receipt of proper documentation, the Authorized Officer is hereby authorized to disburse the proceeds of the Loan to the Borrower.

SECTION 3. Developer is hereby designated as the developer for the Project under Section 5/11-74.4-4 of the Act.

SECTION 4. The sale of the Property to the Developer for \$1.00 per parcel is hereby approved. This approval is expressly conditioned upon the City entering into the Redevelopment Agreement with the Developer. The Authorized Officer is hereby authorized, with the approval of the City's Corporation Counsel as to form and legality, to negotiate, execute and deliver the Redevelopment Agreement, and such other supporting documents as may be necessary or appropriate to carry out and comply with the provisions of the Redevelopment Agreement, with such changes, deletions and insertions as shall be approved by the Authorized Officer.

SECTION 5. The Mayor or her proxy is authorized to execute, and the City Clerk or the Deputy City Clerk is authorized to attest, one or more quitclaim deeds conveying the Property to the Developer, subject to those covenants, conditions and restrictions set forth in the Redevelopment Agreement.

SECTION 6. To the extent that any ordinance, resolution, rule, order or provision of the Municipal Code of Chicago, or part thereof, is in conflict with the provisions of this ordinance, the provisions of this ordinance shall control. If any section, paragraph, clause or provision of this ordinance shall be held invalid, the invalidity of such section, paragraph, clause or provision shall not affect any of the other provisions of this ordinance. Section 2-44-080 of the Municipal Code of Chicago shall not apply to the Project or the Property.

approval.

This ordinance shall be effective as of the date of its passage and

EXHIBIT A

BORROWER:

43rd and Calumet Phase I Limited Partnership, a Delaware limited partnership (the "Borrower"), whose sole general partner is 43rd and Calumet Phase I LLC, an Illinois limited liability company (the "General Partner"), the members of which General Partner will be (i) Habitat 43 Green LLC, an Illinois limited liability company (the "Managing Member") and (ii) P3 Markets 43 Green LLC, an Illinois limited liability company ("GP Member"), and others to be hereafter selected as the limited partners. The members of Managing Member will be Habitat Acquisitions Company LLC ("Acquisitions") and Generations Housing Initiatives, an Illinois not-for-profit corporation ("GHI").

PROJECT:

Construction of a building at the southwest corner of 43rd Street and Calumet Avenue, Chicago IL 60653 with an aggregate of approximately 99 residential dwelling units containing studio, one- and two- bedroom units, approximately 51 of which are reserved for low- and moderate-income persons, together with certain common space, offices, and parking.

LOAN:

Source: Multi-Family Program Funds. These funds may be

disbursed directly to Borrower or to GHI, and then

loaned or contributed to Borrower by GHI.

Amount:

Not to exceed \$3,000,000 Not to exceed 30 years

Term: Interest:

Zero percent per annum, or another interest rate

acceptable to the Authorized Officer

Security:

1.

Non-recourse loan; second mortgage on the Property (the "City Mortgage")

ADDITIONAL FINANCING:

Amount: Not to exceed \$9,600,000 (the "Permanent Loan")

Term: Not to exceed 30 months initially and 40 years after

conversion to permanent loan, or another term

acceptable to the Authorized Officer

Source: Bellwether Enterprise, or another entity acceptable

to the Authorized Officer.

Interest: A fixed rate of interest with a maximum interest rate

not to exceed 4% or the maximum which may be paid under Illinois State Law or another rate or

rates acceptable to the Authorized Officer.

Security: A mortgage on the Property senior to the lien of the

City Mortgage, a pledge of capital contributions and

general partner interests, and a pledge of the

Borrower interests in the Redevelopment Agreement, or such other security as may be

acceptable to the Authorized Officer.

2. Amount: Proceeds: Not to exceed \$1,500,000 (the "DCEO Grant") Either funded directly to Borrower or to GHI, and

then loaned or contributed to Borrower by GHI.

Source:

Grant from the State of Illinois Dept. of Commerce and Economic Opportunity ("DCEO"), or another

entity acceptable to the Authorized Officer.

3. Amount: Approximately \$346,615 (the "IAHTC Equity")

Proceeds: These funds shall be paid in upon Borrower's purchase of the City Property at Closing. These funds may be paid to Borrower or to GHI, and then

loaned or contributed to Borrower by GHI.

Source:

Equity to be derived from the syndication of an annual allocation of Illinois Affordable Housing Tax Credits ("IAHTC Equity") based on the purchase of City Land, fifty percent of which or \$407,783 is viewed as a donation sold as IAHTC Donation Tax Credits to generate the IAHTC Equity.

4. Amount

Approximately \$18,328,167

Proceeds:

All or a portion of these funds may be (i) paid in on a delayed basis and (ii) used to retire a portion of the Construction Loan or the LIHTC Equity Bridge

Loan

Source:

Equity to be derived from the syndication of an annual Low-Income Housing Tax Credit ("LIHTC") allocation of approximately \$1,950,000 by the City.

5. Amount: Not to exceed \$5,000,000

Proceeds:

Funds will be made available during Construction Period either directly to Borrower or to GHI, and then loaned or contributed to Borrower by

GHI.

Source:

Proceeds from a grant of Tax Increment Financing, all of which will be disbursed during three periodic

payments during the construction period.

6. Amount:

Source:

Approximately \$530,000 Deferred Developer Fee

7. Amount. \$10,100

Source:

General Partner

EXHIBIT B

LEGAL DESCRIPTION OF PROPERTY

(SUBJECT TO FINAL TITLE AND SURVEY)

Lots 1 through 5, both inclusive, in the Subdivision of Lots 25 to 28 inclusive in Honore's Subdivision of part of the Northeast 1/4 of the Northeast 1/4 of the Southwest 1/4 of Section 3, Township 38 North, Range 14, East of the Third Principal Meridian, in Cook County, Illinois.

Also

Lots 29 through 32, both inclusive, in H. H. Honore's Subdivision of part of the Northeast 1/4 of the Northeast 1/4 of the Southwest 1/4 of Section 3, Township 38 North, Range 14, East of the Third Principal Meridian, in Cook County, Illinois.

PINS: 20-03-304-024-0000

20-03-304-025-0000 20-03-304-026-0000 20-03-304-044-0000

Addresses: 321 E. 43rd Street

4310-14 South Calumet Avenue

Chicago, Illinois 60653

EXHIBIT C

FORM OF REDEVELOPMENT AGREEMENT (SEE ATTACHED)

[leave blank 3" x 5" space for recorder's office]

This agreement was prepared by and after recording return to:
Randall Johnson, Esq.
Senior Assistant Corporation Counsel
City of Chicago Department of Law
121 North LaSalle Street, Room 600
Chicago, IL 60602

43 GREEN—PHASE I REDEVELOPMENT AGREEMENT

This 43 Green—Phase I Redevelopment Agreement (this "Agreement") is made as of this _____ day of _____, 2021, by and between the City of Chicago, an Illinois municipal corporation (the "City"), through its Department of Planning and Development ('DPD"), 43rd and Calumet Phase I LP, a Delaware Limited Partnership (the "Developer"), 43 Green JV, LLC, a Delaware limited liability company ("GP Manager") and Generations Housing Initiatives, an Illinois not-for-profit ("Generations" or the "Habitat NFP"). The Developer, GP Manager and Generations may collectively be referred to hereinafter as the "Developer Parties".

RECITALS

- A. <u>Constitutional Authority</u>: As a home rule unit of government under Section 6(a), Article VII of the 1970 Constitution of the State of Illinois (the "State"), the City has the power to regulate for the protection of the public health, safety, morals and welfare of its inhabitants, and pursuant thereto, has the power to encourage private development in order to enhance the local tax base, create employment opportunities and to enter into contractual agreements with private parties in order to achieve these goals.
- B. <u>Statutory Authority</u>: The City is authorized under the provisions of the <u>Tax Increment Allocation Redevelopment Act</u>, 65 ILCS 5/11-74.4-1 <u>et seq.</u>, as amended from time to time (the "**Act**"), to finance projects that eradicate blighted conditions and conservation area factors through the use of tax increment allocation financing for redevelopment projects.

- C. <u>City Council Authority</u>: To induce redevelopment pursuant to the Act, the City Council of the City (the "City Council") adopted the following ordinances on March 27, 2002: (1) "An Ordinance of the City of Chicago, Illinois Approving a Redevelopment Plan for the 47th and King Drive Redevelopment Project Area" (the "Plan Adoption Ordinance"); (2) "An Ordinance of the City of Chicago, Illinois Designating the 47th and King Drive Redevelopment Project Area Redevelopment Project Area as a Redevelopment Project Area Pursuant to the Tax Increment Allocation Redevelopment Act"; and (3) "An Ordinance of the City of Chicago, Illinois Adopting Tax Increment Allocation Financing for the "47th and King Drive Redevelopment Project Area Redevelopment Project Area" (the "TIF Adoption Ordinance") (items(1)-(3) collectively referred to herein as the "TIF Ordinances"). The redevelopment project area referred to above (the "Redevelopment Area") is legally described in <u>Exhibit A</u> hereto.
- The Project: Developer intends to purchase (the "Acquisition") certain property located within the Redevelopment Area at the southwest corner of 43rd Street and Calumet Avenue with the following common addresses: 312 East 43rd Street, 4310 South Calumet Avenue, 4312 South Calumet Avenue and 4314 South Calumet Avenue, Chicago, Illinois 60653 and legally described on Exhibit B hereto (the "Property"). All of the Property is currently owned by the City. Developer, within the time frames set forth in Section 3.01 hereof, shall commence and complete construction of an approximately ten (10) story precast concrete building that will contain a total of approximately 97,292 square feet and include (i) approximately ninety-nine (99) studio, one, and two bedroom residential apartment dwellings for market rate residents and those earning up to fifty percent (50%) and sixty percent (60%) of area median income (AMI) (60%); (ii) approximately fivethousand (5,000) square feet of ground floor retail space and (iii) related residential amenities such as (a.) fitness center, (b.) community room with kitchen, (c.) residential lobby, (d.) outdoor terrace, (e.) outdoor green space, (f.) 24 surface parking spaces and, (g.) bike and tenant storage, and (h.) laundry (the "Facility") thereon. The Facility and related improvements (including but not limited to those TIF-Funded Improvements as defined below and set forth on Exhibit C) are collectively referred to herein as the "Project." The completion of the Project would not reasonably be anticipated without the financing contemplated in this Agreement.
- E. <u>Redevelopment Plan</u>: The Project will be carried out in accordance with this Agreement and the City of Chicago 47th/King Drive Redevelopment Project Area Tax Increment Finance Program Redevelopment Plan and Project" (the "**Redevelopment Plan**") included in the Plan Adoption Ordinance and published with the Plan Adoption Ordinance at pages 81231-81457 of the Journal of the Proceedings of the City Council.
- F. <u>City Financing</u>: The City agrees to use, in the amounts set forth in <u>Section 4.03</u> hereof, Incremental Taxes (as defined below), to pay for or reimburse Developer for the costs of TIF-Funded Improvements pursuant to the terms and conditions of this Agreement.

Now, therefore, in consideration of the mutual covenants and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

SECTION 1. RECITALS, HEADINGS AND EXHIBITS

The foregoing recitals are hereby incorporated into this Agreement by reference. The paragraph and section headings contained in this Agreement, including without limitation those set forth in the following table of contents, are for convenience only and are not intended to limit, vary.

define or expand the content thereof. Developer agrees to comply with the requirements set forth in the following exhibits which are attached to and made a part of this Agreement. All provisions listed in the Exhibits have the same force and effect as if they had been listed in the body of this Agreement.

Table of Contents	List of Exhibits
Recitals, Headings and Exhibits	A *Redevelopment Area
2. Definitions	B *Property
3. The Project	C *TIF-Funded Improvements
4. Financing	D [intentionally omitted]
5. Conditions Precedent	E Construction Contract
6. Agreements with Contractors	F Escrow Agreement
7. Completion of Construction or Rehabilitation	G *Permitted Liens
8. Covenants/Representations/Warranties of	H-1 *Project Budget
Developer	H-2 *MBE/WBE Budget
9. Covenants/Representations/Warranties of the	I Approved Prior Expenditures
City	J Opinion of Developer's Counsel
10. Developer's Employment Obligations	K *Preliminary TIF Projection Real
11. Environmental Matters	Estate Taxes
12. Insurance	L Requisition Form [use if no escrow]
13. Indemnification	M *[intentionally omitted]
14. Maintaining Records/Right to Inspect	N Form of Subordination Agreement
15. Defaults and Remedies	O Form of Payment Bond
16. Mortgaging of the Project	P Investor Letter
17. Notice	
18. Miscellaneous	(An asterisk (*) indicates which exhibits are to be recorded.)

SECTION 2. DEFINITIONS

For purposes of this Agreement, in addition to the terms defined in the foregoing recitals, the following terms shall have the meanings set forth below:

"Act" shall have the meaning set forth in the Recitals hereof.

"Acquisition" shall have the meaning set forth in the Recitals hereof.

"Affiliate" shall mean any person or entity directly or indirectly controlling, controlled by or under common control with Developer.

"AIS" shall have the meaning set forth in Section 3.13 hereof.

"Annual Compliance Report" shall mean a signed report from Developer to the City (a) itemizing each of Developer's obligations under the RDA during the preceding calendar year, (b) certifying Developer's compliance or noncompliance with such obligations, (c) attaching evidence (whether or not previously submitted to the City) of such compliance or noncompliance and (d) certifying that Developer is not in default with respect to any provision of the RDA, the agreements

evidencing the Lender Financing, if any, or any related agreements; provided, that the obligations to be covered by the Annual Compliance Report shall include the following: (1) compliance with the Operating Covenant (Section 8.06); (2) compliance with the Jobs Covenant (Section 8.06); (3) delivery of Financial Statements and unaudited financial statements (Section 8.13); (4) delivery of updated insurance certificates, if applicable (Section 8.14); (5) delivery of evidence of payment of Non-Governmental Charges, if applicable (Section 8.15); [(6) delivery of a substitute Letter of Credit, if applicable (Section 8.1 1;) and (7) compliance with all other executory provisions of the RDA. [NOTE: may add to this list any post-Certificate obligations unique to this RDA]

"<u>Available Project Funds</u>" shall have the meaning set forth for such term in <u>Section 4.07</u> hereof.

"Bond(s)" shall have the meaning set forth for such term in Section 8.05 hereof.

"<u>Certificate</u>" shall mean the Certificate of Completion of Construction described in <u>Section</u> 7.01 hereof.

"<u>Change Order</u>" shall mean any amendment or modification to the Scope Drawings, Plans and Specifications or the Project Budget as described in <u>Section 3.03</u>, <u>Section 3.04</u> and <u>Section 3.05</u>, respectively.

"City Contract" shall have the meaning set forth in Section 8.01(I) hereof.

"City Council" shall have the meaning set forth in the Recitals hereof.

"City Funds" shall mean the funds described in Section 4.03(b) hereof.

"Closing Date" shall mean the date of execution and delivery of this Agreement by all parties hereto, which shall be deemed to be the date appearing in the first paragraph of this Agreement.

"Consultant's Report" shall have the meaning set forth in Section 8.27(a) hereof.

"Contract" shall have the meaning set forth in Section 10.03 hereof.

"Contractor" shall have the meaning set forth in Section 10.03 hereof.

"Construction Contract" shall mean that certain contract, substantially in the form attached hereto as Exhibit E, to be entered into between Developer and the General Contractor providing for construction of the Project.

"Corporation Counsel" shall mean the City's Department of Law.

"Deed" shall have the meaning set forth in Section 3.13(a) hereof.

"<u>Developer Parties</u>" shall mean the Developer, its Affiliates, and the respective officers, directors, trustees, employees, agents, successors and assigns of the Developer and its Affiliates.

"<u>DTC Equity</u>" approximately [[[\$346,615 or _____]]] to be derived from the syndication of approximately \$350,000 in Illinois Affordable Housing Tax Credits.

"<u>EDS</u>" shall mean the City's Economic Disclosure Statement and Affidavit, on the City's then-current form, whether submitted in paper or via the City's online submission process.

"Employer(s)" shall have the meaning set forth in Section 10 hereof.

"Employment Plan" shall have the meaning set forth in Section 5.12 hereof.

"EMMA" shall have the meaning set forth in Section 8.27(c) hereof.

"Environmental Documents" shall mean all reports, surveys, field data, correspondence and analytical results prepared by or for the Foundation (or otherwise obtained by the Foundation) regarding the condition of the Property or any portion thereof, including, without limitation, the SRP Documents.

"Environmental Laws" shall mean any and all federal, state or local statutes, laws, regulations, ordinances, codes, rules, orders, licenses, judgments, decrees or requirements relating to public health and safety and the environment now or hereafter in force, as amended and hereafter amended, including but not limited to (i) the Comprehensive Environmental Response, Compensation and Liability Act (42 U.S.C. Section 9601 et seq.); (ii) any so-called "Superfund" or "Superlien" law; (iii) the Hazardous Materials Transportation Act (49 U.S.C. Section 1802 et seq.); (iv) the Resource Conservation and Recovery Act (42 U.S.C. Section 6902 et seq.); (v) the Clean Air Act (42 U.S.C. Section 7401 et seq.); (vi) the Clean Water Act (33 U.S.C. Section 1251 et seq.); (vii) the Toxic Substances Control Act (15 U.S.C. Section 2601 et seq.); (viii) the Federal Insecticide, Fungicide and Rodenticide Act (7 U.S.C. Section 136 et seq.); (ix) the Illinois Environmental Protection Act (415 ILCS 5/1 et seq.); and (x) the Municipal Code.

"Equity" shall mean funds of Developer (other than funds derived from Lender Financing) irrevocably available for the Project, in the amount set forth in Section 4.01 hereof, which amount may be increased pursuant to Section 4.06 (Cost Overruns) or [Section 4.03(b).]

"Escrow" shall mean the construction escrow established pursuant to the Escrow Agreement.

"Escrow Agreement" shall mean the Escrow Agreement establishing a construction escrow, to be entered into as of the date hereof by [the City,] the Title Company (or an affiliate of the Title Company), Developer and Developer's lender(s), substantially in the form of Exhibit F attached hereto.

"Event of Default" shall have the meaning set forth in Section 15 hereof.

"Facility" shall have the meaning set forth in the Recitals hereof.

"<u>Financial Interest</u>" shall have the meaning set forth for such term in Section 2-156-010 of the Municipal Code.

"<u>Financial Statements</u>" shall mean complete audited financial statements of Developer prepared by a certified public accountant in accordance with generally accepted accounting principles and practices consistently applied throughout the appropriate periods.

"Final NFR Letter" shall mean a final comprehensive (if applicable) "No Further Remediation" letter issued by the IEPA approving the use of the Property for the construction, development and operation of the Project in accordance with a site plan approved by the City and the terms and conditions of the SRP Documents, as amended or supplemented from time to time. The Final NFR Letter shall state that the Property meets TACO Tier 1 remediation objectives (residential or commercial as applicable), and the construction worker exposure route as set forth in 35 Ill. Adm. Code Part 742, but may be reasonably conditioned upon use and maintenance of engineered barriers and other institutional or engineering controls acceptable to the IEPA.

"<u>General Contractor</u>" shall mean the general contractor(s) hired by Developer pursuant to Section 6.01.

"<u>Hazardous Materials</u>" shall mean any toxic substance, hazardous substance, hazardous material, hazardous chemical or hazardous, toxic or dangerous waste defined or qualifying as such in (or for the purposes of) any Environmental Law, or any pollutant or contaminant, and shall include, but not be limited to, petroleum (including crude oil), any radioactive material or by-product material, polychlorinated biphenyls and asbestos in any form or condition.

"Human Rights Ordinance" shall have the meaning set forth in Section 10 hereof.

"In Balance" shall have the meaning set forth in Section 4.07 hereof.

"Incremental Taxes" shall mean such ad valorem taxes which, pursuant to the TIF Adoption Ordinance and Section 5/11-74.4-8(b) of the Act, are allocated to and when collected are paid to the Treasurer of the City of Chicago for deposit by the Treasurer into the TIF Fund established to pay Redevelopment Project Costs and obligations incurred in the payment thereof.

"IEPA" shall mean the Illinois Environmental Protection Agency.

"Indemnitee" and "Indemnitees" shall have the meanings set forth in Section 13.01 hereof.

"Laws" shall mean all applicable federal, state, local or other laws (including common law), statutes, codes, ordinances, rules, regulations or other requirements, now or hereafter in effect, as amended or supplemented from time to time, and any applicable judicial or administrative interpretation thereof, including any applicable judicial or administrative orders, consent decrees or judgments, including, without limitation, Sections 7-28 and 11-4 of the Municipal Code relating to waste disposal.

"<u>Lender Financing</u>" shall mean funds borrowed by Developer from lenders and irrevocably available to pay for Costs of the Project, in the amount set forth in <u>Section 4.01</u> hereof.

"Losses," as used in <u>Section 3.13</u> hereof, shall mean any and all debts, liens, claims, causes of action, demands, complaints, legal or administrative proceedings, losses, damages, obligations, liabilities, judgments, amounts paid in settlement, arbitration or mediation awards, interest, fines, penalties, costs and expenses (including, without limitation, reasonable attorney's fees and expenses, consultants' fees and expenses and court costs).

"MBE(s)" shall mean a business identified in the Directory of Certified Minority Business Enterprises published by the City's Department of Procurement Services, or otherwise certified by

the City's Department of Procurement Services as a minority-owned business enterprise, related to the Procurement Program or the Construction Program, as applicable.

"MBE/WBE Budget" shall mean the budget attached hereto as Exhibit H-2, as described in Section 10.03.

"MBE/WBE Program" shall have the meaning set forth in Section 10.03 hereof.

"MSRB" shall have the meaning set forth in Section 8..27(c) hereof

"<u>Municipal Code</u>" shall mean the Municipal Code of the City of Chicago, as amended from time to time.

"New Mortgage" shall have the meaning set forth in Article 16 hereof.

"<u>Non-Governmental Charges</u>" shall mean all non-governmental charges, liens, claims, or encumbrances relating to Developer, the Property or the Project.

"OBM" shall have the meaning set forth in Section 8.27(c) hereof

"Phase I ESA" shall have the meaning set forth in Section 3.13(e) hereof.

"Phase II ESA" shall have the meaning set forth in Section 3.13(e).

"<u>Permitted Liens</u>" shall mean those liens and encumbrances against the Property and/or the Project set forth on <u>Exhibit G</u> hereto.

"Permitted Mortgage" shall have the meaning set forth in Article 16 hereof.

"Plans and Specifications" shall mean final construction documents containing a site plan and working drawings and specifications for the Project, as submitted to the City as the basis for obtaining building permits for the Project.

"Prior Expenditure(s)" shall have the meaning set forth in Section 4.05(a) hereof.

"Project" shall have the meaning set forth in the Recitals hereof.

"Project Budget" shall mean the budget attached hereto as Exhibit H-1, showing the total cost of the Project by line item, furnished by Developer to DPD, in accordance with Section 3.03 hereof.

"Property" shall have the meaning set forth in the Recitals hereof.

"Purchase Price" shall have the meaning set forth in Section 3.13(a) hereof.

"RAP Approval Letter" shall have the meaning set forth in Section 3.13(e).

"REC(s)" shall have the meaning set forth in Section 3.13(e).

"Redevelopment Area" shall have the meaning set forth in the Recitals hereof.

"Redevelopment Plan" shall have the meaning set forth in the Recitals hereof.

"Redevelopment Project Costs" shall mean redevelopment project costs as defined in Section 5/11-74.4-3(q) of the Act that are included in the budget set forth in the Redevelopment Plan or otherwise referenced in the Redevelopment Plan.

"Released Claims" shall have the meaning set forth in Section 3.13(g).

"Remediation Work" shall mean all investigation, sampling, monitoring, testing, removal, response, disposal, storage, remediation, treatment and other activities necessary to obtain a Final NFR Letter for the Property in accordance with the terms and conditions of the SRP Documents, all requirements of the IEPA and all applicable Laws, including, without limitation, all applicable Environmental Laws.

"Requisition Form" shall mean the document, in the form attached hereto as Exhibit L, to be delivered by Developer to DPD pursuant to Section 4.04 of this Agreement.]

"Scope Drawings" shall mean preliminary construction documents containing a site plan and preliminary drawings and specifications for the Project.

"<u>SRP</u>" shall mean the IEPA's Site Remediation Program as set forth in Title XVII of the Illinois Environmental Protection Act, 415 ILCS 5/58 et seq., and the regulations promulgated thereunder.

"SRP Documents" shall mean all documents submitted to the IEPA under the SRP program, as amended or supplemented from time to time, including, without limitation, the Comprehensive Site Investigation and Remediation Objectives Report, the Remedial Action Plan, and the Remedial Action Completion Report.

"Survey" shall mean a plat of survey in the most recently revised form of ALTA/ACSM land title survey of the Property, meeting the 2011 Minimum Standard Detail Requirements for ALTA/ACSM Land Title Surveys, effective February 23, 2011, dated within 75 days prior to the Closing Date, acceptable in form and content to the City and the Title Company, prepared by a surveyor registered in the State of Illinois, certified to the City and the Title Company, and indicating whether the Property is in a flood hazard area as identified by the United States Federal Emergency Management Agency (and updates thereof to reflect improvements to the Property in connection with the construction of the Facility and related improvements as required by the City or lender(s) providing Lender Financing).

"<u>TACO</u>" shall mean the Tiered Approach to Corrective Action Objectives codified at 35 III. Adm. Code Part 742 et seq.

"<u>Term of the Agreement</u>" shall mean the period of time commencing]]on the Closing Date and ending on the later of: (a) [any date to which DPD and Developer have agreed] or (b) the date on which the Redevelopment Area is no longer in effect (through and including [[March 27, 2025 or December 31, 2026]] [Note: the date will be either the 23rd anniversary of the date of the adoption of the TIF Ordinances, or if the Redevelopment Plan provides, December 31st of the year succeeding the year in which the 23rd anniversary occurs.]

- "TIF Adoption Ordinance" shall have the meaning set forth in the Recitals hereof.
- "TIF Bonds" shall have the meaning set forth in the Recitals hereof, if any.
- "TIF Bond Ordinance" shall have the meaning set forth in the Recitals hereof.
- "<u>TIF Fund</u>" shall mean the special tax allocation fund created by the City in connection with the Redevelopment Area into which the Incremental Taxes will be deposited.
- "<u>TIF-Funded Improvements</u>" shall mean those improvements of the Project which (i) qualify as Redevelopment Project Costs, (ii) are eligible costs under the Redevelopment Plan and (iii) the City has agreed to pay for out of the City Funds, subject to the terms of this Agreement. Exhibit C lists the TIF-Funded Improvements for the Project.
 - "TIF Ordinances" shall have the meaning set forth in the Recitals hereof.
 - "Title Commitment" shall have the meaning set forth in Section 3.13(b) hereof.
 - "Title Company" shall mean Chicago Title Insurance Company.
- "<u>Title Policy</u>" shall mean a title insurance policy in the most recently revised ALTA or equivalent form, showing Developer as the insured, noting the recording of this Agreement as an encumbrance against the Property, and a subordination agreement in favor of the City with respect to previously recorded liens against the Property related to Lender Financing, if any, issued by the Title Company.
 - "USTs" shall have the meaning set forth in Section 3.13(f).
- "<u>WARN Act</u>" shall mean the Worker Adjustment and Retraining Notification Act (29 U.S.C. Section 2101 et seq.).
- "WBE(s)" shall mean a business identified in the Directory of Certified Women Business Enterprises published by the City's Department of Procurement Services, or otherwise certified by the City's Department of Procurement Services as a women-owned business enterprise, related to the Procurement Program or the Construction Program, as applicable.

SECTION 3. THE PROJECT

- 3.01 <u>The Project</u>. With respect to the Facility, Developer shall, pursuant to the Plans and Specifications and subject to the provisions of <u>Section 18.17</u> hereof: (i) commence construction no later than [[[November 15, 2021; (ii) receive a partial Certificate of Occupancy no later than December 15, 2022; and (iii) complete construction and conduct business operations therein no later than April 30, 2023 or June 30, 2023]]].
- 3.02 <u>Scope Drawings and Plans and Specifications</u>. Developer has delivered the Scope Drawings and Plans and Specifications to DPD and DPD has approved same. After such initial approval, subsequent proposed changes to the Scope Drawings or Plans and Specifications shall be submitted to DPD as a Change Order pursuant to <u>Section 3.04</u> hereof. The Scope Drawings and Plans and Specifications shall at all times conform to the Redevelopment Plan and all applicable federal, state and local laws, ordinances and regulations. Developer shall submit all

necessary documents to the City's Building Department, Department of Transportation and such other City departments or governmental authorities as may be necessary to acquire building permits and other required approvals for the Project.

- 3.03 Project Budget. Developer has furnished to DPD, and DPD has approved, a Project Budget showing total costs for the Project in an amount not less than Thirty-Eight Million and No/100 Dollars (\$38,000,000). Developer hereby certifies to the City that (a) the City Funds, together with Lender Financing and Equity described in Section 4.02 hereof, shall be sufficient to complete the Project. Developer hereby certifies to the City that (a) it has Lender Financing and Equity in an amount sufficient to pay for all Project costs; and (b) the Project Budget is true, correct and complete in all material respects. Developer shall promptly deliver to DPD certified copies of any Change Orders with respect to the Project Budget for approval pursuant to Section 3.04 hereof.
- 3.04 Change Orders. Except as provided below in this Section 3.04, all Change Orders (and documentation substantiating the need and identifying the source of funding therefor) relating to changes to the [[construction of the]] Project must be submitted by Developer to DPD concurrently with the progress reports described in Section 3.07 hereof; provided, that any Change Order relating to any of the following must be submitted by Developer to DPD for DPD's prior written approval: (a) a reduction in the gross or net square footage of [[the Project or Developer Space]] by five percent (5%) or more (either individually or cumulatively); (b) a change in the use of [[the Project or Developer Space]] to a use other than as described in Recital D to this Agreement; (c) a delay in the completion of the [[construction of the]] Project by six (6) months or more; or (d) Change Orders resulting in an aggregate increase to the Project Budget for the Project of ten percent (10%) or more. Developer shall not authorize or permit the performance of any work relating to any Change Order or the furnishing of materials in connection therewith prior to the receipt by Developer of DPD's written approval (to the extent said City prior approval is required pursuant to the terms of this Agreement). The Construction Contract, and each contract between the General Contractor and any subcontractor, shall contain a provision to this effect. An approved Change Order shall not be deemed to imply any obligation on the part of the City to increase the amount of City Funds which the City has pledged pursuant to this Agreement or provide any other additional assistance to Developer. Notwithstanding anything to the contrary in this Section 3.04, Change Orders other than those set forth above do not require DPD's prior written approval as set forth in this Section 3.04, but DPD shall be notified in writing of all such Change Orders within 10 business days after the execution of such Change Order and Developer, in connection with such notice, shall identify to DPD the source of funding therefor.
- 3.05 <u>DPD Approval</u>. Any approval granted by DPD of the Scope Drawings, Plans and Specifications and the Change Orders is for the purposes of this Agreement only and does not affect or constitute any approval required by any other City department or pursuant to any City ordinance, code, regulation or any other governmental approval, nor does any approval by DPD pursuant to this Agreement constitute approval of the quality, structural soundness or safety of the Property or the Project.
- 3.06 Other Approvals. Any DPD approval under this Agreement shall have no effect upon, nor shall it operate as a waiver of, Developer's obligations to comply with the provisions of Section 5.03 (Other Governmental Approvals) hereof. Developer shall not commence construction of the Project until Developer has obtained all necessary permits and approvals (including but not limited to DPD's approval of the Scope Drawings and Plans and Specifications) and proof of the General Contractor's and each subcontractor's bonding as required hereunder.

- 3.07 Progress Reports and Survey Updates. Developer shall provide DPD with written monthly progress reports detailing the status of the Project, including a revised completion date, if necessary (with any change in completion date being considered a Change Order, requiring DPD's written approval pursuant to Section 3.04). Developer shall provide three (3) copies of an updated Survey to DPD upon the request of DPD or any lender providing Lender Financing, reflecting improvements made to the Property.
- 3.08 <u>Inspecting Agent or Architect.</u> An independent agent or architect (other than Developer's architect) approved by DPD shall be selected to act as the inspecting agent or architect, at Developer's expense, for the Project. The inspecting agent or architect shall perform periodic inspections with respect to the Project, providing certifications with respect thereto to DPD, prior to requests for disbursement for costs related to the Project [hereunder] [pursuant to the Escrow Agreement.] DPD may, in its sole discretion, use the inspecting architect employed by any lender as the inspecting agent or architect.
- 3.09 <u>Barricades</u>. Prior to commencing any construction requiring barricades, Developer shall install a construction barricade of a type and appearance satisfactory to the City and constructed in compliance with all applicable federal, state or City laws, ordinances and regulations. DPD retains the right to approve the maintenance, appearance, color scheme, painting, nature, type, content and design of all barricades.
- 3.10 <u>Signs and Public Relations</u>. Developer shall erect a sign of size and style approved by the City in a conspicuous location on the Property during the Project, indicating that financing has been provided by the City. The City reserves the right to include the name, photograph, artistic rendering of the Project and other pertinent information regarding Developer, the Property and the Project in the City's promotional literature and communications.
- 3.11 <u>Utility Connections</u>. Developer may connect all on-site water, sanitary, storm and sewer lines constructed on the Property to City utility lines existing on or near the perimeter of the Property, provided Developer first complies with all City requirements governing such connections, including the payment of customary fees and costs related thereto.
- 3.12 Permit Fees. In connection with the Project, Developer shall be obligated to pay only those building, permit, engineering, tap on and inspection fees that are assessed on a uniform basis throughout the City of Chicago and are of general applicability to other property within the City of Chicago.
- 3.13 <u>Conveyance of Property from City to Developer</u>. The following provisions shall govern the City's conveyance of the Property to Developer:
- (a) Form of Deed. The City shall convey the Property to Developer by quitclaim deed (the "Deed") for the sum of Four Dollars (or One Dollar per parcel) ("Purchase Price"), subject to the terms of this Agreement and, without limiting the quitclaim nature of the deed, the following:
 - (i) the Redevelopment Plan;
 - (ii) the standard exceptions in an ALTA title insurance policy;

- (iii) all general real estate taxes and any special assessments or other taxes;
- (iv) all easements, encroachments, covenants and restrictions of record and not shown of record:
- (v) such other title defects as may exist; and
- (vi) any and all exceptions caused by the acts of the Developer or its agents.

The Developer and the City acknowledge and agree that the fair market value of the Property exceeds the Purchase Price and that the City has only agreed to convey the Property to Developer for the Purchase Price because the Developer has agreed to execute this Agreement and comply with its terms and conditions.

- Title and Survey. The Developer shall, no later than thirty (30) days prior to the Closing Date obtain at its expense and deliver to the City a Survey of the Property and a commitment for an owner's policy of title insurance issued by the Title Company (the "Title Commitment") in an amount not less than the fair market value of the Property. The Developer shall be solely responsible for and shall pay all costs associated with updating the Title Commitment (including all search, continuation and later-date fees), and obtaining the Title Policy and any endorsements. The City shall have no obligation to cure title defects; provided, however, if there are exceptions for general real estate taxes due or unpaid prior to the Closing Date with respect to the Property or liens for such unpaid property taxes, the City shall, as applicable, request that the County void the unpaid taxe's as provided in Section 21-100 of the Property Tax Code, 35 ILCS 200/21-100, or file an application for a Certificate of Error with the Cook County Assessor, or file a tax injunction suit or petition to vacate a tax sale in the Circuit Court of Cook County. If, after taking the foregoing actions and diligently pursuing same, the Property remains subject to any tax liens, or if the Property is encumbered with any other exceptions that would adversely affect the use and insurability of the Property for the development of the Project, the Developer shall, as its sole remedy, have the option to either (i) proceed with the purchase subject to all defects and exceptions, or (ii) terminate this Agreement, whereupon this Agreement shall be null and void and, except as otherwise specifically provided, neither party shall have any further right, duty or obligation hereunder with respect to the Property. If the Developer elects not to terminate this Agreement, the Developer agrees to accept title subject to all exceptions.
- (c) <u>Closing</u>. The conveyance of the Property shall take place on the Closing Date at the downtown offices of the Title Company or such other place as the parties may mutually agree upon in writing; provided, however, in no event shall the closing of the Property conveyance occur unless the Developer has satisfied all conditions precedent set forth in this Agreement, unless DOH, in its sole discretion, waives such conditions. On or before the Closing Date, the City shall deliver to the Title Company the Deed, all necessary state, county and municipal real estate transfer tax declarations, and an ALTA statement. The City will not provide a gap undertaking. The Developer shall pay to record the Deed and any other documents incident to the conveyance of the Property to the Developer. In the event the Developer requires conveyance through an escrow, Developer shall pay all escrow fees.

(d) "AS IS" SALE. THE DEVELOPER ACKNOWLEDGES THAT IT HAS HAD OR WILL HAVE ADEQUATE OPPORTUNITY TO INSPECT AND EVALUATE THE STRUCTURAL. PHYSICAL AND ENVIRONMENTAL CONDITION AND RISKS OF THE PROPERTY AND ACCEPTS THE RISK THAT ANY INSPECTION MAY NOT DISCLOSE ALL MATERIAL MATTERS AFFECTING THE PROPERTY. THE DEVELOPER AGREES TO ACCEPT THE PROPERTY IN ITS "AS IS," "WHERE IS" AND "WITH ALL FAULTS" CONDITION AT CLOSING WITHOUT ANY COVENANT, REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, OF ANY KIND, AS TO THE STRUCTURAL, PHYSICAL OR ENVIRONMENTAL CONDITION OF THE PROPERTY OR THE SUITABILITY OF THE PROPERTY FOR ANY PURPOSE WHATSOEVER. DEVELOPER ACKNOWLEDGES THAT IT IS RELYING SOLELY UPON ITS OWN INSPECTION AND OTHER DUE DILIGENCE ACTIVITIES AND NOT UPON ANY INFORMATION (INCLUDING. WITHOUT LIMITATION, ENVIRONMENTAL STUDIES OR REPORTS OF ANY KIND) PROVIDED BY OR ON BEHALF OF THE CITY OR ITS AGENTS OR EMPLOYEES WITH RESPECT THE DEVELOPER AGREES THAT IT IS ITS SOLE RESPONSIBILITY AND THERETO. OBLIGATION TO PERFORM ANY REMEDIATION WORK AND TAKE SUCH OTHER ACTION AS IS NECESSARY TO PUT THE PROPERTY IN A CONDITION WHICH IS SUITABLE FOR ITS INTENDED USE.

(e) Environmental Due Diligence.

- Prior to the Closing Date, the Developer shall perform a Phase I environmental site assessment of the Property in accordance with the requirements of the ASTM E 1527-13 standard ("Phase I ESA"). The City's Department of Assets, Information and Services ("AIS") shall have the right to review and approve the sufficiency of the Phase I ESA for the purpose of determining whether any environmental or health risks would be associated with the development of the Project. Upon AIS's request, the Developer shall perform additional studies and tests, including, without limitation, updating or expanding the Phase I ESA. If the Phase I ESA identifies any Recognized Environmental Condition(s) ("REC(s)"), the Developer shall perform a Phase II Environmental Site Assessment ("Phase II ESA"). If the Phase II ESA discloses the presence of contaminants exceeding applicable remediation objectives, the Developer shall enroll the Property in the IEPA's SRP and thereafter take all necessary and proper steps to obtain written approval from the IEPA of a Remedial Action Plan ("RAP Approval Letter"). The Developer acknowledges and agrees that the Closing will not occur, and it may not commence construction, until the IEPA issues, and AIS approves, the RAP Approval Letter for the Property. In the event the remediation is not completed prior to the Closing, the Deed shall include a covenant obligating the grantee to remediate the Property in accordance with the terms of this Section 3.13(e). The City shall have the right to review in advance and approve all documents submitted to the IEPA under the SRP, as amended or supplemented from time to time, including, without limitation, the SRP Documents and any changes thereto, and the Developer's estimate of the cost to perform the Remediation Work.
- (ii) The City shall grant the Developer the right, at its sole cost and expense, to enter the Property to perform the Phase I ESA and any other surveys, environmental assessments, soil tests and other due diligence it deems necessary or desirable to satisfy itself as to the condition of the Property. The obligation of the Developer to purchase the Property is conditioned upon the Developer being satisfied with the condition of the Property. If the Developer determines that it is not satisfied, in its sole and absolute discretion, with the condition of the Property, it may terminate this Agreement by written notice to the City any time prior to the Closing Date, whereupon this Agreement shall be null

and void and, except as otherwise specifically provided, neither party shall have any further right, duty or obligation hereunder with respect to the Property. If the Developer elects not to terminate this Agreement pursuant to this <u>Section 3.13(e)(ii)</u>, the Developer shall be deemed satisfied with the condition of the Property.

- Environmental Remediation. Upon receipt of the RAP Approval Letter for the Property, the Developer covenants and agrees to complete all Remediation Work necessary to obtain a Final NFR Letter using all reasonable means. The Developer shall cooperate and consult with the City at all relevant times (and in all cases upon the City's request) with respect to environmental matters. The Developer shall bear sole responsibility for all aspects of the Remediation Work including, but not limited to, the removal of pre-existing building foundations, soil exceeding residential (or commercial, as applicable) remediation objectives as determined by 35 III. Adm. Code Part 742, demolition debris, and the removal or treatment of Hazardous Materials. In addition, the Developer shall remove and close any identified underground storage tanks ("USTs") in accordance with applicable regulations, including 41 III. Adm. Code Part 175, and shall properly address any identified leaking USTs in accordance with 35 III. Adm. Code Part 734. The Developer shall promptly transmit to the City copies of all Environmental Documents prepared or received with respect to the Remediation Work, including, without limitation, any written communications delivered to or received from the IEPA or other regulatory agencies. The Developer acknowledges and agrees that the City will not issue a Certificate of Occupancy for the Property until the IEPA has issued, the City has approved, and the Developer has recorded with the Office of the Recorder of Deeds of Cook County, a Final NFR Letter for the Property, which approval shall not be unreasonably withheld. The Developer must abide by the terms and conditions of the Final NFR letter.
- Release and Indemnification. The Developer, on behalf of itself and the Developer Parties, shall be deemed to release, relinquish and forever discharge the City, its officers, agents and employees, from and against any and all Losses which the Developer or any of the Developer Parties ever had, now have, or hereafter may have, whether grounded in tort or contract or otherwise, in any and all courts or other forums, of whatever kind or nature, whether known or unknown, foreseen or unforeseen, now existing or occurring after the Closing Date, based upon, arising out of or in any way connected with, directly or indirectly (i) any environmental contamination, pollution or hazards associated with the Property or any improvements, facilities or operations located or formerly located thereon, including, without limitation, any release, emission, discharge, generation, transportation, treatment, storage or disposal of Hazardous Materials, or threatened release, emission or discharge of Hazardous Materials; (ii) the structural, physical or environmental condition of the Property, including, without limitation, the presence or suspected presence of Hazardous Materials in, on, under or about the Property or the migration of Hazardous Materials from or to the Property; (iii) any violation of, compliance with, enforcement of or liability under any Environmental Laws, including, without limitation, any governmental or regulatory body response costs, natural resource damages or Losses arising under CERCLA, and (iv) any investigation, cleanup, monitoring, remedial, removal or restoration work required by any federal, state or local governmental agency or political subdivision or other third party in connection or associated with the Property or any improvements, facilities or operations located or formerly located thereon (collectively, "Released Claims"). Furthermore, upon the Closing, the Developer shall indemnify, defend (through an attorney reasonably acceptable to the City) and hold the City and its officers, agents and employees harmless from and against any and all Losses which may be made or asserted by any third parties (including, without limitation, any of the Developer Parties) arising out of or in any way connected with, directly or indirectly, any of the Released Claims.

- shall run with the Property, and shall be binding upon all successors and assigns of the Developer with respect to the Property, including, without limitation, each and every person, firm, corporation, limited liability company, trust or other entity owning, leasing, occupying, using or possessing any portion of the Property under or through the Developer following the date of the Deed. The Developer acknowledges and agrees that the foregoing covenant of release constitutes a material inducement to the City to enter into this Agreement, and that, but for such release, the City would not have agreed to convey the Property to the Developer. It is expressly agreed and understood by and between the Developer and the City that, should any future obligation of the Developer or any of the Developer Parties arise or be alleged to arise in connection with any environmental, soil or other condition of the Property, neither the Developer nor any of the Developer Parties will assert that those obligations must be satisfied in whole or in part by the City because Section 3.13(g) contains a full, complete and final release of all such claims.
- (i) Notification to the Cook County Assessor of Change in Use and Ownership. Prior to the Closing Date, Developer shall complete a letter of notification, in accordance with 35 ILCS 200/15-20, notifying the Cook County Assessor that there has been a change in use and ownership of the Property. On the Closing Date, Developer shall pay to the Title Company the cost of sending the notification to the Cook County Assessor via certified mail, return receipt requested. After delivery of the notification, Developer shall forward a copy of the return receipt to DOH, with a copy to the City's Corporation Counsel's office.
- (g) <u>Survival</u>. This <u>Section 3.13</u> shall survive the Closing Date or any termination of this Agreement (regardless of the reason for such termination).

SECTION 4. FINANCING

4.01 <u>Total Project Cost and Sources of Funds</u>. The cost of the Project is estimated to be **[[[\$38,504,863]]]**, to be applied in the manner set forth in the Project Budget. Such costs shall be funded from the following sources:

Equity (subject to <u>Sections [4.03(b)]</u> and <u>4.06</u>) Lender Financing—Loans other than City Loan Lender Financing—City Loan Estimated City Funds (subject to <u>Section 4.03</u>) \$[19,525,863 (includes deferred developer fee)] \$[10,979,001(includes \$1.5MM of DCEO funds)] \$[3,000,000 (CITY Loan—probably HOME)] \$[5,000,000 (TIF Grant)]

ESTIMATED TOTAL

\$[38,504,863]

4.02 <u>Developer Funds</u>. Equity and/or Lender Financing may be used to pay any Project cost, including but not limited to Redevelopment Project Costs. Equity and/or Lender Financing shall be used to pay all Project costs, including but not limited to Redevelopment Project costs and costs of TIF-Funded Improvements.

4.03 City Funds.

- (a) <u>Uses of City Funds</u>. City Funds may only be used to pay directly or reimburse Developer for costs of TIF-Funded Improvements that constitute Redevelopment Project Costs. <u>Exhibit C</u> sets forth, by line item, the TIF-Funded Improvements for the Project, and the maximum amount of costs that may be paid by or reimbursed from City Funds for each line item therein (subject to <u>Sections 4.03(b) and 4.05(d)</u>), contingent upon receipt by the City of documentation satisfactory in form and substance to DPD evidencing such cost and its eligibility as a Redevelopment Project Cost.
- (b) <u>Sources of City Funds</u>. Subject to the terms and conditions of this Agreement, including but not limited to this <u>Section 4.03</u> and <u>Section 5</u> hereof, the City hereby agrees to provide City funds from the sources and in the amounts described directly below (the "City Funds") to pay for or reimburse Developer for the costs of the TIF-Funded Improvements:

Source of City Funds Maximum Amount

Incremental Taxes \$5,000,000

<u>provided</u>, <u>however</u>, that the total amount of City Funds expended for TIF-Funded Improvements shall be an amount not to exceed the lesser of Five Million and No/100 Dollars (\$5,000,000) or [Twelve and 9854/10000 percent ([12.9854%) of the actual total Project costs; and <u>provided further</u>, that the \$5,000,000 to be derived from Incremental Taxes shall be available to pay costs related to TIF-Funded Improvements and allocated by the City for that purpose only so long as:

- (i) The amount of the Incremental Taxes deposited into the TIF Fund shall be sufficient to pay for such costs; and
- (ii) The City has been reimbursed from Incremental Taxes for the amount previously disbursed by the City for TIF-Funded Improvements.

Developer acknowledges and agrees that the City's obligation to pay for TIF-Funded Improvements up to a maximum of \$5,000,000 is contingent upon the fulfillment of the conditions set forth in Section 4.03(a) and Section 4.03(b). In the event that such conditions are not fulfilled, the amount of Equity to be contributed by Developer pursuant to Section 4.01 hereof shall increase proportionately.

(c) INTENTIONALLY OMITTED

- (d) <u>Retainage</u>. The last disbursement of City Funds shall only be released upon completion of construction and issuance of the Certificate pursuant to Section 7.01.
- 4.04 <u>Construction Escrow</u>; <u>Requisition Form</u>. (a)]The City and Developer hereby agree to enter into the Escrow Agreement. All disbursements of Project funds [[[(except for the Prior Expenditures, if any, and acquisition costs disbursed through a deed and money escrow at the closing)]]] shall be made through the funding of draw requests with respect thereto pursuant to the Escrow Agreement and this Agreement. In case of any conflict between the terms of this Agreement and the Escrow Agreement, the terms of this Agreement shall control. [[[The City must

receive copies of, and retains the right to approve, any draw requests and related documents submitted to the Title Company for disbursements under the Escrow Agreement.]]]

(b) Along with the relevant draw request, upon completion of each of one-third of construction, two-thirds of construction and final completion, Developer shall also provide DPD with a Requisition Form, along with the documentation described therein. As set forth above, requisition for reimbursement of TIF-Funded Improvements shall be made not more than three times during construction of the Project and after at least each of (i) one-third of construction completion, (ii) two-thirds of construction completion and (iii) upon final completion. Developer shall meet with DPD at the request of DPD to discuss the Requisition Form(s) previously delivered.

4.05 Treatment of Prior Expenditures and Subsequent Disbursements.

(a) <u>Prior Expenditures</u>. Only those expenditures made by Developer with respect to the Project prior to the Closing Date, evidenced by documentation satisfactory to DPD and approved by DPD as satisfying costs covered in the Project Budget, shall be considered previously contributed Equity or Lender Financing hereunder (the "Prior Expenditures"). DPD shall have the right, in its sole discretion, to disallow any such expenditure as a Prior Expenditure. <u>Exhibit I</u> hereto sets forth the prior expenditures approved by DPD [as of the date hereof] as Prior Expenditures. [[[Prior Expenditures made for items other than TIF-Funded Improvements shall not be reimbursed to Developer, but shall reduce the amount of Equity and/or Lender Financing required to be contributed by Developer pursuant to Section 4.01 hereof.]]]

(b) INTENTIONALLY OMITTED.

(c) INTENTIONALLY OMITTED.

(d) <u>Allocation Among Line Items</u>. Disbursements for expenditures related to TIF-Funded Improvements may be allocated to and charged against the appropriate line only, with transfers of costs and expenses from one line item to another, without the prior written consent of DPD, being prohibited; <u>provided</u>, <u>however</u>, that such transfers among line items, in an amount not to exceed \$25,000 or \$100,000 in the aggregate, may be made without the prior written consent of DPD.

(e) INTENTIONALLY OMITTED

- 4.06 <u>Cost Overruns</u>. If the aggregate cost of the TIF-Funded Improvements exceeds City Funds available pursuant to <u>Section 4.03</u> hereof, or if the cost of completing the Project exceeds the Project Budget, Developer shall be solely responsible for such excess cost, and shall hold the City harmless from any and all costs and expenses of completing the TIF-Funded Improvements in excess of City Funds and of completing the Project.
- 4.07 <u>Preconditions of Disbursement</u>. Prior to each disbursement of City Funds hereunder, Developer shall submit documentation regarding the applicable expenditures to DPD, which shall be satisfactory to DPD in its sole discretion. Delivery by Developer to DPD of any request for disbursement of City Funds hereunder shall, in addition to the items therein expressly set forth, constitute a certification to the City, as of the date of such request for disbursement, that:
- (a) the total amount of the disbursement request represents the actual amount payable to (or paid to) the General Contractor and/or subcontractors who have performed work on the Project, and/or their payees;

- (b) all amounts shown as previous payments on the current disbursement request have been paid to the parties entitled to such payment;
- (c) Developer has approved all work and materials for the current disbursement request, and such work and materials conform to the Plans and Specifications;
- (d) the representations and warranties contained in this Redevelopment Agreement are true and correct and Developer is in compliance with all covenants contained herein;
- (e) Developer has received no notice and has no knowledge of any liens or claim of lien either filed or threatened against the Property except for the Permitted Liens [[or any liens that Developer bonds over]];
- (f) no Event of Default or condition or event which, with the giving of notice or passage of time or both, would constitute an Event of Default exists or has occurred; and
- (g) the Project is In Balance. The Project shall be deemed to be in balance ("In Balance") only if the total of the available Project funds equals or exceeds the aggregate of the amount necessary to pay all unpaid Project costs incurred or to be incurred in the completion of the Project. "Available Project Funds" as used herein shall mean: (i) the undisbursed City Funds; (ii) the undisbursed Lender Financing, if any; (iii) the undisbursed Equity and (iv) any other amounts deposited by Developer pursuant to this Agreement. Developer hereby agrees that, if the Project is not In Balance, Developer shall, within 10 days after a written request by the City, deposit with the escrow agent or will make available (in a manner acceptable to the City), cash in an amount that will place the Project In Balance, which deposit shall first be exhausted before any further disbursement of the City Funds shall be made.

The City shall have the right, in its discretion, to require Developer to submit further documentation as the City may require in order to verify that the matters certified to above are true and correct, and any [disbursement] [execution of a Certificate of Expenditure] by the City shall be subject to the City's review and approval of such documentation and its satisfaction that such certifications are true and correct; provided, however, that nothing in this sentence shall be deemed to prevent the City from relying on such certifications by Developer. In addition, Developer shall have satisfied all other preconditions of [disbursement of City Funds for each disbursement] [execution of a Certificate of Expenditure], including but not limited to requirements set forth in the Bond Ordinance, if any, TIF Bond Ordinance, if any, the Bonds, if any, the TIF Bonds, if any, the TIF Ordinances, this Agreement and/or the Escrow Agreement.

4.08 <u>Conditional Grant</u>. The City Funds being provided hereunder are being granted on a conditional basis, subject to the Developer's compliance with the provisions of this Agreement. The City Funds are subject to being reimbursed [[as set forth herein]].

4.09 INTENTIONALLY OMITTED.

SECTION 5. CONDITIONS PRECEDENT

The following conditions have been complied with to the City's satisfaction on or prior to the Closing Date:

- 5.01 <u>Project Budget</u>. Developer has submitted to DPD, and DPD has approved, a Project Budget in accordance with the provisions of Section 3.03 hereof.
- 5.02 <u>Scope Drawings and Plans and Specifications</u>. Developer has submitted to DPD, and DPD has approved, the Scope Drawings and Plans and Specifications accordance with the provisions of <u>Section 3.02</u> hereof.
- 5.03 Other Governmental Approvals. Developer has secured all other necessary approvals and permits required by any state, federal, or local statute, ordinance or regulation and has submitted evidence thereof to DPD.
- 5.04 Financing. Developer has furnished proof reasonably acceptable to the City that Developer has Equity and Lender Financing in the amounts set forth in Section 4.01 hereof to complete the Project and satisfy its obligations under this Agreement. If a portion of such funds consists of Lender Financing, Developer has furnished proof as of the Closing Date that the proceeds thereof are available to be drawn upon by Developer as needed and are sufficient (along with other sources set forth in [Section 4.01]) to complete the Project. Developer acknowledges that the City will be a party to any construction escrow agreement entered into by Developer regarding the Lender Financing. Any liens against the Property in existence at the Closing Date have been subordinated to certain encumbrances of the City set forth herein pursuant to a Subordination Agreement, in a form acceptable to the City, executed on or prior to the Closing Date, which is to be recorded, at the expense of Developer, with the Office of the Recorder of Deeds of Cook County.
- 5.05 Acquisition and Title. On the Closing Date, Developer has furnished the City with a copy of the Title Policy for the Property, certified by the Title Company, showing Developer as the named insured. The Title Policy is dated as of the Closing Date and contains only those title exceptions listed as Permitted Liens on Exhibit G hereto and evidences the recording of this Agreement pursuant to the provisions of Section 8.18 hereof. The Title Policy also contains such endorsements as shall be required by Corporation Counsel, including but not limited to an owner's comprehensive endorsement and satisfactory endorsements regarding zoning (e.g. 3.1 with parking), contiguity, location, access and survey. Developer has provided to DPD, on or prior to the Closing Date, documentation related to the purchase of the Property and certified copies of all easements and encumbrances of record with respect to the Property not addressed, to DPD's satisfaction, by the Title Policy and any endorsements thereto.
- 5.06 Evidence of Clean Title. Developer, at its own expense, has provided the City with searches as indicated in the chart below under Developer's name [[(and the following trade names of Developer: ______)]] showing no liens against Developer, the Property or any fixtures now or hereafter affixed thereto, except for the Permitted Liens:

Jurisdiction	Searches
Secretary of State	UCC, Federal tax
Cook County Recorder	UCC, Fixtures, Federal tax, State tax, Memoranda of judgments
U.S. District Court	Pending suits and judgments
Clerk of Circuit Court, Cook County	Pending suits and judgments

5.07 Surveys. Developer has furnished the City with three (3) copies of the Survey.

- 5.08 <u>Insurance</u>. Developer, at its own expense, has insured the Property in accordance with <u>Section 12</u> hereof, and has delivered certificates required pursuant to <u>Section 12</u> hereof evidencing the required coverages to DPD.
- 5.09 Opinion of Developer's Counsel. On the Closing Date, Developer has furnished the City with an opinion of counsel, substantially in the form attached hereto as Exhibit J, with such changes as required by or acceptable to Corporation Counsel. If Developer has engaged special counsel in connection with the Project, and such special counsel is unwilling or unable to give some of the opinions set forth in Exhibit J hereto, such opinions were obtained by Developer from its general corporate counsel.
- 5.10 <u>Evidence of Prior Expenditures</u>. Developer has provided evidence satisfactory to DPD in its sole discretion of the Prior Expenditures in accordance with the provisions of <u>Section 4.05(a)</u> hereof.
- 5.11 <u>Financial Statements</u>. Developer has provided Financial Statements to DPD for its most recent fiscal year, and audited or unaudited interim financial statements.
- 5.12 <u>Documentation</u>; <u>Employment Plan</u>. The Developer has provided documentation to HED, satisfactory in form and substance to HED, with respect to current employment matters in connection with the construction or rehabilitation work on the Project, including the reports described in Section 8.07. At least thirty (30) days prior to the Closing Date, the Developer has met with either (a) the Workforce Solutions division of DPD or (b) the manager of Project from DOH to review employment opportunities with the Developer during construction and after construction work on the Project is completed. On or before the Closing Date, Developer has provided to DPD or DOH, the Employment Plan for the Project (the "**Employment Plan**") which has been approved. The Employment Plan includes, without limitation, the Developer's estimates of future job openings, titles, position descriptions, qualifications, recruiting, training, placement and such other information as DPD has requested relating to the Project. It is currently anticipated that one hundred twenty-five (125) construction related jobs will be created during the construction period and approximately twenty-five (25) permanent jobs will be created related to property management and in the commercial space that is part of the Project.
- 5.13 Environmental. Developer has provided DPD with copies of that certain phase I environmental audit completed with respect to the Property [and any phase II environmental audit with respect to the Property required by the City.] Developer has provided the City with a letter from the environmental engineer(s) who completed such audit(s), authorizing the City to rely on such audits.
- 5.14 Corporate Documents; Economic Disclosure Statement. Developer has provided a copy of its Articles or Certificate of Incorporation [or Organization, as applicable] containing the original certification of the Secretary of State of its state of incorporation [or organization, as applicable]; certificates of good standing from the Secretary of State of its state of incorporation [or organization, as applicable] and all other states in which Developer is qualified to do business; a secretary's certificate in such form and substance as the Corporation Counsel may require; by-laws of the corporation [or operating agreement of the limited liability company, as applicable]; and such other corporate documentation as the City has requested. [Note: may revise this section to refer to other appropriate organizational documents]

Developer has provided to the City an EDS, dated as of the Closing Date, which is incorporated by reference, and Developer further will provide any other affidavits or certifications as may be required by federal, state or local law in the award of public contracts, all of which affidavits or certifications are incorporated by reference. Notwithstanding acceptance by the City of the EDS, failure of the EDS to include all information required under the Municipal Code renders this Agreement voidable at the option of the City. Developer and any other parties required by this **Section 5.14** to complete an EDS must promptly update their EDS(s) on file with the City whenever any information or response provided in the EDS(s) is no longer complete and accurate, including changes in ownership and changes in disclosures and information pertaining to ineligibility to do business with the City under Chapter 1-23 of the Municipal Code, as such is required under Sec. 2-154-020, and failure to promptly provide the updated EDS(s) to the City will constitute an event of default under this Agreement.

5.15 <u>Litigation</u>. Developer has provided to Corporation Counsel and DPD, a description of all pending or threatened litigation or administrative proceedings involving Developer, specifying, in each case, the amount of each claim, an estimate of probable liability, the amount of any reserves taken in connection therewith and whether (and to what extent) such potential liability is covered by insurance.

SECTION 6. AGREEMENTS WITH CONTRACTORS

6.01 Bid Requirement for General Contractor and Subcontractors. (a) Except as set forth in Section 6.01(b) below, prior to entering into an agreement with a General Contractor or any subcontractor for construction of the Project, Developer shall solicit, or shall cause the General Contractor to solicit, bids from qualified contractors eligible to do business with, [and having an office located in,] the City of Chicago, and shall submit all bids received to DPD for its inspection and written approval. (i) For the TIF-Funded Improvements, Developer shall select the General Contractor (or shall cause the General Contractor to select the subcontractor) submitting the lowest responsible bid who can complete the Project in a timely manner. If Developer selects a General Contractor (or the General Contractor selects any subcontractor) submitting other than the lowest responsible bid for the TIF-Funded Improvements, the difference between the lowest responsible bid and the bid selected may not be paid out of City Funds. (ii) For Project work other than the TIF-Funded Improvements, if Developer selects a General Contractor (or the General Contractor selects any subcontractor) who has not submitted the lowest responsible bid, the difference between the lowest responsible bid and the higher bid selected shall be subtracted from the actual total Project costs for purposes of the calculation of the amount of City Funds to be contributed to the Project pursuant to Section 4.03(b) hereof [If your transaction doesn't use the percentage formula for calculation of the amount of City Funds, delete (ii).] Developer shall submit copies of the Construction Contract to DPD in accordance with Section 6.02 below. Photocopies of all subcontracts entered or to be entered into in connection with the TIF-Funded Improvements shall be provided to DPD within five (5) business days of the execution thereof. Developer shall ensure that the General Contractor shall not (and shall cause the General Contractor to ensure that the subcontractors shall not) begin work on the Project until the Plans and Specifications have been approved by DPD and all requisite permits have been obtained.

[[[(b)] If, prior to entering into an agreement with a General Contractor for construction of the Project, Developer does not solicit bids pursuant to <u>Section 6.01(a)</u> hereof, then the fee of the General Contractor proposed to be paid out of City Funds shall not exceed 10% of the total amount of the Construction Contract. Except as explicitly stated in this paragraph, all other provisions of

<u>Section 6.01(a)</u> shall apply, including but not limited to the requirement that the General Contractor shall solicit competitive bids from all subcontractors. Developer and City may agree on additional provisions]]]

- 6.02 Construction Contract. Prior to the execution thereof, Developer shall deliver to DPD a copy of the proposed Construction Contract with the General Contractor selected to handle the Project in accordance with Section 6.01 above, for DPD's prior written approval, which shall be granted or denied within ten (10) business days after delivery thereof. Within ten (10) business days after execution of such contract by Developer, the General Contractor and any other parties thereto, Developer shall deliver to DPD and Corporation Counsel a certified copy of such contract together with any modifications, amendments or supplements thereto.
- 6.03 Performance and Payment Bonds. Prior to commencement of construction of any portion of the Project, Developer shall require that the General Contractor be bonded for its performance and payment by sureties having an AA rating or better using American Institute of Architect's Form No. A311 or its equivalent. Prior to the commencement of any portion of the Project which includes work on the public way, Developer shall require that the General Contractor be bonded for its payment by sureties having an AA rating or better using a bond in the form attached as Exhibit O hereto. The City shall be named as obligee or co-obligee on any such bonds.
- 6.04 <u>Employment Opportunity</u>. Developer shall contractually obligate and cause the General Contractor and each subcontractor to agree to the provisions of Section 10 hereof.
- 6.05 Other Provisions. In addition to the requirements of this Section 6, the Construction Contract and each contract with any subcontractor shall contain provisions required pursuant to Section 3.04 (Change Orders), Section 8.09 (Prevailing Wage), Section 10.01(e) (Employment Opportunity), Section 10.02 (City Resident Employment Requirement) [Note: confirm with DPD whether City residency requirements are to apply to each contract, or to the Project as a whole.] Section 10.03 (MBE/WBE Requirements, as applicable), Section 12 (Insurance) and Section 14.01 (Books and Records) hereof. Photocopies of all contracts or subcontracts entered or to be entered into in connection with the TIF-Funded Improvements shall be provided to DPD within five (5) business days of the execution thereof.

SECTION 7. COMPLETION OF CONSTRUCTION OR REHABILITATION

- 7.01 Certificate of Completion of Construction or Rehabilitation. Upon completion of the construction of the Project in accordance with the terms of this Agreement [and after the final disbursement from the Escrow], and upon Developer's written request, DPD shall issue to Developer a Certificate in recordable form certifying that Developer has fulfilled its obligation to complete the Project in accordance with the terms of this Agreement. DPD shall respond to Developer's written request for a Certificate within forty-five (45) days by issuing either a Certificate or a written statement detailing the ways in which the Project does not conform to this Agreement or has not been satisfactorily completed, and the measures which must be taken by Developer in order to obtain the Certificate. Developer may resubmit a written request for a Certificate upon completion of such measures.
- 7.02 Effect of Issuance of Certificate; Continuing Obligations. The Certificate relates only to the [construction/ rehabilitation] of the Project, and upon its issuance, the City will certify that the terms of the Agreement specifically related to Developer's obligation to complete such activities

have been satisfied. After the issuance of a Certificate, however, all executory terms and conditions of this Agreement and all representations and covenants contained herein will continue to remain in full force and effect throughout the Term of the Agreement as to the parties described in the following paragraph, and the issuance of the Certificate shall not be construed as a waiver by the City of any of its rights and remedies pursuant to such executory terms.

Those covenants specifically described at Sections 8.02, 8.06, 8.19 and 8.24 as covenants that run with the land are the only covenants in this Agreement intended to be binding upon any transferee of the Property (including an assignee as described in the following sentence) throughout the Term of the Agreement notwithstanding the issuance of a Certificate; provided, that upon the issuance of a Certificate, the covenants set forth in Section 8.02 shall be deemed to have been fulfilled. The other executory terms of this Agreement that remain after the issuance of a Certificate shall be binding only upon Developer or a permitted assignee of Developer who, pursuant to [Section 18.15] of this Agreement, has contracted to take an assignment of Developer's rights under this Agreement and assume Developer's liabilities hereunder.

- 7.03 <u>Failure to Complete</u>. If Developer fails to complete the Project in accordance with the terms of this Agreement, then the City has, but shall not be limited to, any of the following rights and remedies:
- (a) the right to terminate this Agreement and cease all disbursement of City Funds not yet disbursed pursuant hereto;
- (b) the right (but not the obligation) to complete those TIF-Funded Improvements that are public improvements and to pay for the costs of TIF-Funded Improvements (including interest costs) out of City Funds or other City monies. In the event that the aggregate cost of completing the TIF-Funded Improvements exceeds the amount of City Funds available pursuant to Section 4.01, Developer shall reimburse the City for all reasonable costs and expenses incurred by the City in completing such TIF-Funded Improvements in excess of the available City Funds; and
 - (c) the right to seek reimbursement of the City Funds from Developer.
- 7.04 <u>Notice of Expiration of Term of Agreement</u>. Upon the expiration of the Term of the Agreement, DPD shall provide Developer, at Developer's written request, with a written notice in recordable form stating that the Term of the Agreement has expired.

SECTION 8. COVENANTS/REPRESENTATIONS/WARRANTIES OF DEVELOPER.

- 8.01 <u>General</u>. Developer, GP Manager and Generations each individually and collectively represent, warrant and covenant, as of the date of this Agreement and as of the date of each disbursement of City Funds hereunder, that:
- (a) (i) Developer is a Delaware limited partnership duly organized, validly existing, qualified to do business in its state of incorporation/organization and in Illinois, and licensed to do business in any other state where, due to the nature of its activities or properties, such qualification or license is required, (ii) GP Manager is a Delaware limited liability company duly organized, validly existing, qualified to do business in its state of incorporation/organization and in Illinois, and licensed to do business in any other state where, due to the nature of its activities or properties, such qualification or license is required, and (iii) Generations is an Illinois not-for-profit corporation duly organized,

validly existing, qualified to do business in Illinois, and licensed to do business in any other state where, due to the nature of its activities or properties, such qualification or license is required ;

- (b) Developer, GP Manager and Generations each has the right, power and authority to enter into, execute, deliver and perform this Agreement;
- (c) the execution, delivery and performance by each of Developer, GP Manager and Generations of this Agreement has been duly authorized by all necessary partnership, company and corporate, as applicable, action, and does not and will not violate its [Articles of Incorporation/Articles of Organization] or [by-laws/partnership agreement/operating agreement] as amended and supplemented, any applicable provision of law, or constitute a breach of, default under or require any consent under any agreement, instrument or document to which Developer is now a party or by which Developer is now or may become bound;
- (d) unless otherwise permitted or not prohibited pursuant to or under the terms of this Agreement, Developer shall acquire and shall maintain good, indefeasible and merchantable fee simple title to the Property (and all improvements thereon) free and clear of all liens (except for the Permitted Liens, Lender Financing as disclosed in the Project Budget and non-governmental charges that Developer is contesting in good faith pursuant to Section 8.15 hereof)
- (e) Developer is now and for the Term of the Agreement shall remain solvent and able to pay its debts as they mature;
- (f) there are no actions or proceedings by or before any court, governmental commission, board, bureau or any other administrative agency pending, threatened or affecting Developer which would impair its ability to perform under this Agreement;
- (g) Developer has and shall maintain all government permits, certificates and consents (including, without limitation, appropriate environmental approvals) necessary to conduct its business and to construct, complete and operate the Project;
- (h) Developer is not in default with respect to any indenture, loan agreement, mortgage, deed, note or any other agreement or instrument related to the borrowing of money to which Developer is a party or by which Developer is bound;
- (i) the Financial Statements are, and when hereafter required to be submitted will be, complete, correct in all material respects and accurately present the assets, liabilities, results of operations and financial condition of Developer, and there has been no material adverse change in the assets, liabilities, results of operations or financial condition of Developer since the date of Developer's most recent Financial Statements;
- (j) prior to the issuance of a Certificate, Developer shall not do any of the following without the prior written consent of DPD: (1) be a party to any merger, liquidation or consolidation; (2) sell, transfer, convey, lease or otherwise dispose of all or substantially all of its assets or any portion of the Property (including but not limited to any fixtures or equipment now or hereafter attached thereto) except in the ordinary course of business; (3) enter into any transaction outside the ordinary course of Developer's business; (4) assume, guarantee, endorse, or otherwise become liable in connection with the obligations of any other person or entity; or (5) enter into any transaction that would cause a material and detrimental change to Developer's financial condition;

- (k) Developer has not incurred, and, prior to the issuance of a Certificate, shall not, without the prior written consent of the Commissioner of DPD, allow the existence of any liens against the Property (or improvements thereon) other than the Permitted Liens; or incur any indebtedness, secured or to be secured by the Property (or improvements thereon) or any fixtures now or hereafter attached thereto, except Lender Financing disclosed in the Project Budget; and [See Note to subsection (d) above.]
- (I) has not made or caused to be made, directly or indirectly, any payment, gratuity or offer of employment in connection with the Agreement or any contract paid from the City treasury or pursuant to City ordinance, for services to any City agency ("City Contract") as an inducement for the City to enter into the Agreement or any City Contract with Developer in violation of Chapter 2-156-120 of the Municipal Code;
- (m) neither Developer nor any affiliate of Developer is listed on any of the following lists maintained by the Office of Foreign Assets Control of the U.S. Department of the Treasury, the Bureau of Industry and Security of the U.S. Department of Commerce or their successors, or on any other list of persons or entities with which the City may not do business under any applicable law, rule, regulation, order or judgment: the Specially Designated Nationals List, the Denied Persons List, the Unverified List, the Entity List and the Debarred List. For purposes of this subparagraph (m) only, the term Affiliate," when used to indicate a relationship with a specified person or entity, means a person or entity that, directly or indirectly, through one or more intermediaries, controls, is controlled by or is under common control with such specified person or entity, and a person or entity shall be deemed to be controlled by another person or entity, if controlled in any manner whatsoever that results in control in fact by that other person or entity (or that other person or entity and any persons or entities with whom that other person or entity is acting jointly or in concert), whether directly or indirectly and whether through share ownership, a trust, a contract or otherwise.
- (n) Developer understands that (i) the City Funds are limited obligations of the City, payable solely from moneys on deposit in the [[[______ Account]]] of the TIF Fund; (ii) the City Funds do not constitute indebtedness of the City within the meaning of any constitutional or statutory provision or limitation; (iii) Developer will have no right to compel the exercise of any taxing power of the City for payment of the City Funds; and (iv) the City Funds do not and will not represent or constitute a general obligation or a pledge of the faith and credit of the City, the State of Illinois or any political subdivision thereof;
- (o) Developer has sufficient knowledge and experience in financial and business matters, including municipal projects and revenues of the kind represented by the City Funds, and has been supplied with access to information to be able to evaluate the risks associated with the receipt of City Funds;
- (p) Developer understands that there is no assurance as to the amount or timing of receipt of City Funds, and that the amounts of City Funds actually received by such party are likely to be substantially less than the maximum amounts set forth in Section 4.03(b);
- (q) Developer understands it may not sell, assign, pledge or otherwise transfer its interest in this Agreement or City Funds in whole or in part except in accordance with the terms of Section 18.21 of this Agreement, and, to the fullest extent permitted by law, agrees to indemnify the City for any losses, claims, damages or expenses relating to or based upon any sale, assignment, pledge or transfer of City Funds in violation of this Agreement; and

- (r) Developer acknowledges that with respect to City Funds, the City has no obligation to provide any continuing disclosure to the Electronic Municipal Market Access System maintained by the Municipal Securities Rulemaking Board, to any holder of a note relating to City Funds or any other person under Rule 15c2-12 of the Commission promulgated under the Securities Exchange Act of 1934 or otherwise, and shall have no liability with respect thereto.
- 8.02 <u>Covenant to Redevelop</u>. Upon DPD's approval of the Project Budget, the Scope Drawings and Plans and Specifications as provided in <u>Sections 3.02</u> and <u>3.03</u> hereof, and Developer's receipt of all required building permits and governmental approvals, Developer shall redevelop the Property in accordance with this Agreement and all Exhibits attached hereto, the TIF Ordinances, [the Bond Ordinance, the TIF Bond Ordinance], the Scope Drawings, Plans and Specifications, Project Budget and all amendments thereto, and all federal, state and local laws, ordinances, rules, regulations, executive orders and codes applicable to the Project, the Property and/or Developer. The covenants set forth in this Section shall run with the land and be binding upon any transferee, but shall be deemed satisfied upon issuance by the City of a Certificate with respect thereto.
- 8.03 <u>Redevelopment Plan</u>. Developer represents that the Project is and shall be in compliance with all of the terms of the Redevelopment Plan, which is hereby incorporated by reference into this Agreement.
- 8.04 <u>Use of City Funds</u>. City Funds disbursed to Developer shall be used by Developer solely to pay for (or to reimburse Developer for its payment for) the TIF-Funded Improvements as provided in this Agreement.
- 8.05 Other Bonds. Developer shall, at the request of the City, agree to any reasonable amendments to this Agreement that are necessary or desirable in order for the City to issue (in its sole discretion) any bonds in connection with the Redevelopment Area, the proceeds of which may be used to reimburse the City for expenditures made in connection with, or provide a source of funds for the payment for, the TIF-Funded Improvements (the "Bonds"); provided, however, that any such amendments shall not have a material adverse effect on Developer or the Project. Developer shall, at Developer's expense, cooperate and provide reasonable assistance in connection with the marketing of any such Bonds, including but not limited to providing written descriptions of the Project, making representations, providing information regarding its financial condition and assisting the City in preparing an offering statement with respect thereto.
- 8.06 <u>Job Creation and Retention; Covenant to Remain in the City.</u> [[[Not less than OR It is currently anticipated that]]] one hundred twenty- five (125) full-time equivalent, construction related jobs will be created by Developer during the construction period; and [[[not less OR approximately twenty-five]]] additional [[full-time and/or part-time]] equivalent, permanent jobs related to management of the Facility and/or jobs at the commercial space that is part of the Facility shall be created by Developer within one (1) year of completion of the Project. The full-time equivalent, permanent jobs shall be retained by Developer through the Term of the Agreement.

The covenants set forth in this Section shall run with the land and be binding upon any transferee.

8.07. Employment Opportunity[; Progress Reports]. Developer covenants and agrees to abide by, and contractually obligate and use reasonable efforts to cause the General Contractor and each subcontractor to abide by the terms set forth in Section 10 hereof. Developer shall deliver

to the City written progress reports detailing compliance with the requirements of <u>Sections 8.09</u>, <u>10.02 and 10.03</u> of this Agreement. Such reports shall be delivered to the City when the Project is **[[[25%**, 50%, 70% OR one-thirds, two-thirds**]]]** and 100% completed (based on the amount of expenditures incurred in relation to the Project Budget). If any such reports indicate a shortfall in compliance, Developer shall also deliver a plan to DPD which shall outline, to DPD's satisfaction, the manner in which Developer shall correct any shortfall.

- 8.08. <u>Employment Profile</u>. Developer shall submit, and contractually obligate and cause the General Contractor or any subcontractor to submit, to DPD, from time to time, statements of its employment profile upon DPD's request.
- 8.09. <u>Prevailing Wage</u>. Developer covenants and agrees to pay, and to contractually obligate and cause the General Contractor and each subcontractor to pay, the prevailing wage rate as ascertained by the federal government pursuant to the Davis-Bacon Act, to all their respective employees working on constructing the Project or otherwise completing the TIFoFunder Improvements. All such contracts shall list the specified rates to be paid to all laborers, workers and mechanics for each craft or type of worker or mechanic employed pursuant to such contract. If federal prevailing wage rates are revised, the revised rates shall apply to all such contracts. Upon the City's request, Developer shall provide the City with copies of all such contracts entered into by Developer or the General Contractor to evidence compliance with this <u>Section 8.09</u>.
- 8.10. <u>Arms-Length Transactions</u>. Unless DPD has given its prior written consent with respect thereto, no Affiliate of Developer may receive any portion of City Funds, directly or indirectly, in payment for work done, services provided or materials supplied in connection with any TIF-Funded Improvement. Developer shall provide information with respect to any entity to receive City Funds directly or indirectly (whether through payment to the Affiliate by Developer and reimbursement to Developer for such costs using City Funds, or otherwise), upon DPD's request, prior to any such disbursement.
- 8.11. <u>Conflict of Interest</u>. Pursuant to Section 5/11-74.4-4(n) of the Act, Developer represents, warrants and covenants that, to the best of its knowledge, no member, official, or employee of the City, or of any commission or committee exercising authority over the Project, the Redevelopment Area or the Redevelopment Plan, or any consultant hired by the City or Developer with respect thereto, owns or controls, has owned or controlled or will own or control any interest, and no such person shall represent any person, as agent or otherwise, who owns or controls, has owned or controlled, or will own or control any interest, direct or indirect, in Developer's business, the Property or any other property in the Redevelopment Area.
- 8.12. <u>Disclosure of Interest</u>. Developer's counsel has no direct or indirect financial ownership interest in Developer, the Property or any other aspect of the Project.
- 8.13. <u>Financial Statements</u>. Developer shall obtain and provide to DPD Financial Statements for Developer's fiscal year ended 2020 and each year thereafter for the Term of the Agreement. In addition, Developer shall submit unaudited financial statements as soon as reasonably practical following the close of each fiscal year and for such other periods as DPD may request.
- 8.14. <u>Insurance</u>. Developer, at its own expense, shall comply with all provisions of <u>Section 12</u> hereof.

8.15 Non-Governmental Charges. (a) Payment of Non-Governmental Charges. Except for the Permitted Liens, Developer agrees to pay or cause to be paid when due any Non-Governmental Charge assessed or imposed upon the Project, the Property or any fixtures that are or may become attached thereto, which creates, may create, or appears to create a lien upon all or any portion of the Property or Project; provided however, that if such Non-Governmental Charge may be paid in installments, Developer may pay the same together with any accrued interest thereon in installments as they become due and before any fine, penalty, interest, or cost may be added thereto for nonpayment. Developer shall furnish to DPD, within thirty (30) days of DPD's request, official receipts from the appropriate entity, or other proof satisfactory to DPD, evidencing payment of the Non-Governmental Charge in question.

(b) Right to Contest. Developer has the right, before any delinquency occurs:

- (i) to contest or object in good faith to the amount or validity of any Non-Governmental Charge by appropriate legal proceedings properly and diligently instituted and prosecuted, in such manner as shall stay the collection of the contested Non-Governmental Charge, prevent the imposition of a lien or remove such lien, or prevent the sale or forfeiture of the Property (so long as no such contest or objection shall be deemed or construed to relieve, modify or extend Developer's covenants to pay any such Non-Governmental Charge at the time and in the manner provided in this Section 8.15); or
- (ii) at DPD's sole option, to furnish a good and sufficient bond or other security satisfactory to DPD in such form and amounts as DPD shall require, or a good and sufficient undertaking as may be required or permitted by law to accomplish a stay of any such sale or forfeiture of the Property or any portion thereof or any fixtures that are or may be attached thereto, during the pendency of such contest, adequate to pay fully any such contested Non-Governmental Charge and all interest and penalties upon the adverse determination of such contest.
- 8.16 <u>Developer's Liabilities</u>. Developer shall not enter into any transaction that would materially and adversely affect its ability to perform its obligations hereunder or to repay any material liabilities or perform any material obligations of Developer to any other person or entity. Developer shall immediately notify DPD of any and all events or actions which may materially affect Developer's ability to carry on its business operations or perform its obligations under this Agreement or any other documents and agreements.
- 8.17 <u>Compliance with Laws</u>. To the best of Developer's knowledge, after diligent inquiry, the Property and the Project are and shall be in compliance with all applicable federal, state and local laws, statutes, ordinances, rules, regulations, executive orders and codes pertaining to or affecting the Project and the Property. Upon the City's request, Developer shall provide evidence satisfactory to the City of such compliance.
- 8.18 Recording and Filing. Developer shall cause this Agreement, certain exhibits (as specified by Corporation Counsel), all amendments and supplements hereto to be recorded and filed against the Property on the date hereof in the conveyance and real property records of the county in which the Project is located. [This Agreement shall be recorded prior to any mortgage made in connection with Lender Financing.] [Note: If this Agreement is not recorded first, a subordination agreement will have to be prepared and executed.] Developer shall pay all fees and charges incurred in connection with any such recording. Upon recording, Developer shall

28

immediately transmit to the City an executed original of this Agreement showing the date and recording number of record.

8.19 Real Estate Provisions.

(a) Governmental Charges.

- (i) <u>Payment of Governmental Charges</u>. Developer agrees to pay or cause to be paid when due all Governmental Charges (as defined below) which are assessed or imposed upon Developer, the Property or the Project, or become due and payable, and which create, may create, [or appear to create-?] a lien upon Developer or all or any portion of the Property or the Project. "Governmental Charge" shall mean all federal, State, county, the City, or other governmental (or any instrumentality, division, agency, body, or department thereof) taxes, levies, assessments, charges, liens, claims or encumbrances (except for those assessed by foreign nations, states other than the State of Illinois, counties of the State other than Cook County, and municipalities other than the City) relating to Developer, the Property or the Project including but not limited to real estate taxes.
- (ii) Right to Contest. Developer has the right before any delinquency occurs to contest or object in good faith to the amount or validity of any Governmental Charge by appropriate legal proceedings properly and diligently instituted and prosecuted in such manner as shall stay the collection of the contested Governmental Charge and prevent the imposition of a lien or the sale or forfeiture of the Property. Developer's right to challenge real estate taxes applicable to the Property is limited as provided for in Section 8.19(c) below; provided, that such real estate taxes must be paid in full when due and may be disputed only after such payment is made. No such contest or objection shall be deemed or construed in any way as relieving, modifying or extending Developer's covenants to pay any such Governmental Charge at the time and in the manner provided in this Agreement unless Developer has given prior written notice to DPD of Developer's intent to contest or object to a Governmental Charge and, unless, at DPD's sole option,
- (i) Developer shall demonstrate to DPD's satisfaction that legal proceedings instituted by Developer contesting or objecting to a Governmental Charge shall conclusively operate to prevent or remove a lien against, or the sale or forfeiture of, all or any part of the Property to satisfy such Governmental Charge prior to final determination of such proceedings; and/or
- (ii) Developer shall furnish a good and sufficient bond or other security satisfactory to DPD in such form and amounts as DPD shall require, or a good and sufficient undertaking as may be required or permitted by law to accomplish a stay of any such sale or forfeiture of the Property during the pendency of such contest, adequate to pay fully any such contested Governmental Charge and all interest and penalties upon the adverse determination of such contest.
- (b) <u>Developer's Failure To Pay Or Discharge Lien</u>. If Developer fails to pay any Governmental Charge or to obtain discharge of the same, Developer shall advise DPD thereof in writing, at which time DPD may, but shall not be obligated to, and without waiving or releasing any obligation or liability of Developer under this Agreement, in DPD's sole discretion, make such payment, or any part thereof, or obtain such discharge and take any other action with respect thereto which DPD deems advisable. All sums so paid by DPD, if any, and any expenses, if any.

including reasonable attorneys' fees, court costs, expenses and other charges relating thereto, shall be promptly disbursed to DPD by Developer. Notwithstanding anything contained herein to the contrary, this paragraph shall not be construed to obligate the City to pay any such Governmental Charge. Additionally, if Developer fails to pay any Governmental Charge, the City, in its sole discretion, may require Developer to submit to the City audited Financial Statements at Developer's own expense.

- [[[(c) Real Estate Taxes. Except as otherwise approved in writing by the City in connection with an application for a Class 9 tax classification incentive (the "Class 9 Incentive") applied for by Developer as such Class 9 incentive has been established by the Cook County Property Tax Incentive Assessment Classification Ordinance (the "Classification Ordinance") applied for by Developer pursuant to the Classification Ordinance, Developer shall conduct itself with respect to the Property pursuant to the restrictions set forth below.
 - (i) Acknowledgment of Real Estate Taxes. Developer agrees that (A) for the purpose of this Agreement, the [total projected] minimum assessed value of the Property [that is necessary to support the debt service indicated] ("Minimum Assessed Value") is shown on Exhibit K attached hereto and incorporated herein by reference for the years noted on Exhibit K; [note: if you are drafting an agreement for a pay as you go transaction, there will be no debt service now, and Exhibit K will be used to allow the parties to estimate future increment and as the exhibit for (iii) below. Also, may need Exhibit K in case bonds related to the project are issued in the future] (B) Exhibit K sets forth the specific improvements which will generate the fair market values, assessments, equalized assessed values and taxes shown thereon [NOTE: Exhibit K does not often set forth specific improvements, so this may need to be modified for each transaction]; and (C) the real estate taxes anticipated to be generated and derived from the respective portions of the Property and the Project for the years shown are fairly and accurately indicated in Exhibit K.
 - (ii) Real Estate Tax Exemption. With respect to the Property or the Project, neither Developer nor any agent, representative, lessee, tenant, assignee, transferee or successor in interest to Developer shall, during the Term of this Agreement, seek, or authorize any exemption (as such term is used and defined in the Illinois Constitution, Article IX, Section 6 (1970)) for any year that the Redevelopment Plan is in effect.
 - (iii) No Reduction in Real Estate Taxes. Neither Developer nor any agent, representative, lessee, tenant, assignee, transferee or successor in interest to Developer shall, during the Term of this Agreement, directly or indirectly, initiate, seek or apply for proceedings in order to lower the assessed value of all or any portion of the Property or the Project below the amount of the Minimum Assessed Value as shown in Exhibit K for the applicable year.
 - (iv) <u>No Objections</u>. Neither Developer nor any agent, representative, lessee, tenant, assignee, transferee or successor in interest to Developer, shall object to or in any way seek to interfere with, on procedural or any other grounds, the filing of any Underassessment Complaint or subsequent proceedings related thereto with the Cook County Assessor or with the Cook County Board of Appeals, by either the City or any taxpayer. The term AUnderassessment Complaint" as used in this Agreement shall mean any complaint seeking to increase the assessed value of the Property up to (but not above) the Minimum Assessed Value as shown in Exhibit K.

- (v) Covenants Running with the Land. The parties agree that the restrictions contained in this Section 8.19(c) are covenants running with the land and this Agreement shall be recorded by Developer as a memorandum thereof, at Developer's expense, with the Cook County Recorder of Deeds on the Closing Date. These restrictions shall be binding upon Developer and its agents, representatives, lessees, successors, assigns and transferees from and after the date hereof, provided however, that the covenants shall be released when the Redevelopment Area is no longer in effect. Developer agrees that any sale, lease, conveyance, or transfer of title to all or any portion of the Property or Redevelopment Area from and after the date hereof shall be made explicitly subject to such covenants and restrictions. Notwithstanding anything contained in this Section 8.19(c) to the contrary, the City, in its sole discretion and by its sole action, without the joinder or concurrence of Developer, its successors or assigns, may waive and terminate Developer's covenants and agreements set forth in this Section 8.19(c).[][][]
- (d) <u>Notification to the Cook County Assessor of Change in Use and Ownership.</u> Prior to the Closing Date, Developer shall complete a letter of notification, in accordance with 35 ILCS 200/15-20, notifying the Cook County Assessor that there has been a change in use and ownership of the Property. On the Closing Date, Developer shall pay to the Title Company the cost of sending the notification to the Cook County Assessor via certified mail, return receipt requested. After delivery of the notification, Developer shall forward a copy of the return receipt to DPD, with a copy to the City's Corporation Counsel's office.
- 8.20. <u>Annual Report(s)</u>. Beginning with the issuance of the Certificate and continuing throughout the Term of the Agreement, Developer shall submit to DPD the Annual Compliance Report within 30 days after the end of the calendar year to which the Annual Compliance Report relates.
- 8.21. <u>Inspector General</u>. It is the duty of Developer and the duty of any bidder, proposer, contractor, subcontractor, and every applicant for certification of eligibility for a City contract or program, and all of Developer's officers, directors, agents, partners, and employees and any such bidder, proposer, contractor, subcontractor or such applicant: (a) to cooperate with the Inspector General in any investigation or hearing undertaken pursuant to Chapter 2-56 of the Municipal Code and (b) to cooperate with the Legislative Inspector General in any investigation undertaken pursuant to Chapter 2-55 of the Municipal Code. Developer represents that it understands and will abide by all provisions of Chapters 2-56 and 2-55 of the Municipal Code and that it will inform subcontractors of this provision and require their compliance.

8.22 [INTENTIONALLY OMITTED]

8.23. FOIA and Local Records Act Compliance.

(a) FOIA. The Developer acknowledges that the City is subject to the Illinois Freedom of Information Act, 5 ILCS 140/1 et. seq., as amended ("FOIA"). The FOIA requires the City to produce records (very broadly defined in FOIA) in response to a FOIA request in a very short period of time, unless the records requested are exempt under the FOIA. If the Developer receives a request from the City to produce records within the scope of FOIA, then the Developer covenants to comply with such request within 48 hours of the date of such request. Failure by the Developer to timely comply with such request shall be an Event of Default.

- (b) Exempt Information. Documents that the Developer submits to the City under Section 8.21, (Annual Compliance Report) or otherwise during the Term of the Agreement that contain trade secrets and commercial or financial information may be exempt if disclosure would result in competitive harm. However, for documents submitted by the Developer to be treated as a trade secret or information that would cause competitive harm, FOIA requires that Developer mark any such documents as "proprietary, privileged or confidential." If the Developer marks a document as "proprietary, privileged and confidential", then DPD will evaluate whether such document may be withheld under the FOIA. DPD, in its discretion, will determine whether a document will be exempted from disclosure, and that determination is subject to review by the Illinois Attorney General's Office and/or the courts.
- (c) Local Records Act. The Developer acknowledges that the City is subject to the Local Records Act, 50 ILCS 205/1 et. seq, as amended (the "Local Records Act"). The Local Records Act provides that public records may only be disposed of as provided in the Local Records Act. If requested by the City, the Developer covenants to use its best efforts consistently applied to assist the City in its compliance with the Local Records Act
- 8.24 Affordable Housing Covenant. Developer agrees and covenants to the City that, prior to any foreclosure of the Property by a lender providing Lender Financing, the provisions of that certain Regulatory Agreement executed by Developer and DPD as of the date hereof shall govern the terms of Developer's obligation to provide affordable housing. Following foreclosure, if any, and from the date of such foreclosure through the Term of the Agreement, the following provisions shall govern the terms of the obligation to provide affordable housing under this Agreement:
- (a) The Facility shall be operated and maintained solely as residential rental housing for Low Income Families (as defined below) [[which Facility includes commercial space as set forth in the recitals herein]];
- (b) [[[Except for residential units for market rate tenants as agreed upon by the City in the final construction approval]]], all of the units in the Facility shall be available for occupancy to and be occupied solely by families qualifying as Low Income Families (as defined below) upon initial occupancy; and
- (c) All of the units in the Facility for Low Income Families has monthly rents not in excess of thirty percent (30%) of the maximum allowable income for a Low Income Family (with the applicable Family size for such units determined in accordance with the rules specified in Section 42(g)(2) of the Internal Revenue Code of 1986, as amended); provided, however, that for any unit occupied by a Family (as defined below) that no longer qualifies as a Low Income Family due to an increase in such Family's income since the date of its initial occupancy of such unit, the maximum monthly rent for such unit shall not exceed thirty percent (30%) of such Family's monthly income.
 - (d) As used in this Section 8.25, the following terms has the following meanings:
 - (i). "Family" shall mean one or more individuals, whether or not related by blood or marriage; and
 - (ii). "Low Income Families" shall mean Families whose annual income does not exceed [[[sixty (60%)]]] of the Chicago-area median income, adjusted for Family size, as such annual income and Chicago-area median income are determined from time to time by

the United States Department of Housing and Urban Development, and thereafter such income limits shall apply to this definition.

- (e) The covenants set forth in this <u>Section 8.25</u> shall run with the land and be binding upon any transferee.
- (f) The City and Developer may enter into a separate agreement to implement the provisions of this <u>Section 8.25.</u>]
- **[[**[8.25 <u>Job Readiness Program</u>. Developer shall undertake a job readiness program, as described in [<u>Exhibit</u>] hereto, to work with the City, through the Mayor's Office of Workforce Development, to participate in job training programs to provide job applicants for the jobs created by the Project and the Developer's operation of the Facility.]]]
- 8.26. Compliance with Multi-Project Labor Agreement. Developer shall cause the General Contractor to comply with that certain Settlement Agreement dated November 3, 2011, by and among the City, the Chicago Regional Council of Carpenters, the Metropolitan Pier and Exposition Authority, the Public Building Commission of the City, and the State, because the Project budget is in excess of \$25,000,000, and, therefore, is subject to the provisions of that certain City Multi-Project Labor Agreement (the "MPLA") dated February 9, 2011, by and among the City and the labor organizations comprising the Chicago & Cook County Building & Construction Trades Council. Developer shall cause the General Contractor to comply with the MPLA to the fullest extent legally permissible without violating other requirements applicable to the construction of the Project, including, without limitation, Sections 8.09, 10.02 and 10.03 hereof. At the direction of DPD, affidavits and other supporting documentation shall be required of Developer, General Contractor and the subcontractors to verify or clarify compliance with the MPLA.
- [[[- this section may be removed upon agreement of the parties [8.27] Continuing Information Agreement. The Developer covenants and agrees as follows:
- (a) For each tax collection year following the date the Certificate is issued pursuant to <u>Section 7.01</u> hereof, the Developer shall engage a recognized financial consultant to prepare an annual continuing information report (the "Consultant's Report") that sets forth as of its date the following information:
- (i) a description of the Redevelopment Area, including all redevelopment agreements relating to the Redevelopment Area;
 - (ii) a description of the Project;
- (iii) a status update of the Project, including information on construction, occupancy, leasing or sales, as applicable;
 - (iv) a listing of the top ten taxpayers in the Redevelopment Area; and
- (v) for the property within the Redevelopment Area from which incremental taxes will be used to pay the City Note(s), by tax codes or PINs, as applicable, for the last five collection years:
 - (A) the equalized assessed value;

- (B) the initial equalized assessed value;
- (C) the incremental equalized assessed value;
- (D) the composite tax rates;
- (E) the available incremental taxes:
- (F) the debt service of the City Note(s); and
- (G) the debt service coverage ratio.
- (b) The Consultant's Report shall be signed by the consultant and available to the Developer no later than February 1 following each tax collection year. If the Developer fails to obtain the Consultant's report by the date required, DPD may engage its own financial consultant to prepare the report and the cost thereof shall be reimbursed to the City from monies available in the TIF Fund.
 - (c) Each Consultant's Report shall include in a prominent place the following disclaimer:

"The City's Office of Budget and Management ("OBM") produces five year District Projection Reports for each TIF district in the City for the purpose of evaluating resources and project balances. This information is used by the OBM to determine how much funding has been committed and how much funding is available for potential projects. The reports and the projections including therein are not audited and do not represent a final accounting of funds. The reports are not prepared for investors or as a basis for making investment decisions with respect to any notes, bonds or other debt obligations of the City that are payable from available incremental taxes. Investors in such obligations are cautioned not to rely on any of the information contained in the District Projection Reports."

- (d) If any Consultant's Report contains projections, the consultant shall consider the publicly available TIF-wide projections from the OBM and provide an explanation as to the discrepancy, if any, between the consultant's projections and those of OBM. In addition, the consultant shall state in the Consultants Report all assumptions used in formulating the projections.
- (e) The Developer shall cause the following documents to be disseminated through the Electronic Municipal Market Access ("EMMA") system for municipal securities disclosure maintained by the Municipal Securities Rulemaking Board (the "MSRB") or through any other electronic format or system prescribed by the MSRB for purposes of Rule 15c2-12 under the Securities Exchange Act of 1934, as amended:
 - (i) Within five (5) days of the issuance of the Certificate:
 - (A) The Redevelopment Agreement, including the specimen of the City Note;
 - (B) The original feasibility study delivered to DPD; and
 - (C) On the date of the closing of any third party sale of the City Note or any beneficial interests therein by the Developer: the private placement or limited offering memorandum used by or on behalf of the Developer in connection with the sale.

- (ii) By February 15 of each year:
- (A) All redevelopment agreements with respect to the Redevelopment Area that have been made publicly available by DPD;
- (B) Annual financial statements of the Redevelopment Area that have been made publicly available by DPD; and
 - (C) The Consultant's Report.
- (f) The Developer shall deliver to DPD each of the documents set forth in subparagraph (e) above on the date such documents are disseminated through EMMA.
- (g) The Developer shall be responsible to pay or make provision for all costs and expenses relating to Developer's obligations hereunder, including but not limited to the fees and expenses of the Consultant and the costs of filing with EMMA. In addition to and not in limitation of the other provisions of this Agreement, Developer agrees to pay upon demand the City's out-of-pocket expenses, including attorney's fees, incurred in connection with the enforcement of the provisions of this Agreement. This includes, subject to any limits under applicable law, attorney's fees and legal expenses.]]]
- 8.28. <u>Survival of Covenants</u>. All warranties, representations, covenants and agreements of Developer contained in this <u>Section 8</u> and elsewhere in this Agreement shall be true, accurate and complete at the time of Developer's execution of this Agreement, and shall survive the execution, delivery and acceptance hereof by the parties hereto and (except as provided in <u>Section 7</u> hereof upon the issuance of a Certificate) shall be in effect throughout the Term of the Agreement.

SECTION 9. COVENANTS/REPRESENTATIONS/WARRANTIES OF CITY

- 9.01. <u>General Covenants</u>. The City represents that it has the authority as a home rule unit of local government to execute and deliver this Agreement and to perform its obligations hereunder.
- 9.02. <u>Survival of Covenants</u>. All warranties, representations, and covenants of the City contained in this <u>Section 9</u> or elsewhere in this Agreement shall be true, accurate, and complete at the time of the City's execution of this Agreement, and shall survive the execution, delivery and acceptance hereof by the parties hereto and be in effect throughout the Term of the Agreement.

SECTION 10. DEVELOPER'S EMPLOYMENT OBLIGATIONS

- 10.01. Employment Opportunity. Developer, on behalf of itself and its successors and assigns, hereby agrees, and shall contractually obligate its or their various contractors, subcontractors or any Affiliate of Developer operating on the Property (collectively, with Developer, the "Employers" and individually an AEmployer") to agree, that for the Term of this Agreement with respect to Developer and during the period of any other party's provision of services in connection with the construction of the Project or occupation of the Property:
- (a) No Employer shall discriminate against any employee or applicant for employment based upon race, religion, color, sex, national origin or ancestry, age, handicap or disability, sexual orientation, military discharge status, marital status, parental status or source of income as defined

in the City of Chicago Human Rights Ordinance, Chapter 2-160, Section 2-160-010 et seq., Municipal Code, except as otherwise provided by said ordinance and as amended from time to time (the "Human Rights Ordinance"). Each Employer shall take affirmative action to ensure that applicants are hired and employed without discrimination based upon race, religion, color, sex, national origin or ancestry, age, handicap or disability, sexual orientation, military discharge status, marital status, parental status or source of income and are treated in a non-discriminatory manner with regard to all job-related matters, including without limitation: employment, upgrading, demotion or transfer; recruitment or recruitment advertising; layoff or termination; rates of pay or other forms of compensation; and selection for training, including apprenticeship. Each Employer agrees to post in conspicuous places, available to employees and applicants for employment, notices to be provided by the City setting forth the provisions of this nondiscrimination clause. In addition, the Employers, in all solicitations or advertisements for employees, shall state that all qualified applicants shall receive consideration for employment without discrimination based upon race, religion, color, sex, national origin or ancestry, age, handicap or disability, sexual orientation, military discharge status, marital status, parental status or source of income.

- (b) To the greatest extent feasible, each Employer is required to present opportunities for training and employment of low- and moderate-income residents of the City and preferably of the Redevelopment Area; and to provide that contracts for work in connection with the construction of the Project be awarded to business concerns that are located in, or owned in substantial part by persons residing in, the City and preferably in the Redevelopment Area.
- (c) Each Employer shall comply with all federal, state and local equal employment and affirmative action statutes, rules and regulations, including but not limited to the City's Human Rights Ordinance and the Illinois Human Rights Act, 775 ILCS 5/1-101 et seq. (1993), and any subsequent amendments and regulations promulgated thereto.
- (d) Each Employer, in order to demonstrate compliance with the terms of this Section, shall cooperate with and promptly and accurately respond to inquiries by the City, which has the responsibility to observe and report compliance with equal employment opportunity regulations of federal, state and municipal agencies.
- (e) Each Employer shall include the foregoing provisions of subparagraphs (a) through (d) in every contract entered into in connection with the Project, and shall require inclusion of these provisions in every subcontract entered into by any subcontractors, and every agreement with any Affiliate operating on the Property, so that each such provision shall be binding upon each contractor, subcontractor or Affiliate, as the case may be.
- (f) Failure to comply with the employment obligations described in this <u>Section 10.01</u> shall be a basis for the City to pursue remedies under the provisions of <u>Section 15.02</u> hereof.
- 10.02 <u>City Resident Construction Worker Employment Requirement</u>. Developer agrees for itself and its successors and assigns, and shall contractually obligate its General Contractor and shall cause the General Contractor to contractually obligate its subcontractors, as applicable, to agree, that during the construction of the Project they shall comply with the minimum percentage of total worker hours performed by actual residents of the City as specified in Section 2-92-330 of the Municipal Code (at least 50 percent of the total worker hours worked by persons on the site of the Project shall be performed by actual residents of the City); <u>provided</u>, <u>however</u>, that in addition to complying with this percentage, Developer, its General Contractor and each subcontractor shall be

required to make good faith efforts to utilize qualified residents of the City in both unskilled and skilled labor positions.

Developer may request a reduction or waiver of this minimum percentage level of Chicagoans as provided for in Section 2-92-330 of the Municipal Code in accordance with standards and procedures developed by the Chief Procurement Officer of the City.

"Actual residents of the City" shall mean persons domiciled within the City. The domicile is an individual's one and only true, fixed and permanent home and principal establishment.

Developer, the General Contractor and each subcontractor shall provide for the maintenance of adequate employee residency records to show that actual Chicago residents are employed on the Project. Each Employer shall maintain copies of personal documents supportive of every Chicago employee's actual record of residence.

Weekly certified payroll reports (U.S. Department of Labor Form WH-347 or equivalent) shall be submitted to the Commissioner of DPD in triplicate, which shall identify clearly the actual residence of every employee on each submitted certified payroll. The first time that an employee's name appears on a payroll, the date that the Employer hired the employee should be written in after the employee's name.

Developer, the General Contractor and each subcontractor shall provide full access to their employment records to the Chief Procurement Officer, the Commissioner of DPD, the Superintendent of the Chicago Police Department, the Inspector General or any duly authorized representative of any of them. Developer, the General Contractor and each subcontractor shall maintain all relevant personnel data and records for a period of at least three (3) years after final acceptance of the work constituting the Project.

At the direction of DPD, affidavits and other supporting documentation will be required of Developer, the General Contractor and each subcontractor to verify or clarify an employee's actual address when doubt or lack of clarity has arisen.

Good faith efforts on the part of Developer, the General Contractor and each subcontractor to provide utilization of actual Chicago residents (but not sufficient for the granting of a waiver request as provided for in the standards and procedures developed by the Chief Procurement Officer) shall not suffice to replace the actual, verified achievement of the requirements of this Section concerning the worker hours performed by actual Chicago residents.

When work at the Project is completed, in the event that the City has determined that Developer has failed to ensure the fulfillment of the requirement of this Section concerning the worker hours performed by actual Chicago residents or failed to report in the manner as indicated above, the City will thereby be damaged in the failure to provide the benefit of demonstrable employment to Chicagoans to the degree stipulated in this Section. Therefore, in such a case of non-compliance, it is agreed that 1/20 of 1 percent (0.0005) of the aggregate hard construction costs set forth in the Project budget (the product of .0005 x such aggregate hard construction costs) (as the same shall be evidenced by approved contract value for the actual contracts) shall be surrendered by Developer to the City in payment for each percentage of shortfall toward the stipulated residency requirement. Failure to report the residency of employees entirely and correctly shall result in the surrender of the entire liquidated damages as if no Chicago residents were employed in either of the categories. The willful falsification of statements and the certification

of payroll data may subject Developer, the General Contractor and/or the subcontractors to prosecution. Any retainage to cover contract performance that may become due to Developer pursuant to Section 2-92-250 of the Municipal Code may be withheld by the City pending the Chief Procurement Officer's determination as to whether Developer must surrender damages as provided in this paragraph.

Nothing herein provided shall be construed to be a limitation upon the Notice of Requirements for Affirmative Action to Ensure Equal Employment Opportunity, Executive Order 11246 " and Standard Federal Equal Employment Opportunity, Executive Order 11246," or other affirmative action required for equal opportunity under the provisions of this Agreement or related documents.

Developer shall cause or require the provisions of this <u>Section 10.02</u> to be included in all construction contracts and subcontracts related to the Project.

- 10.03. MBE/WBE Commitment. Developer agrees for itself and its successors and assigns, and, if necessary to meet the requirements set forth herein, shall contractually obligate the General Contractor to agree that during the Project:
- (a) Consistent with the findings which support, as applicable, (i) the Minority-Owned and Women-Owned Business Enterprise Procurement Program, Section 2-92-420 et seq., Municipal Code (the "Procurement Program"), and (ii) the Minority- and Women-Owned Business Enterprise Construction Program, Section 2-92-650 et seq., Municipal Code (the "Construction Program," and collectively with the Procurement Program, the "MBE/WBE Program"), and in reliance upon the provisions of the MBE/WBE Program to the extent contained in, [and as qualified by, the provisions of this Section 10.03, during the course of the Project, at least the following percentages of the MBE/WBE Budget (as set forth in Exhibit H-2 hereto) shall be expended for contract participation by MBEs and by WBEs:
 - (1) At least 26 percent by MBEs.
 - (2) At least six percent by WBEs.
- (b) For purposes of this Section 10.03 only, Developer (and any party to whom a contract is let by Developer in connection with the Project) shall be deemed a "contractor" and this Agreement (and any contract let by Developer in connection with the Project) shall be deemed a Acontract" or a Aconstruction contract" as such terms are defined in Sections 2-92-420 and 2-92-670, Municipal Code, as applicable.
- (c) Consistent with Sections 2-92-440 and 2-92-720, Municipal Code, Developer's MBE/WBE commitment may be achieved in part by Developer's status as an MBE or WBE (but only to the extent of any actual work performed on the Project by Developer) or by a joint venture with one or more MBEs or WBEs (but only to the extent of the lesser of (i) the MBE or WBE participation in such joint venture or (ii) the amount of any actual work performed on the Project by the MBE or WBE), by Developer utilizing a MBE or a WBE as the General Contractor (but only to the extent of any actual work performed on the Project by the General Contractor), by subcontracting or causing the General Contractor to subcontract a portion of the Project to one or more MBEs or WBEs, or by the purchase of materials or services used in the Project from one or more MBEs or WBEs, or by any combination of the foregoing. Those entities which constitute both a MBE and a WBE shall not be credited more than once with regard to Developer's MBE/WBE commitment as described in this Section 10 03. In accordance with Section 2-92-730, Municipal Code, Developer shall not

substitute any MBE or WBE General Contractor or subcontractor without the prior written approval of DPD.

- (d) Developer shall deliver quarterly reports to the City's monitoring staff during the Project describing its efforts to achieve compliance with this MBE/WBE commitment. Such reports shall include, inter alia, the name and business address of each MBE and WBE solicited by Developer or the General Contractor to work on the Project, and the responses received from such solicitation, the name and business address of each MBE or WBE actually involved in the Project, a description of the work performed or products or services supplied, the date and amount of such work, product or service, and such other information as may assist the City's monitoring staff in determining Developer's compliance with this MBE/WBE commitment. Developer shall maintain records of all relevant data with respect to the utilization of MBEs and WBEs in connection with the Project for at least five years after completion of the Project, and the City's monitoring staff shall have access to all such records maintained by Developer, on five Business Days' notice, to allow the City to review Developer's compliance with its commitment to MBE/WBE participation and the status of any MBE or WBE performing any portion of the Project.
- (e) Upon the disqualification of any MBE or WBE General Contractor or subcontractor, if such status was misrepresented by the disqualified party, Developer shall be obligated to discharge or cause to be discharged the disqualified General Contractor or subcontractor, and, if possible, identify and engage a qualified MBE or WBE as a replacement. For purposes of this subsection (e), the disqualification procedures are further described in Sections 2-92-540 and 2-92-730, Municipal Code, as applicable.
- (f) Any reduction or waiver of Developer's MBE/WBE commitment as described in this Section 10.03 shall be undertaken in accordance with Sections 2-92-450 and 2-92-730, Municipal Code, as applicable.
- (g) Prior to the commencement of the Project, Developer shall be required to meet with the City's monitoring staff with regard to Developer's compliance with its obligations under this Section 10.03. The General Contractor and all major subcontractors shall be required to attend this preconstruction meeting. During said meeting, Developer shall demonstrate to the City's monitoring staff its plan to achieve its obligations under this Section 10.03, the sufficiency of which shall be approved by the City's monitoring staff. During the Project, Developer shall submit the documentation required by this Section 10.03 to the City's monitoring staff, including the following: (i) subcontractor's activity report; (ii) contractor's certification concerning labor standards and prevailing wage requirements; (iii) contractor letter of understanding; (iv) monthly utilization report; (v) authorization for payroll agent; (vi) certified payroll; (vii) evidence that MBE/WBE contractor associations have been informed of the Project via written notice and hearings; and (viii) evidence of compliance with job creation/job retention requirements. Failure to submit such documentation on a timely basis, or a determination by the City's monitoring staff, upon analysis of the documentation, that Developer is not complying with its obligations under this Section 10.03, shall, upon the delivery of written notice to Developer, be deemed an Event of Default. Upon the occurrence of any such Event of Default, in addition to any other remedies provided in this Agreement, the City may: (1) issue a written demand to Developer to halt the Project, (2) withhold any further payment of any City Funds to Developer or the General Contractor, or (3) seek any other remedies against Developer available at law or in equity.

SECTION 11. ENVIRONMENTAL MATTERS

- Developer has obtained a Phase I environmental site assessment of the Property in accordance with the requirements of the ASTM E1527-13 standard, and has furnished a copy of the Phase I report to AIS. If the Phase I report identifies any recognized environmental conditions ("RECs"), Developer shall perform a Phase II environmental site assessment of the Property for the purpose of determining whether any environmental or health risks would be associated with the development of the Project. If the Phase II report discloses the presence of contaminants exceeding TACO Tier I residential remediation objectives on or under the Property, Developer shall enroll the Property in the IEPA's SRP Program and take all necessary steps to obtain a letter from the IEPA approving a Remedial Action Plan ("RAP Approval Letter"). Unless AIS determines, in its sole discretion, that it is not necessary to enroll the Property in the SRP, Developer acknowledges and agrees that it may not commence construction on the Property, and the City will not make any payments to Developer of City Funds, until the IEPA issues and AIS approves the RAP Approval Letter.
- 11.02 Environmental Remediation. If the Property does not meet TACO Tier I residential remediation objectives as determined pursuant to Section 11.01 above and, following their review of the Phase II, the City provides the first installment of the City Funds and Developer maintains its intent to undertake the Project, the Developer shall complete all Environmental Remediation Work necessary to obtain a Final NFR Letter, including, without limitation, preparing and submitting a Remedial Action Completion Report to the IEPA. The Developer shall continuously and diligently pursue the Final NFR Letter using all reasonable means. The Developer acknowledges and agrees that the City will not issue a Certificate until the IEPA has issued, and DOE has approved, a Final NFR Letter for the Property, unless DOE has previously determined that it was not necessary to enroll the Property in the SRP. The City shall have the right to approve any changes or modifications to the Remediation Objectives Report, Remedial Action Plan, Remedial Action Completion Report or other documents submitted to the IEPA in connection with the Draft NFR Letter or Final NFR Letter, which approval shall not be unreasonably withheld. The Developer shall bear sole responsibility for completing all aspects of the Environmental Remediation Work and any other investigative and cleanup costs associated with the Property and any improvements, facilities or operations located or formerly located thereon, including, without limitation, the removal and disposal of all Hazardous Substances, debris and other materials excavated during the performance of the Environmental Remediation Work or construction of the Project. The Developer shall promptly transmit to the City copies of any written communications received from the IEPA or other regulatory agencies with respect to the Environmental Remediation Work.
- 11.03 AAS IS@ SALE. THE DEVELOPER ACKNOWLEDGES THAT IT HAS HAD ADEQUATE OPPORTUNITY TO INSPECT AND EVALUATE THE STRUCTURAL, PHYSICAL AND ENVIRONMENTAL CONDITION AND RISKS OF THE PROPERTY AND ACCEPTS THE RISK THAT ANY INSPECTION MAY NOT DISCLOSE ALL MATERIAL MATTERS AFFECTING THE PROPERTY. THE DEVELOPER AGREES TO ACCEPT THE PROPERTY IN ITS AAS IS,@ AWHERE IS@ AND AWITH ALL FAULTS@ CONDITION AT CLOSING WITHOUT ANY COVENANT, REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, OF ANY KIND, AS TO THE STRUCTURAL, PHYSICAL OR ENVIRONMENTAL CONDITION OF THE PROPERTY OR THE SUITABILITY OF THE PROPERTY FOR ANY PURPOSE WHATSOEVER. THE DEVELOPER ACKNOWLEDGES THAT IT IS RELYING SOLELY UPON ITS OWN INSPECTION AND OTHER DUE DILIGENCE ACTIVITIES AND NOT UPON ANY INFORMATION (INCLUDING, WITHOUT LIMITATION, ENVIRONMENTAL STUDIES OR REPORTS OF ANY KIND) PROVIDED BY OR ON BEHALF OF THE CITY OR ITS AGENTS OR EMPLOYEES WITH RESPECT

THERETO. THE DEVELOPER AGREES THAT IT IS THE DEVELOPER=S SOLE RESPONSIBILITY AND OBLIGATION TO PERFORM ANY ENVIRONMENTAL REMEDIATION WORK AND TAKE SUCH OTHER ACTION AS IS NECESSARY TO PUT THE PROPERTY IN A CONDITION WHICH IS SUITABLE FOR ITS INTENDED USE.

- 11.04 Release and Indemnification. Without limiting any other provisions hereof, the Developer, for itself and its successors and assigns, hereby completely and forever waives, releases and discharges the Indemnitees from and against any and all Losses, whether direct or indirect, known or unknown, foreseen or unforeseen, now existing or occurring after the Closing Date, and regardless of whether caused by or within the control of the Developer, based upon, arising out of, or related to: (a) Developer=s failure to perform the Environmental Remediation Work (if applicable); (b) any environmental contamination, pollution or hazards associated with all or any portion of the Property or any improvements, facilities or operations located or formerly located thereon, including, without limitation, the presence or suspected presence of Hazardous Substances in, on, under or about the Property, or the escape, seepage, leakage, spillage, release, emission, discharge, generation, transportation, treatment, storage or disposal of Hazardous Substances associated with all or any portion of the Property, or threatened release, emission or discharge of Hazardous Substances from all or any portion of the Property; (c) the structural, physical or environmental condition of the Property; and (d) any violation of, compliance with, enforcement of or liability under any Environmental Laws, including, without limitation, any Losses arising under CERCLA, and (e) any investigation, cleanup, monitoring, remedial, removal or restoration work required by any federal, state or local governmental agency or political subdivision or other third party in connection or associated with the Property or any improvements, facilities or operations located or formerly located thereon (collectively, "Released Claims"). Furthermore, the Developer shall defend, indemnify, and hold the Indemnitees harmless from and against any and all Losses which may be made or asserted by any third parties arising out of or in any way connected with, directly or indirectly, any of the Released Claims.
- 11.05 Release Runs with the Land. The covenant of release in Section 11.04 shall run with the Property, and shall be binding upon all successors and assigns of the Developer with respect to the Property, including, without limitation, each and every person, firm, corporation, limited liability company, trust or other entity owning, leasing, occupying, using or possessing any portion of the Property under or through the Developer following the date of the Deed. The Developer acknowledges and agrees that the foregoing covenant of release constitutes a material inducement to the City to enter into this Agreement, and that, but for such release, the City would not have agreed to convey the Property to MHL. It is expressly agreed and understood by and between the Developer and the City that, should any future obligation of the Developer, or its successors or assigns, arise or be alleged to arise in connection with any environmental, soil or other condition of the Property, neither the Developer, nor its successors or assigns, will assert that those obligations must be satisfied in whole or in part by the City because Section 11.04 contains a full, complete and final release of all such claims.
- 11.06 <u>Survival</u>. This **Section 11** shall survive the Closing or any termination of this Agreement (regardless of the reason for such termination).

Developer hereby represents and warrants to the City that Developer has conducted environmental studies sufficient to conclude that the Project may be constructed, completed and operated in accordance with all Environmental Laws and this Agreement and all Exhibits attached hereto, the Scope Drawings, Plans and Specifications and all amendments thereto, [the Bond Ordinance] and the Redevelopment Plan.

Without limiting any other provisions hereof, Developer agrees to indemnify, defend and hold the City harmless from and against any and all losses, liabilities, damages, injuries, costs, expenses or claims of any kind whatsoever including, without limitation, any losses, liabilities, damages, injuries, costs, expenses or claims asserted or arising under any Environmental Laws incurred, suffered by or asserted against the City as a direct or indirect result of any of the following, regardless of whether or not caused by, or within the control of Developer: (i) the presence of any Hazardous Material on or under, or the escape, seepage, leakage, spillage, emission, discharge or release of any Hazardous Material from (A) all or any portion of the Property or (B) any other real property in which Developer, or any person directly or indirectly controlling, controlled by or under common control with Developer, holds any estate or interest whatsoever (including, without limitation, any property owned by a land trust in which the beneficial interest is owned, in whole or in part, by Developer), or (ii) any liens against the Property permitted or imposed by any Environmental Laws, or any actual or asserted liability or obligation of the City or Developer or any of its Affiliates under any Environmental Laws relating to the Property.

SECTION 12. INSURANCE

Developer must provide and maintain, at Developer's own expense, or cause to be provided and maintained during the term of this Agreement, the insurance coverage and requirements specified below, insuring all operations related to the Agreement.

- (a) Prior to execution and delivery of this Agreement.
 - (i) Workers Compensation and Employers Liability

Workers Compensation Insurance, as prescribed by applicable law covering all employees who are to provide work under this Agreement and Employers Liability coverage with limits of not less than \$100,000 each accident, illness or disease.

(ii) Commercial General Liability (Primary and Umbrella)

Commercial General Liability Insurance or equivalent with limits of not less than \$1,000,000 per occurrence for bodily injury, personal injury, and property damage liability. Coverages must include the following: All premises and operations, products/completed operations independent contractors, separation of insureds, defense, and contractual liability (with no limitation endorsement). The City of Chicago is to be named as an additional insured on a primary, non-contributory basis for any liability arising directly or indirectly from the work.

(iii) All Risk Property

All Risk Property Insurance at replacement value of the property to protect against loss of, damage to, or destruction of the building/facility. The City is to be named as an additional insured and loss payee/mortgagee if applicable.

(b) <u>Construction</u>. Prior to the construction of any portion of the Project, Developer will cause its architects, contractors, subcontractors, project managers and other parties constructing the Project to procure and maintain the following kinds and amounts of insurance.

(i) Workers Compensation and Employers Liability

Workers Compensation Insurance, as prescribed by applicable law covering all employees who are to provide work under this Agreement and Employers Liability coverage with limits of not less than \$500,000 each accident, illness or disease.

(ii) Commercial General Liability (Primary and Umbrella)

Commercial General Liability Insurance or equivalent with limits of not less than \$2,000,000 per occurrence for bodily injury, personal injury, and property damage liability. Coverages must include the following: All premises and operations, products/completed operations (for a minimum of two (2) years following project completion), explosion, collapse, underground, separation of insureds, defense, and contractual liability (with no limitation endorsement). The City of Chicago is to be named as an additional insured on a primary, non-contributory basis for any liability arising directly or indirectly from the work.

(iii) Automobile Liability (Primary and Umbrella)

When any motor vehicles (owned, non-owned and hired) are used in connection with work to be performed, the Automobile Liability Insurance with limits of not less than \$2,000,000 per occurrence for bodily injury and property damage. The City of Chicago is to be named as an additional insured on a primary, non-contributory basis.

(iv) Railroad Protective Liability

When any work is to be done adjacent to or on railroad or transit property, Developer must provide cause to be provided with respect to the operations that Contractors perform, Railroad Protective Liability Insurance in the name of railroad or transit entity. The policy must have limits of not less than \$2,000,000 per occurrence and \$6,000,000 in the aggregate for losses arising out of injuries to or death of all persons, and for damage to or destruction of property, including the loss of use thereof.

(v) All Risk /Builders Risk

When Developer undertakes any construction, including improvements, betterments, and/or repairs, Developer must provide or cause to be provided All Risk Builders Risk Insurance at replacement cost for materials, supplies, equipment, machinery and fixtures that are or will be part of the project. The City of Chicago is to be named as an additional insured and loss payee/mortgagee if applicable.

(vi) Professional Liability

When any architects, engineers, construction managers or other professional consultants perform work in connection with this Agreement, Professional Liability Insurance covering acts, errors, or omissions must be maintained with limits of not less than \$\frac{1,000,000}{2,000}\$. Coverage must include contractual liability. When policies are renewed or replaced, the policy retroactive date must coincide with, or precede, start of work on the Contract. A claims-made policy which is not renewed or replaced must have an extended reporting period of two (2) years.

(vii) Valuable Papers

When any plans, designs, drawings, specifications and documents are produced or used under this Agreement, Valuable Papers Insurance must be maintained in an amount to insure against any loss whatsoever, and must have limits sufficient to pay for the re-creation and reconstruction of such records.

(viii) Contractors Pollution Liability

When any remediation work is performed which may cause a pollution exposure, Developer must cause remediation contractor to provide Contractor Pollution Liability covering bodily injury, property damage and other losses caused by pollution conditions that arise from the contract scope of work with limits of not less than \$1,000,000 per occurrence. Coverage must include completed operations, contractual liability, defense, excavation, environmental cleanup, remediation and disposal. When policies are renewed or replaced, the policy retroactive date must coincide with or precede, start of work on the Agreement. A claimsmade policy which is not renewed or replaced must have an extended reporting period of two (2) years. The City of Chicago is to be named as an additional insured.

(c) Post Construction:

(i) All Risk Property Insurance at replacement value of the property to protect against loss of, damage to, or destruction of the building/facility. The City is to be named as an additional insured and loss payee/mortgagee if applicable.

(d) Other Requirements:

Developer must furnish the City of Chicago, Department of Planning and Development, City Hall, Room 1000, 121 North LaSalle Street 60602, original Certificates of Insurance, or such similar evidence, to be in force on the date of this Agreement, and Renewal Certificates of Insurance, or such similar evidence, if the coverages have an expiration or renewal date occurring during the term of this Agreement. Developer must submit evidence of insurance on the City of Chicago Insurance Certificate Form (copy attached) or equivalent prior to closing. The receipt of any certificate does not constitute agreement by the City that the insurance requirements in the Agreement have been fully met or that the insurance policies indicated on the certificate are in compliance with all Agreement requirements. The failure of the City to obtain certificates or other insurance evidence from Developer is not a waiver by the City of any requirements for Developer to obtain and maintain the specified coverages. Developer shall advise all insurers of the Agreement provisions regarding insurance. Non-conforming insurance does not relieve Developer of the obligation to provide insurance as specified herein. Nonfulfillment of the insurance conditions may constitute a violation of the Agreement, and the City retains the right to stop work and/or terminate agreement until proper evidence of insurance is provided.

The insurance must provide for 60 days prior written notice to be given to the City in the event coverage is substantially changed, canceled, or non-renewed.

Any deductibles or self insured retentions on referenced insurance coverages must be borne by Developer and Contractors.

Developer hereby waives and agrees to require their insurers to waive their rights of subrogation against the City of Chicago, its employees, elected officials, agents, or representatives.

The coverages and limits furnished by Developer in no way limit Developer's liabilities and responsibilities specified within the Agreement or by law.

Any insurance or self insurance programs maintained by the City of Chicago do not contribute with insurance provided by Developer under the Agreement.

The required insurance to be carried is not limited by any limitations expressed in the indemnification language in this Agreement or any limitation placed on the indemnity in this Agreement given as a matter of law.

If Developer is a joint venture or limited liability company, the insurance policies must name the joint venture or limited liability company as a named insured.

Developer must require Contractor and subcontractors to provide the insurance required herein, or Developer may provide the coverages for Contractor and subcontractors. All Contractors and subcontractors are subject to the same insurance requirements of Developer unless otherwise specified in this Agreement.

If Developer, any Contractor or subcontractor desires additional coverages, the party desiring the additional coverages is responsible for the acquisition and cost.

The City of Chicago Risk Management Department maintains the right to modify, delete, alter or change these requirements.

SECTION 13. INDEMNIFICATION

- 13.01. General Indemnity. Developer agrees to indemnify, pay, defend and hold the City, and its elected and appointed officials, employees, agents and affiliates (individually an "Indemnitee," and collectively the "Indemnitees") harmless from and against, any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, claims, costs, expenses and disbursements of any kind or nature whatsoever (and including without limitation, the reasonable fees and disbursements of counsel for such Indemnitees in connection with any investigative, administrative or judicial proceeding commenced or threatened, whether or not such Indemnitees shall be designated a party thereto), that may be imposed on, suffered, incurred by or asserted against the Indemnitees in any manner relating or arising out of:
 - (i) Developer's failure to comply with any of the terms, covenants and conditions contained within this Agreement, including, be not limited to. Section 8.27; or
 - (ii) Developer's or any contractor's failure to pay General Contractors, subcontractors or materialmen in connection with the TIF-Funded Improvements or any other Project improvement; or
 - (iii) the existence of any material misrepresentation or omission in this Agreement, any official statement, limited offering memorandum or private placement memorandum or the Redevelopment Plan or any other document related to this Agreement that is the result

of information supplied or omitted by Developer or any Affiliate Developer or any agents, employees, contractors or persons acting under the control or at the request of Developer or any Affiliate of Developer; or

(iv) Developer's failure to cure any misrepresentation in this Agreement or any other agreement relating hereto;

provided, however, that Developer shall have no obligation to an Indemnitee arising from the wanton or willful misconduct of that Indemnitee. To the extent that the preceding sentence may be unenforceable because it is violative of any law or public policy, Developer shall contribute the maximum portion that it is permitted to pay and satisfy under the applicable law, to the payment and satisfaction of all indemnified liabilities incurred by the Indemnitees or any of them. The provisions of the undertakings and indemnification set out in this Section 13.01 shall survive the termination of this Agreement.

SECTION 14. MAINTAINING RECORDS/RIGHT TO INSPECT

- 14.01. <u>Books and Records</u>. Developer shall keep and maintain separate, complete, accurate and detailed books and records necessary to reflect and fully disclose the total actual cost of the Project and the disposition of all funds from whatever source allocated thereto, and to monitor the Project. All such books, records and other documents, including but not limited to Developer's loan statements, if any, General Contractors' and contractors' sworn statements, general contracts, subcontracts, purchase orders, waivers of lien, paid receipts and invoices, shall be available at Developer's offices for inspection, copying, audit and examination by an authorized representative of the City, at Developer's expense. Developer shall incorporate this right to inspect, copy, audit and examine all books and records into all contracts entered into by Developer with respect to the Project.
- 14.02. <u>Inspection Rights</u>. Upon three (3) business days' notice, any authorized representative of the City has access to all portions of the Project and the Property during normal business hours for the Term of the Agreement.

SECTION 15. DEFAULT AND REMEDIES

- 15.01. Events of Default. The occurrence of any one or more of the following events, subject to the provisions of <u>Section 15.03</u>, shall constitute an AEvent of Default" by Developer hereunder:
- (a) the failure of Developer to perform, keep or observe any of the covenants, conditions, promises, agreements or obligations of Developer under this Agreement or any related agreement;
- , (b) the failure of Developer to perform, keep or observe any of the covenants, conditions, promises, agreements or obligations of Developer under any other agreement with any person or entity if such failure may have a material adverse effect on Developer's business, property, assets, operations or condition, financial or otherwise;
- (c) the making or furnishing by Developer to the City of any representation, warranty, certificate, schedule, report or other communication within or in connection with this Agreement or any related agreement which is untrue or misleading in any material respect;

- (d) except as otherwise permitted hereunder, the creation (whether voluntary or involuntary) of, or any attempt to create, any lien or other encumbrance upon the Property, including any fixtures now or hereafter attached thereto, other than the Permitted Liens, or the making or any attempt to make any levy, seizure or attachment thereof;
- (e) the commencement of any proceedings in bankruptcy by or against Developer or for the liquidation or reorganization of Developer, or alleging that Developer is insolvent or unable to pay its debts as they mature, or for the readjustment or arrangement of Developer's debts, whether under the United States Bankruptcy Code or under any other state or federal law, now or hereafter existing for the relief of debtors, or the commencement of any analogous statutory or non-statutory proceedings involving Developer; provided, however, that if such commencement of proceedings is involuntary, such action shall not constitute an Event of Default unless such proceedings are not dismissed within sixty (60) days after the commencement of such proceedings;
- (f) the appointment of a receiver or trustee for Developer, for any substantial part of Developer's assets or the institution of any proceedings for the dissolution, or the full or partial liquidation, or the merger or consolidation, of Developer; <u>provided</u>, <u>however</u>, that if such appointment or commencement of proceedings is involuntary, such action shall not constitute an Event of Default unless such appointment is not revoked or such proceedings are not dismissed within sixty (60) days after the commencement thereof;
- (g) the entry of any judgment or order against Developer which remains unsatisfied or undischarged and in effect for sixty (60) days after such entry without a stay of enforcement or execution;
- (h) the occurrence of an event of default under the Lender Financing, which default is not cured within any applicable cure period;
- (i) the dissolution of Developer [[[or the death of any natural person who owns a material interest in Developer]]];
- (j) the institution in any court of a criminal proceeding (other than a misdemeanor) against Developer or any natural person who owns a material interest in Developer, which is not dismissed within thirty (30) days, or the indictment of Developer or any natural person who owns a material interest in Developer, for any crime (other than a misdemeanor);
- (k) prior to the [issuance of the Certificate] [expiration of the Term of the Agreement] [some other time period], the sale or transfer of [all] [a majority] [some other percentage] of the ownership interests of Developer without the prior written consent of the City; or
- (I) The failure of Developer, or the failure by any party that is a Controlling Person (defined in Section 1-23-010 of the Municipal Code) with respect to Developer, to maintain eligibility to do business with the City in violation of Section 1-23-030 of the Municipal Code; such failure shall render this Agreement voidable or subject to termination, at the option of the Chief Procurement Officer.

For purposes of <u>Sections 15.01(i)</u> and <u>15.01(j)</u> hereof, a person with a material interest in Developer shall be one owning in excess of [ten (10%)] of Developer's [issued and outstanding shares of stock][membership interests][partnership interests].

15.02. Remedies. Upon the occurrence of an Event of Default, the City may terminate this Agreement and any other agreements to which the City and Developer are or shall be parties, suspend disbursement of City Funds, place a lien on the Project in the amount of City Funds paid, [and/or] seek reimbursement of any City Funds paid [and/or draw down up to the entire balance of the Letter of Credit as set forth in this Section 15.02 below]. The City may, in any court of competent jurisdiction by any action or proceeding at law or in equity, pursue and secure any available remedy, including but not limited to damages, injunctive relief or the specific performance of the agreements contained herein. Upon the occurrence of an Event of Default under Section 8.06, Developer shall be obligated to repay to the City all previously disbursed City Funds. [In addition to other instances set forth in this Agreement, the City may draw on the Letter of Credit if Developer defaults under the Jobs Covenant and/or Operating Covenant as set forth in Section 8.06 subject, in the case of the Jobs Covenant, to the single cure period, if applicable, described in Section 8.06(c).]

15.03 Curative Period. In the event Developer shall fail to perform a monetary covenant which Developer is required to perform under this Agreement, notwithstanding any other provision of this Agreement to the contrary, an Event of Default shall not be deemed to have occurred unless Developer has failed to perform such monetary covenant within ten (10) days of its receipt of a written notice from the City specifying that it has failed to perform such monetary covenant. In the event Developer shall fail to perform a non-monetary covenant which Developer is required to perform under this Agreement, notwithstanding any other provision of this Agreement to the contrary, an Event of Default shall not be deemed to have occurred unless Developer has failed to cure such default within thirty (30) days of its receipt of a written notice from the City specifying the nature of the default; provided, however, with respect to those non-monetary defaults which are not capable of being cured within such thirty (30) day period, Developer shall not be deemed to have committed an Event of Default under this Agreement if it has commenced to cure the alleged default within such thirty (30) day period and thereafter diligently and continuously prosecutes the cure of such default until the same has been cured; and provided, further, that there shall be no cure period under this Section 15.03 with respect to Developer's failure to comply with any requirement hereunder for which an additional cure period (aside from that provided in this Section 15.03) is provided (e.g. See Section 8.06?).

SECTION 16. MORTGAGING OF THE PROJECT

All mortgages or deeds of trust in place as of the date hereof with respect to the Property or any portion thereof are listed on Exhibit G hereto (including but not limited to mortgages made prior to or on the date hereof in connection with Lender Financing) and are referred to herein as the AExisting Mortgages." Any mortgage or deed of trust that Developer may hereafter elect to execute and record or permit to be recorded against the Property or any portion thereof is referred to herein as a ANew Mortgage." Any New Mortgage that Developer may hereafter elect to execute and record or permit to be recorded against the Property or any portion thereof with the prior written consent of the City is referred to herein as a "Permitted Mortgage." It is hereby agreed by and between the City and Developer as follows:

(a) In the event that a mortgagee or any other party shall succeed to Developer's interest in the Property or any portion thereof pursuant to the exercise of remedies under a New Mortgage (other than a Permitted Mortgage), whether by foreclosure or deed in lieu of foreclosure, and in conjunction therewith accepts an assignment of Developer's interest hereunder in accordance with Section 18.15 hereof, the City may, but shall not be obligated to, attorn to and recognize such party as the successor in interest to Developer for all purposes under this

Agreement and, unless so recognized by the City as the successor in interest, such party shall be entitled to no rights or benefits under this Agreement, but such party shall be bound by those provisions of this Agreement that are covenants expressly running with the land.

- (b) In the event that any mortgagee shall succeed to Developer's interest in the Property or any portion thereof pursuant to the exercise of remedies under an Existing Mortgage or a Permitted Mortgage, whether by foreclosure or deed in lieu of foreclosure, and in conjunction therewith accepts an assignment of Developer's interest hereunder in accordance with Section 18.15 hereof, the City hereby agrees to attorn to and recognize such party as the successor in interest to Developer for all purposes under this Agreement so long as such party accepts all of the obligations and liabilities of ADeveloper" hereunder [Note: in a transaction where a City Note is issued, confirm with DPD whether the assignee can receive payments under the City Note without additional City consent]; provided, however, that, notwithstanding any other provision of this Agreement to the contrary, it is understood and agreed that if such party accepts an assignment of Developer's interest under this Agreement, such party has no liability under this Agreement for any Event of Default of Developer which accrued prior to the time such party succeeded to the interest of Developer under this Agreement, in which case Developer shall be solely responsible. However, if such mortgagee under a Permitted Mortgage or an Existing Mortgage does not expressly accept an assignment of Developer's interest hereunder, such party shall be entitled to no rights and benefits under this Agreement, and such party shall be bound only by those provisions of this Agreement, if any, which are covenants expressly running with the land.
- (c) Prior to the issuance by the City to Developer of a Certificate pursuant to <u>Section 7</u> hereof, no New Mortgage shall be executed with respect to the Property or any portion thereof without the prior written consent of the Commissioner of DPD.

SECTION 17. NOTICE

Unless otherwise specified, any notice, demand or request required hereunder shall be given in writing at the addresses set forth below, by any of the following means: (a) personal service; (b) telecopy or facsimile; (c) overnight courier, or (d) registered or certified mail, return receipt requested.

If to the City:	If to Developer:
City of Chicago Department of Planning and Development 121 North LaSalle Street, Room 1000 Chicago, Illinois 60602 Attention: Commissioner	43 rd and Calumet Phase I LP c/o The Habitat Company 350 W. Hubbard St., Suite 500 Chicago, IL 60654 Attention:
With Copies To:	With Copies To:
City of Chicago Department of Law 121 North LaSalle Street, Room 600 Chicago, Illinois 60602 Attention: Finance and Economic Development Division	Katten Muchin Rosenman LLP 525 W. Monroe St., Suite Chicago, IL 60661 Attention: David Cohen, Partner

Such addresses may be changed by notice to the other parties given in the same manner provided above. Any notice, demand, or request sent pursuant to either clause (a) or (b) hereof shall be deemed received upon such personal service or upon dispatch. Any notice, demand or request sent pursuant to clause (c) shall be deemed received on the day immediately following deposit with the overnight courier and any notices, demands or requests sent pursuant to subsection (d) shall be deemed received two (2) business days following deposit in the mail.

SECTION 18. MISCELLANEOUS

- 18.01 Amendment. This Agreement and the Exhibits attached hereto may not be amended or modified without the prior written consent of the parties hereto; provided, however, that the City, in its sole discretion, may amend, modify or supplement the Redevelopment Plan without the consent of any party hereto. It is agreed that no material amendment or change to this Agreement shall be made or be effective unless ratified or authorized by an ordinance duly adopted by the City Council. The term Amaterial" for the purpose of this Section 18.01 shall be defined as any deviation from the terms of the Agreement which operates to cancel or otherwise reduce any developmental, construction or job-creating obligations of Developer (including those set forth in Sections 10.02 and 10.03 hereof) by more than five percent (5%) or materially changes the Project site or character of the Project or any activities undertaken by Developer affecting the Project site, the Project, or both, or increases any time agreed for performance by Developer by more than [ninety (90)] days.
- 18.02 Entire Agreement. This Agreement (including each Exhibit attached hereto, which is hereby incorporated herein by reference) constitutes the entire Agreement between the parties hereto and it supersedes all prior agreements, negotiations and discussions between the parties relative to the subject matter hereof.
- 18.03 <u>Limitation of Liability</u>. No member, official or employee of the City shall be personally liable to Developer or any successor in interest in the event of any default or breach by the City or for any amount which may become due to Developer from the City or any successor in interest or on any obligation under the terms of this Agreement.
- 18.04 <u>Further Assurances</u>. Developer agrees to take such actions, including the execution and delivery of such documents, instruments, petitions and certifications as may become necessary or appropriate to carry out the terms, provisions and intent of this Agreement.
- 18.05 <u>Waiver</u>. Waiver by the City or Developer with respect to any breach of this Agreement shall not be considered or treated as a waiver of the rights of the respective party with respect to any other default or with respect to any particular default, except to the extent specifically waived by the City or Developer in writing. No delay or omission on the part of a party in exercising any right shall operate as a waiver of such right or any other right unless pursuant to the specific terms hereof. A waiver by a party of a provision of this Agreement shall not prejudice or constitute a waiver of such party's right otherwise to demand strict compliance with that provision or any other provision of this Agreement. No prior waiver by a party, nor any course of dealing between the parties hereto, shall constitute a waiver of any such parties' rights or of any obligations of any other party hereto as to any future transactions.
- 18.06 Remedies Cumulative. The remedies of a party hereunder are cumulative and the exercise of any one or more of the remedies provided for herein shall not be construed as a waiver of any other remedies of such party unless specifically so provided herein.

- 18.07 <u>Disclaimer</u>. Nothing contained in this Agreement nor any act of the City shall be deemed or construed by any of the parties, or by any third person, to create or imply any relationship of third-party beneficiary, principal or agent, limited or general partnership or joint venture, or to create or imply any association or relationship involving the City.
- 18.08 <u>Counterparts</u>. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall constitute one and the same agreement.
- 18.09 <u>Severability</u>. If any provision in this Agreement, or any paragraph, sentence, clause, phrase, word or the application thereof, in any circumstance, is held invalid, this Agreement shall be construed as if such invalid part were never included herein and the remainder of this Agreement shall be and remain valid and enforceable to the fullest extent permitted by law.
- 18.10 <u>Conflict</u>. In the event of a conflict between any provisions of this Agreement and the provisions of the TIF Ordinances and/or the [the Bond Ordinance, if any,] such ordinance(s) shall prevail and control.
- 18.11 <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the internal laws of the State of Illinois, without regard to its conflicts of law principles.
- 18.12 Form of Documents. All documents required by this Agreement to be submitted, delivered or furnished to the City shall be in form and content satisfactory to the City.
- 18.13 <u>Approval</u>. Wherever this Agreement provides for the approval or consent of the City, DPD or the Commissioner, or any matter is to be to the City's, DPD's or the Commissioner's satisfaction, unless specifically stated to the contrary, such approval, consent or satisfaction shall be made, given or determined by the City, DPD or the Commissioner in writing and in the reasonable discretion thereof. The Commissioner or other person designated by the Mayor of the City shall act for the City or DPD in making all approvals, consents and determinations of satisfaction, granting the Certificate or otherwise administering this Agreement for the City.
- 18.14 <u>Assignment</u>. Developer may not sell, assign or otherwise transfer its interest in this Agreement in whole or in part without the written consent of the City. Any successor in interest to Developer under this Agreement shall certify in writing to the City its agreement to abide by all remaining executory terms of this Agreement, including but not limited to [Sections 8.19 [Real Estate Provisions] and 8.23 (Survival of Covenants) hereof, for the Term of the Agreement. Developer consents to the City's sale, transfer, assignment or other disposal of this Agreement at any time in whole or in part.
- 18.15 <u>Binding Effect</u>. This Agreement shall be binding upon Developer, the City and their respective successors and permitted assigns (as provided herein) and shall inure to the benefit of Developer, the City and their respective successors and permitted assigns (as provided herein). Except as otherwise provided herein, this Agreement shall not run to the benefit of, or be enforceable by, any person or entity other than a party to this Agreement and its successors and permitted assigns. This Agreement should not be deemed to confer upon third parties any remedy, claim, right of reimbursement or other right.
- 18.16 <u>Force Majeure</u>. Neither the City nor Developer nor any successor in interest to either of them shall be considered in breach of or in default of its obligations under this Agreement in the event of any delay caused by damage or destruction by fire or other casualty, strike, shortage

of material, unusually adverse weather conditions such as, by way of illustration and not limitation, severe rain storms or below freezing temperatures of abnormal degree or for an abnormal duration, tornadoes or cyclones, and other events or conditions beyond the reasonable control of the party affected which in fact interferes with the ability of such party to discharge its obligations hereunder. The individual or entity relying on this section with respect to any such delay shall, upon the occurrence of the event causing such delay, immediately give written notice to the other parties to this Agreement. The individual or entity relying on this section with respect to any such delay may rely on this section only to the extent of the actual number of days of delay effected by any such events described above.

- 18.17 <u>Business Economic Support Act</u>. Pursuant to the Business Economic Support Act (30 ILCS 760/1 et seq.), if Developer is required to provide notice under the WARN Act, Developer shall, in addition to the notice required under the WARN Act, provide at the same time a copy of the WARN Act notice to the Governor of the State, the Speaker and Minority Leader of the House of Representatives of the State, the President and minority Leader of the Senate of State, and the Mayor of each municipality where Developer has locations in the State. Failure by Developer to provide such notice as described above may result in the termination of all or a part of the payment or reimbursement obligations of the City set forth herein.
- 18.18 <u>Venue and Consent to Jurisdiction</u>. If there is a lawsuit under this Agreement, each party may hereto agrees to submit to the jurisdiction of the courts of Cook County, the State of Illinois and the United States District Court for the Northern District of Illinois.
- 18.19 <u>Costs and Expenses</u>. In addition to and not in limitation of the other provisions of this Agreement, Developer agrees to pay upon demand the City's out-of-pocket expenses, including attorney's fees, incurred in connection with the enforcement of the provisions of this Agreement. This includes, subject to any limits under applicable law, attorney's fees and legal expenses, whether or not there is a lawsuit, including attorney's fees for bankruptcy proceedings (including efforts to modify or vacate any automatic stay or injunction), appeals and any anticipated post-judgment collection services. Developer also will pay any court costs, in addition to all other sums provided by law.
- 18.20 Business Relationships. Developer acknowledges (A) receipt of a copy of Section 2-156-030 (b) of the Municipal Code, (B) that Developer has read such provision and understands that pursuant to such Section 2-156-030 (b), it is illegal for any elected official of the City, or any person acting at the direction of such official, to contact, either orally or in writing, any other City official or employee with respect to any matter involving any person with whom the elected City official or employee has a business relationship that creates a "Financial Interest" (as defined in Section 2-156-010 of the Municipal Code)(a "Financial Interest"), or to participate in any discussion in any City Council committee hearing or in any City Council meeting or to vote on any matter involving any person with whom the elected City official or employee has a business relationship that creates a Financial Interest, or to participate in any discussion in any City Council committee hearing or in any City Council meeting or to vote on any matter involving the person with whom an elected official has a business relationship that creates a Financial Interest, and (C) that a violation of Section 2-156-030 (b) by an elected official, or any person acting at the direction of such official, with respect to any transaction contemplated by this Agreement shall be grounds for termination of this Agreement and the transactions contemplated hereby. Developer hereby represents and warrants that, to the best of its knowledge after due inquiry, no violation of Section 2-156-030 (b) has occurred with respect to this Agreement or the transactions contemplated hereby.

[Note: Insert the following section if a City Note is being issued]

18.21 Restrictions on Transfer. The City Note(s) may only be offered, sold, pledged or otherwise transferred in principal amounts of not less than \$100,000 and only to (a) an institutional "accredited investor" within the meaning of rule 501(a) (1), (2), (3) or (7) under the Securities Act of 1933 (the "Securities Act') that delivers to the City an Investor Letter in the form of Exhibit P hereto, or (b) a person (other than a dealer) who the seller reasonably believes is a "qualified institutional buyer" within the meaning of Rule 144A(a)(1) of the Securities Act. Any holder of the City Note(s) is required to notify any purchaser of the City Note(s) of the resale restrictions referred to above. Notwithstanding the forgoing, if any transfer of the City Note(s) is to a dealer meeting the requirements of Section 144A(a)(1)(ii) or (iii), such dealer shall deliver to the City an Investor Letter with the modifications set forth on the Exhibit P.

[THE REMAINDER OF THIS PAGE IS INTENTIONALLY LEFT BLANK.]

IN WITNESS WHEREOF, the parties hereto have caused this Redevelopment Agreement to be executed on or as of the day and year first above written.

CITY OF CHICAGO

Ву: _		
		te D. Cox, Commissioner, tment of Planning and Development
		UMET PHASE I LP, mited Partnership
	Ву:	43 rd AND CALUMET PHASE I, LLC, an Illinois limited liability company Its general partner
	Ву:	43 GREEN JV LLC, a Delaware limited liability company, its Manager and sole member
	Ву:	HABITAT 43 GREEN LLC, an Illinois limited liability company, its Managing Member
	Ву:	HABITAT ACQUISITIONS COMPANY LLC an Illinois limited liability company, its Manager
	Ву:	THE HABITAT COMPANY LLC, an Illinois limited liability company, its Manager
		By: Name: Daniel E. Levin Title: Manager
		NS HOUSING INITIATIVES, for -profit corporation
Зу:	Name: Title	
	EEN JV ware lin	LLC, nited liability company
Зу:		
	Name Title	

STATE OF ILLINOIS	
COUNTY OF COOK) SS)
I,aforesaid, DO HEREB	, a notary public in and for the said County, in the State Y CERTIFY that, personally known
of the City of Chicago (is subscribed to the acknowledged that he given to him/her by the	Y CERTIFY that, personally known
GIVEN	under my hand and official seal thisth day of,
	Notary Public
	My Commission Expires

STATE OF ILLINOIS)
) SS
COUNTY OF COOK)

I, the undersigned, a Notary Public in and for the county and State aforesaid, do hereby certify that Daniel E. Levin, personally known to me to be the Manager of The Habitat Company LLC, the manager of Habitat Acquisitions Company LLC, an Illinois limited liability company ("Habitat Acquisitions"), the Manager of Habitat 43 Green, LLC, an Illinois limited liability company ("Habitat 43 Green"), the managing member of 43 Green JV LLC, an Illinois limited liability company ("43 Green JV"), a Delaware limited liability company and the manager and sole member of 43rd and Calumet Phase I, LLC, an Illinois limited liability company (the "General Partner") and the general partner of 43rd and Calumet Phase I LP, a Delaware limited partnership (the "Developer"), and personally known to me to be the same person whose name is subscribed to the foregoing instrument. appeared before me this day in person and severally acknowledged that as such officer, he signed and delivered the said instrument, pursuant to authority given by the members of Habitat, Habitat Acquisitions, Parkside, on behalf of the General Partner, as the free and voluntary act of such person, and as the free and voluntary act and deed of the Rental Owner, for the uses and purposes therein set forth

	GIVEN under my hand and official seal this day of	··
	Notary Public	
	My Commission Expires	
(SEAL)		

I, the undersigned, a Notary Public in and for the county and State aforesaid, do hereby certify that, personally known to me to be the of Generations Housing Initiatives, an Illinois not-for-profit corporation ("Generations"), and personally known to me to be the same person whose name is subscribed to the foregoing instrument, appeared before me this day in person and severally acknowledged that as such officer, s/he signed and delivered the said instrument, pursuant to authority given by the Board as the free and voluntary act of such person, and as the free and voluntary act and deed of Generations, for the uses and purposes therein set forth. Given under my hand and official seal this day of, 20 Notary Public STATE OF ILLINOIS) I, the undersigned, a Notary Public in and for the county and State aforesaid, do hereby certify that, personally known to me to be the of 43 Green JV, LLC, a Delaware limited liability company ion ("LAC"), and personally known to me to be the same person whose name is subscribed to the foregoing instrument, appeared before me this day in person and severally acknowledged that as such officer, s/he signed and delivered the said instrument, pursuant to authority given by the Board as the free and voluntary act of such person, and as the free and voluntary act and deed of LAC, for the uses and purposes therein set forth. Given under my hand and official seal this day of, 20 Notary Public	STATE OF ILLINOIS)) ss COUNTY OF COOK)		
Notary Public STATE OF ILLINOIS) I, the undersigned, a Notary Public in and for the county and State aforesaid, do hereby certify that, personally known to me to be the of 43 Green JV, LLC, a Delaware limited liability company ion ("LAC"), and personally known to me to be the same person whose name is subscribed to the foregoing instrument, appeared before me this day in person and severally acknowledged that as such officer, s/he signed and delivered the said instrument, pursuant to authority given by the Board as the free and voluntary act of such person, and as the free and voluntary act and deed of LAC, for the uses and purposes therein set forth. Given under my hand and official seal this day of, 20	certify that, personally known Housing Initiatives, an Illinois not-for-profit come to be the same person whose name is sume this day in person and severally acknowled the said instrument, pursuant to authority give person, and as the free and voluntary act as	own to me to be the corporation (" Generations "), and person bscribed to the foregoing instrument, a ledged that as such officer, s/he signed wen by the Board as the free and volun	of Generations onally known to ppeared before d and delivered tary act of such
STATE OF ILLINOIS)) ss COUNTY OF COOK) I, the undersigned, a Notary Public in and for the county and State aforesaid, do hereby certify that, personally known to me to be the of 43 Green JV, LLC, a Delaware limited liability company ion ("LAC"), and personally known to me to be the same person whose name is subscribed to the foregoing instrument, appeared before me this day in person and severally acknowledged that as such officer, s/he signed and delivered the said instrument, pursuant to authority given by the Board as the free and voluntary act of such person, and as the free and voluntary act and deed of LAC, for the uses and purposes therein set forth. Given under my hand and official seal this day of, 20	Given under my hand and official sea	al this day of	, 20
I, the undersigned, a Notary Public in and for the county and State aforesaid, do hereby certify that, personally known to me to be the of 43 Green JV, LLC, a Delaware limited liability company ion ("LAC"), and personally known to me to be the same person whose name is subscribed to the foregoing instrument, appeared before me this day in person and severally acknowledged that as such officer, s/he signed and delivered the said instrument, pursuant to authority given by the Board as the free and voluntary act of such person, and as the free and voluntary act and deed of LAC, for the uses and purposes therein set forth. Given under my hand and official seal this day of, 20 Notary Public	(SEAL)	Notary Public	
certify that, personally known to me to be the of 43 Green JV, LLC, a Delaware limited liability company ion ("LAC"), and personally known to me to be the same person whose name is subscribed to the foregoing instrument, appeared before me this day in person and severally acknowledged that as such officer, s/he signed and delivered the said instrument, pursuant to authority given by the Board as the free and voluntary act of such person, and as the free and voluntary act and deed of LAC, for the uses and purposes therein set forth. Given under my hand and official seal this day of, 20 Notary Public) ss		
Notary Public	certify that, personally know a Delaware limited liability company ion ("I person whose name is subscribed to the for person and severally acknowledged that a instrument, pursuant to authority given by the	vn to me to be the of 43 of 43 AC"), and personally known to me to regoing instrument, appeared before as such officer, s/he signed and delified Board as the free and voluntary act of	Green JV, LLC, o be the same me this day in vered the said of such person,
ullet	Giveń under my hand and official sea	al this day of	, 20
·			
	(SEAL)	Notary Public	

EXHIBIT A REDEVELOPMENT AREA

EXHIBIT B

PROPERTY

[Subject to Survey and Title Insurance]

EXHIBIT C

TIF-FUNDED IMPROVEMENTS

<u>Line Item</u> Cost

TIF ELIGIBLE COSTS BUDGET

LINDO GOGE	DDG UIGH BUDGER	MID DITOYDAD
HARD COSTS	PROJECT BUDGET	TIF ELIGIBLE
		EXPENSES
Net Residential	\$21,247,308	\$5,473,307
Construction		
Pre-cast	2,977,059	766,890
Manufacturing		
Hard Cost	325,000	83,720
Adjustment		
CTA Flagman &	75,000	19,320
Winter Conditions		
General Conditions	1,829,436	471,263
Insurance	665,260	
Bonding	228,222	
Overhead	0	
Profit:	872,674	
Contingency@ 5%	1,414,720	364,432
TOTAL HARD COSTS	\$29,634,679	\$7,178,932

SOFT COSTS	PROJECT BUDGET	TIF ELIGIBLE
		EXPENSES
Environmental	\$74,435	\$74,435
Remediation		
TOTAL SOFT COSTS	\$74,435	. \$74,435

*TOTAL

[Note: If the Amount listed on Schedule C exceeds the maximum amount of City assistance, add the footnote below stating that, notwithstanding the total of TIF-Funded Improvements, the assistance to be provided by the City is limited to [describe City assistance limit in Agreement, or refer to a Section number]. It can be useful to list all potential eligible costs on Exhibit C, even if in an aggregate amount greater than the City assistance, because it can produce a clear basis of authority for DPD to pay eligible costs up to the maximum assistance amount even if some of the specific line item amounts are lower than expected.]

[[[The Budget above is approximate and may change by up to five percent (5%)]]].

^{*}Notwithstanding the total of TIF-Funded Improvements or the amount of TIF-eligible costs, the assistance to be provided by the City is limited to the amount described in **Section 4.03** and shall not exceed the lesser of [\$5,000,000 or 12.9854%] of the Project Budget.

EXHIBIT E CONSTRUCTION CONTRACT

EXHIBIT F

ESCROW AGREEMENT

EXHIBIT G

PERMITTED LIENS

1. Liens or encumbrances against the Property:

Those matters set forth as Schedule B title exceptions in the owner's title insurance policy issued by the Title Company as of the date hereof, but only so long as applicable title endorsements issued in conjunction therewith on the date hereof, if any, continue to remain in full force and effect.

2. Liens or encumbrances against Developer or the Project, other than liens against the Property, if any:

[To be completed by Developer's counsel, subject to City approval.]

EXHIBIT H-1 PROJECT BUDGET

HARD COSTS	PROJECT BUDGET	AFFORDABLE ALLOCATION	MARKET RATE ALLOCATION
Net Residential Construction	\$21,247,308	\$10,946,613	\$10,300,695
Pre-cast Manufacturing	2,977,059	1,533,781	1,443,278
Hard Cost Adjustment	325,000	467,440	157,560
CTA Flagman & Winter Conditions	75,000	38,640	36,360
General Conditions	1,829,436	942,525	886,911
Insurance	665,260	342,742	322,518
Bonding	228,222	117,580	110,642
Overhead	0	0	0
Profit	872,674	449,602	423,072
Contingency@ 5%	1,414,720	728,864	685,856
TOTAL HARD COSTS	\$29,634,679	\$15,267,787	\$14,366,892

SOFT COSTS	PROJECT BUDGET	AFFORDABLE ALLOCATION	MARKET RATE ALLOCATION
FFE	305,000	157,135	147,864
Permit Fees	140,000	72,128	67,872
Permit Review Fees	79,649	41,034	38,612
Direct Developer Review	19,432	10,011	9,421
Testing During Construction	100,000	51,520	48,480
Streets	20,000	10,304	9,696
Parkway Tree Fee	15,000	7,728	`7,272
Temporary Power	15,000	7,728	7,272
People's Gas	10,000	5,152	4,848
Utility Consultant	16,225	8,359	7,866
Utility Costs-ComEd	60,000	30,912	29,088
Construction Soft Cost Contingency	40,232	20,728	19,504
Environmental Remediation	74,435	38,349	36,086
Environmental Monitoring	131,000	67,491	63,509
LEED Commissioning Agent	45,800	23,596	22,204
IEPA Fees	20,000	10,304	9,696
Architect -Design & Supervision	1,286,020	667,710	628,310
Architect - Misc & Reimbursables	12,000	6,182	5,818
Tie off Anchor Design	20,000	10,304	9,696
Geotechnical Analysis	26,000	13,395	12,605
Permit Expeditor	15,400	7,934	7,466
Plats & Surveys	10,000	5,152	4,848
PD Survey	5,750	2,962	2,788
Cost Certification/Audit	10,000	5,152	4,848
Legal Fees-Organizational	175,000	90,160	84,840
Legal Fees-Zoning	120,000	61,824	58,176
Legal Fees-Syndicator	50,000	25,760	24,240
TIF Consultant	30,000	15,456	14,544
Appraisal	10,000	5,152	4,848
Market Study	10,000	5,152	4,848
Phase I Environmental Report	2,250	1,159	1,091
Phase II Environmental Report	15,460	7,965	7,495
Title & Recording	58,225	29,998	28,227
Tax Credit Reservation Fee	109,733	56,535	53,199
Application Fees	42,000	21,638	20,362
Construction Points	175,000	90,160	84,840
Perm Loan Points	135,000	69,552	65,448
Construction Inspection	45,000	23,184	21,816
Legal Fees- Lender	100,000	51,520	49,480
Construction Period Interest	1,022,975	527,037	495,938
Other Lender Fees	122,500	63,112	59,388
Construction Period Insurance	63,080	32,499	30,581
Construction Period Taxes	25,000	12,880	12,120
Commercial Space Soft Costs	284,460	146,554	137,906
Other Construction Period	198,061	102,041	96,020
Leasing Personnel	35,000	18,032	16,968
Advertising	10,000	5,152	4,848
Model Units	10,000	5,152	4,848
Other Marketing & Leasing	15,000	7,728	7,272
Lease Up Reserve	64,350	33,153	31,197
Insurance Reserve	47,250	24,343	22,907
Real Estate Tax Reserve	50,000	25,760	24,240
Operating Reserve	1,200,000	618,240	581,760
Replacement Reserve	44,550	22,952	21,598
Developer Fee	1,457,350	750,827	706,523
Deferred Developer i ee	<u>656,000</u>	337,971	318,029
TOTAL SOFT COSTS	8,870,184	4,569,919	4,300,265
TOTAL PROJECT COSTS	38,870,184	19,837,706	18,667,158

[[[The Budget above is approximate and may change by up to five percent (5%)]]].

EXHIBIT H-2

MBE/WBE BUDGET

MBE/WBE ELIGIBLE COSTS

HARD COSTS	PROJECT BUDGET	MBE/WBE ELIGIBLE EXPENSES
Net Residential Construction	\$21,247,308	\$21,247,308
Pre-cast Manufacturing	2,977,059	0
Hard Cost Adjustment	325,000	325,000
CTA Flagman & Winter	75,000	75,000
Conditions		
General Conditions	1,829,436	1,829,436
Insurance	665,260	665,260
Bonding	228,222	228,222
Overhead	0	0
Profit	872,674	872,674
Contingency@ 5%	1,414,720	<u>0</u>
TOTAL HARD COSTS	\$29,634,679	\$25,242,900

EXHIBIT I APPROVED PRIOR EXPENDITURES

EXHIBIT J

OPINION OF DEVELOPER'S COUNSEL

[To be retyped on Developer's Counsel's letterhead]

City of Chicago 121 North LaSalle Street Chicago, IL 60602
ATTENTION: Corporation Counsel
Ladies and Gentlemen:
We have acted as counsel to, an [Illinois] (the ADeveloper"), in connection with the purchase of certain land and the construction of certain facilities thereon located in the Redevelopment Project Area (the "Project"). In that capacity, we have examined, among other things, the following agreements, instruments and documents of even date herewith, hereinafter referred to as the ADocuments":
(a) Redevelopment Agreement (the "Agreement") of ever date herewith, executed by Developer and the City of Chicago (the "City");
[(b) the Escrow Agreement of even date herewith executed by Developer and the City;]
(c) [insert other documents including but not limited to documents related to purchase and financing of the Property and all lender financing related to the Project]; and
(d) all other agreements, instruments and documents executed in connection with the foregoing.
In addition to the foregoing, we have examined
(a) the original or certified, conformed or photostatic copies of Developer's (i) Articles of Incorporation, as amended to date, (ii) qualifications to do business and certificates of good standing in all states in which Developer is qualified to do business, (iii) By-Laws, as amended to date, and (iv) records of all corporate proceedings relating to the Project [revise if Developer is not a corporation]; and
(b) such other documents, records and legal matters as we have deemed necessary o relevant for purposes of issuing the opinions hereinafter expressed.

than those of Developer), the authenticity of documents submitted to us as originals and conformity to the originals of all documents submitted to us as certified, conformed or photostatic copies.

In all such examinations, we have assumed the genuineness of all signatures (other

Based on the foregoing, it is our opinion that:

- 1. Developer is a corporation duly organized, validly existing and in good standing under the laws of its state of [incorporation] [organization], has full power and authority to own and lease its properties and to carry on its business as presently conducted, and is in good standing and duly qualified to do business as a foreign [corporation] [entity] under the laws of every state in which the conduct of its affairs or the ownership of its assets requires such qualification, except for those states in which its failure to qualify to do business would not have a material adverse effect on it or its business.
- Developer has full right, power and authority to execute and deliver the Documents to which it is a party and to perform its obligations thereunder. Such execution, delivery and performance will not conflict with, or result in a breach of, Developer's [Articles of Incorporation or By-Laws] [describe any formation documents if Developer is not a corporation] or result in a breach or other violation of any of the terms, conditions or provisions of any law or regulation, order, writ, injunction or decree of any court, government or regulatory authority, or, to the best of our knowledge after diligent inquiry, any of the terms, conditions or provisions of any agreement, instrument or document to which Developer is a party or by which Developer or its properties is bound. To the best of our knowledge after diligent inquiry, such execution, delivery and performance will not constitute grounds for acceleration of the maturity of any agreement, indenture, undertaking or other instrument to which Developer is a party or by which it or any of its property may be bound, or result in the creation or imposition of (or the obligation to create or impose) any lien, charge or encumbrance on, or security interest in, any of its property pursuant to the provisions of any of the foregoing, other than liens or security interests in favor of the lender providing Lender Financing (as defined in the Agreement).
- 3. The execution and delivery of each Document and the performance of the transactions contemplated thereby have been duly authorized and approved by all requisite action on the part of Developer.
- 4. Each of the Documents to which Developer is a party has been duly executed and delivered by a duly authorized officer of Developer, and each such Document constitutes the legal, valid and binding obligation of Developer, enforceable in accordance with its terms, except as limited by applicable bankruptcy, reorganization, insolvency or similar laws affecting the enforcement of creditors' rights generally.
- 5. Exhibit A attached hereto (a) identifies each class of capital stock of Developer, (b) sets forth the number of issued and authorized shares of each such class, and (c) identifies the record owners of shares of each class of capital stock of Developer and the number of shares held of record by each such holder. To the best of our knowledge after diligent inquiry, except as set forth on Exhibit A, there are no warrants, options, rights or commitments of purchase, conversion, call or exchange or other rights or restrictions with respect to any of the capital stock of Developer. Each outstanding share of the capital stock of Developer is duly authorized, validly issued, fully paid and nonassessable.
- 6. To the best of our knowledge after diligent inquiry, no judgments are outstanding against Developer, nor is there now pending or threatened, any litigation, contested claim or governmental proceeding by or against Developer or affecting Developer or its property, or seeking to restrain or enjoin the performance by Developer of the Agreement or the transactions contemplated by the Agreement, or contesting the validity thereof. To the best of our knowledge

after diligent inquiry, Developer is not in default with respect to any order, writ, injunction or decree of any court, government or regulatory authority or in default in any respect under any law, order, regulation or demand of any governmental agency or instrumentality, a default under which would have a material adverse effect on Developer or its business.

- 7. To the best of our knowledge after diligent inquiry, there is no default by Developer or any other party under any material contract, lease, agreement, instrument or commitment to which Developer is a party or by which the company or its properties is bound.
- 8. To the best of our knowledge after diligent inquiry, all of the assets of Developer are free and clear of mortgages, liens, pledges, security interests and encumbrances except for those specifically set forth in the Documents.
- 9. The execution, delivery and performance of the Documents by Developer have not and will not require the consent of any person or the giving of notice to, any exemption by, any registration, declaration or filing with or any taking of any other actions in respect of, any person, including without limitation any court, government or regulatory authority.
- 10. To the best of our knowledge after diligent inquiry, Developer owns or possesses or is licensed or otherwise has the right to use all licenses, permits and other governmental approvals and authorizations, operating authorities, certificates of public convenience, goods carriers permits, authorizations and other rights that are necessary for the operation of its business.
- 11. A federal or state court sitting in the State of Illinois and applying the choice of law provisions of the State of Illinois would enforce the choice of law contained in the Documents and apply the law of the State of Illinois to the transactions evidenced thereby.

We are attorneys admitted to practice in the State of Illinois and we express no opinion as to any laws other than federal laws of the United States of America and the laws of the State of Illinois.

[Note: include a reference to the laws of the state of incorporation/organization of Developer, if other than Illinois.]

This opinion is issued at Developer's request for the benefit of the City and its counsel, and may not be disclosed to or relied upon by any other person.

Vary truly yours

very truly	yours,	
D		
ву		
By [.] Name [.]		

EXHIBIT K

PRELIMINARY TIF PROJECTION -- REAL ESTATE TAXES

EXHIBIT L

REQUISITION FORM

STATE OF ILLINOIS)		
) SS COUNTY OF COOK)		
The affiant,, of, a(the ADeveloper"), hereby certifies that with respect to that		
(the ADeveloper"), hereby certifies that with respect to that certain Redevelopment Agreement between Developer and the City of Chicago dated, (the "Agreement"):		
A. Expenditures for the Project, in the total amount of \$, have been made:		
B. This paragraph B sets forth and is a true and complete statement of all costs of TIF-Funded Improvements for the Project reimbursed by the City to date:		
\$		
C. Developer requests reimbursement for the following cost of TIF-Funded Improvements:		
\$		
D. None of the costs referenced in paragraph C above have been previously reimbursed by the City.		
E. Developer hereby certifies to the City that, as of the date hereof:		
1. Except as described in the attached certificate, the representations and warranties contained in the Agreement are true and correct and Developer is in compliance with all applicable covenants contained herein		
2. No event of Default or condition or event which, with the giving of notice or passage of time or both, would constitute an Event of Default, exists or has occurred.		
All capitalized terms which are not defined herein has the meanings given such terms in the Agreement.		

[Developer]	
By:	
Name	•
Title:	
Subscribed and sworn before me this	day of
 ·	
My commission expires:	
Agreed and accepted:	
Nama	
Name	
Title:	
City of Chicago	
Department of Planning and Development	opment

EXHIBIT M

FORM OF NOTE

INVESTMENT IN THIS NOTE INVOLVES A HIGH DEGREE OF RISK. IT IS SUITABLE ONLY FOR PERSONS WHO ARE ABLE TO BEAR THE ECONOMIC RISKS OF THIS INVESTMENT, INCLUDING TOTAL LOSS. NO ASSURANCE CAN BE PROVIDED THAT THE HOLDER OF THIS NOTE WILL NOT LOSE ITS ENTIRE INVESTMENT IN THIS NOTE. SEE "NOTEHOLDER RISKS" ATTACHED TO THIS NOTE.

THIS NOTE IS SUITABLE ONLY FOR PERSONS WHO HAVE NO NEED FOR LIQUIDITY. THIS NOTE MAY ONLY BE TRANSFERRED IN THE MANNER AND SUBJECT TO THE LIMITATIONS PROVIDED IN THE REDEVELOPMENT AGREEMENT.

THE CITY DOES NOT ENDORSE PROJECTIONS OF ANY KIND FROM ANY SOURCE AS TO THE SUFFICIENCY OF ALLOCATED [AVAILABLE] INCREMENTAL TAXES TO PAY PRINCIPAL OF AND INTEREST ON THIS NOTE. INVESTORS WHO RELY ON SUCH PROJECTIONS DO SO AT THEIR OWN RISK.

AVAILABLE INCRE ACCOUNT, AS DEFI THE HOLDER OF T	INTEREST ON THIS NOTE ARE PAYABLE SOLEL MENTAL TAXES ON DEPOSIT IN THE [
REGISTERED NO. R-1		MAXIMUM AMOUNT \$1
	UNITED STATES OF AMERICA STATE OF ILLINOIS COUNTY OF COOK CITY OF CHICAGO CREMENT ALLOCATION REVENUE NOTE (EDEVELOPMENT PROJECT), [TAXABLE] SERIES	[A]
Registered Owner:	[Developer]	
Interest Rate:	per annum	
Maturity Date:	[no later than the earlier of (a) two issuance date, or (b) the date on which the Project in effect]	

¹ Or such lower amount determined in accordance with Section 4.03(b)

Principal of and interest on this Note, payable solely from the [Available Incremental Taxes] [USE APPLICABLE TERM] (as defined in the hereinafter defined Redevelopment Agreement), is due March 1 of each year until the earlier of Maturity or until this Note is paid in full. Payments shall first be applied to interest. The principal of and interest on this Note are payable in lawful money of the United States of America, and shall be made to the Registered Owner hereof as shown on the registration books of the City maintained by the Comptroller of the City, as registrar and paying agent (the "Registrar"), at the close of business on the fifteenth day of the month immediately prior to the applicable payment, maturity or redemption date, and shall be paid by check or draft of the Registrar, payable in lawful money of the United States of America, mailed to the address of such Registered Owner as it appears on such registration books or at such other address furnished in writing by such Registered Owner to the Registrar; provided, that the final installment of principal and accrued but unpaid interest will be payable solely upon presentation of this Note at the principal office of the Registrar in Chicago, Illinois or as otherwise directed by the City. The Registered Owner of this Note shall note on the Payment Record attached hereto the amount and the date of any payment of the principal of this Note promptly upon receipt of such payment.

This Note is issued by the City in the principa	amount of [advances made from time to time
by the Registered Owner] up to \$	for the purpose of paying the costs of certain
eligible redevelopment project costs incurred by	[INSERT NAME OF
DEVELOPER] (the "Project"), which were [acquired],	[constructed] and [installed] in connection with
the development of an approximately [acr	e/ square foot] site/building in the
Redevelopment Project Area (tl	ne " <u>Project Area</u> ") in the City, all in accordance
with the Constitution and the laws of the State of	Illinois, and particularly the Tax Increment

² Or such lower amount determined in accordance with Section 4.03(b)

Allocation Redevelopment Act (65 ILCS 5/11-74.4-1 et seq.) (the "TIF Act"), the Local Government Debt Reform Act (30 ILCS 350/1 et seq.) and an Ordinance adopted by the City Council of the City on ______, ____ (the AOrdinance"), in all respects as by law required.

The City has assigned and pledged certain rights, title and interest of the City in and to certain incremental ad valorem tax revenues from the Project Area which the City is entitled to receive pursuant to the TIF Act and the Ordinance, in order to pay the principal of and interest of this Note. Reference is hereby made to the aforesaid Ordinance and the Redevelopment Agreement for a description, among others, with respect to the determination, custody and application of said revenues, the nature and extent of such security with respect to this Note and the terms and conditions under which this Note is issued and secured. THIS NOTE IS A SPECIAL LIMITED OBLIGATION OF THE CITY, AND PRINCIPAL OF AND INTEREST ON THIS NOTE ARE PAYABLE SOLELY FROM ALLOCATED [AVAILABLE] INCREMENTAL TAXES ON DEPOSIT IN THE _____ ACCOUNT OF THE TIF FUND (AS DEFINED IN THE REDEVELOPMENT AGREEMENT) AFTER PAYMENT OF ALL OBLIGATIONS HAVING A PRIORITY OVER THIS NOTE, IF ANY. THE HOLDER OF THIS NOTE ACCEPTS THE RISK THAT THE AMOUNT OF ALLOCATED [AVAILABLE] INCREMENTAL TAXES MAY NOT BE SUFFICIENT TO PAY THE PRINCIPAL OF OR INTEREST ON THIS NOTE. THIS NOTE SHALL NOT BE DEEMED TO CONSTITUTE AN INDEBTEDNESS OR A LOAN AGAINST THE GENERAL TAXING POWERS OR CREDIT OF THE CITY WITHIN THE MEANING OF ANY CONSTITUTIONAL OR STATUTORY PROVISION. THE REGISTERED OWNER OF THIS NOTE SHALL NOT HAVE THE RIGHT TO COMPEL ANY EXERCISE OF THE TAXING POWER OF THE CITY, THE STATE OF ILLINOIS OR ANY POLITICAL SUBDIVISION THEREOF TO PAY THE PRINCIPAL OF OR INTEREST ON THIS NOTE. The principal of this Note is subject to redemption on any date, as a whole or in part, at a redemption price of 100% of the principal amount thereof being redeemed. There shall be no prepayment penalty. Notice of any such redemption shall be sent by registered or certified mail not less than five (5) days nor more than sixty (60) days prior to the date fixed for redemption to the registered owner of this Note at the address shown on the registration books of the City maintained by the Registrar or at such other address as is furnished in writing by such Registered Owner to the Registrar.

This Note is issued in fully registered form in the denomination of its outstanding principal amount. This Note may not be exchanged for a like aggregate principal amount of notes or other denominations.

THIS NOTE MAY NOT BE OFFERED. SOLD. PLEDGED OR OTHERWISE TRANSFERRED EXCEPT TO (I) AN INSTITUTIONAL "ACCREDITED INVESTOR" WITHIN THE MEANING OF RULE 501(a)(1), (2), (3) or (7) UNDER THE SECURITIES ACT OF 1933 (THE "SECURITIES ACT") THAT DELIVERS TO THE CITY AN INVESTOR LETTER IN THE FORM OF EXHIBIT P TO THE REDEVELOPMENT AGREEMENT REFERENCED BELOW, OR (II) A PERSON (OTHER THAN A DEALER) WHOM THE SELLER REASONABLY BELIEVES IS A "QUALIFIED INSTITUTIONAL BUYER" WITHIN THE MEANING OF RULE 144A(a)(1) UNDER THE SECURITIES ACT. ANY HOLDER OF THIS NOTE IS REQUIRED TO NOTIFY ANY POTENTIAL PURCHASER OF THIS NOTE OF THE RESALE RESTRICTIONS REFERRED TO ABOVE. This Note is transferable by the Registered Owner hereof in person or by its attorney duly authorized in writing at the principal office of the Registrar in Chicago, Illinois, but only in the manner and subject to the limitations provided in the Ordinance, and upon surrender and cancellation of this Note. Upon such transfer, a new Note of authorized denomination of the same maturity and for the same aggregate principal amount will be issued to the transferee in exchange herefor. The Registrar shall not be required to transfer this Note during the period beginning at the close of business on the fifteenth day of the month immediately prior to the maturity date of this Note nor to transfer this Note after notice calling this Note or a portion hereof for redemption has been mailed, nor during a period of five (5) days next preceding mailing of a notice of redemption of this Note. Such transfer shall be in accordance with the form at the end of this Note.

This Note hereby authorized shall be executed and delivered as the Ordinance and the Redevelopment Agreement provide.

Pursuant to the Redevelopment Agreement dated as of ______, _____ between the City and the Registered Owner (the "Redevelopment Agreement"), the Registered Owner has agreed to [acquire] and [construct] the Project and to advance funds for the [construction of certain facilities] related to the Project on behalf of the City. The cost of such acquisition and construction in the amount of \$[______] shall be deemed to be a disbursement of the proceeds of this Note.

Pursuant to Section 15.02 of the Redevelopment Agreement, the City has reserved the right to [suspend] [terminate] payments of principal of and interest on this Note upon the occurrence of certain conditions, [and the City has reserved the right to offset liquidated damage amounts owed to the City against the principal amount outstanding under this Note]. The City shall not be obligated to make payments under this Note if an Event of Default (as defined in the Redevelopment Agreement), or condition or event that with notice or the passage of time or both would constitute an Event of Default, has occurred. Such rights shall survive any transfer of this Note.

In the event of a conflict between any provisions of this Note and the provisions of the Redevelopment Agreement, the Redevelopment Agreement shall prevail and control.

The City and the Registrar may deem and treat the Registered Owner hereof as the absolute owner hereof for the purpose of receiving payment of or on account of principal hereof and for all other purposes and neither the City nor the Registrar shall be affected by any notice to the contrary, unless transferred in accordance with the provisions hereof.

This Note may be transferred only in the manner and subject to the limitations provided in Section [18.21] of the Redevelopment Agreement.

It is hereby certified and recited that all conditions, acts and things required by law to exist, to happen, or to be done or performed precedent to and in the issuance of this Note did exist, have happened, have been done and have been performed in regular and due form and time as required by law; that the issuance of this Note, together with all other obligations of the City, does not exceed or violate any constitutional or statutory limitation applicable to the City.

This Note shall not be valid or become obligatory for any purpose until the certificate of authentication hereon shall have been signed by the Registrar.

[THE REMAINDER OF THIS PAGE INTENTIONALLY LEFT BLANK]

IN WITNESS WHEREOF, the City of Chicago, Cook County, Illinois, by its City Council, has caused its official seal to be imprinted by facsimile hereon or hereunto affixed, and has caused this		
Note to be signed by the duly authorized signature of the Mayor and attested by the duly authorized		
signature of the City Clerk of the City, all as of,		
Mayor		
(SEAL)		

City Clerk

Attest:

CERTIFICATE OF AUTHENTICATION

Comptroller Date:

Registrar and Paying Agent Comptroller of the City of Chicago, Cook County, Illinois

[Note: Use Applicable Definition: (Available Incremental Taxes v. Incremental Taxes]

[The following "Noteholder Risks" constitutes an integral part of this Note.]

NOTEHOLDER RISKS

The purchase of or investment in the Note involves certain risks. Each prospective holder or purchaser of the Note, or any interest therein, should make an independent evaluation of the financial and business risks associated with holding or having an investment interest in the Note. Certain of these risks are set forth below. The following summary is not intended to be complete and does not purport to identify all possible risks that should be considered by prospective holders of the Note or any interests therein. Capitalized terms used herein have the meanings set forth in the Note

All prospective holders of the Note are urged to consult with their financial adviser and legal counsel before acquiring the Note or any interest therein.

Loss of Investment

Investment in the Note involves a high degree of risk. It is suitable only for persons who are able to bear the economic risks of the investment, including total loss. No assurance can be provided that prospective holders of the Note will not lose their entire investment in the Note.

Lack of Liquidity

The Note is suitable only for persons who have no need for liquidity. The transferability of the Note is restricted. The Note may only be transferred in the manner and subject to the limitations provided in the Redevelopment Agreement. Investors in the Note must be prepared to hold the Note until the maturity of the Note.

Reliance on Projections

The City does not endorse projections of any kind from any source as to the sufficiency of Available Incremental Taxes to pay principal of and interest on the Note. Investors who rely on any such projections do so at their own risk.

The City's Office of Budget and Management ("OBM") produces five-year projection reports for each TIF district in the City for the purpose of evaluating resources and project balances ("District Projection Reports"). This information, which is currently publicly available, is used by the OBM to determine how much funding has been committed and how much funding is available for potential projects. The District Projection Reports and the projections included therein are not audited and do not represent a final accounting of funds. The District Projection Reports are not prepared for investors or as a basis for making investment decisions with respect to any notes, bonds or other debt obligations of the City that are payable from Incremental Taxes, including the Note. Prospective investors in the Note are cautioned not to rely on any of the information contained in the District Projection Reports.

Limited Obligations

THE NOTE IS A SPECIAL LIMITED OBLIGATION OF THE CITY PAYABLE SOLELY FROM THE ALLOCATED [AVAILABLE] INCREMENTAL TAXES AFTER PAYMENT OF ALL OBLIGATIONS HAVING A PRIORITY OVER THE NOTE, IF ANY, AND SHALL BE A VALID CLAIM ONLY AGAINST SAID SOURCES. THE NOTE DOES NOT CONSTITUTE AN INDEBTEDNESS OR A LOAN AGAINST THE GENERAL TAXING POWERS OR CREDIT OF THE CITY, WITHIN THE MEANING OF ANY CONSTITUTIONAL OR STATUTORY PROVISION. THE NOTEHOLDER HAS NO RIGHT TO COMPEL ANY EXERCISE OF THE TAXING POWER OF THE CITY, THE STATE OF ILLINOIS OR ANY POLITICAL SUBDIVISION THEREOF TO PAY THE PRINCIPAL OF OR INTEREST ON THE NOTE.

There can be no assurance that Available Incremental Taxes will be sufficient for payment of amounts due and owing on the Note.

Limited Information

The Note was issued to the Developer under the Redevelopment Agreement as part of a commercial transaction negotiated by the Developer and the City. [The Developer] engaged a [consultant] to deliver a [feasibility report][projection report] to the City in connection with the Project, which included certain information about the Project Area, the Project and historical and projected [Available Incremental Taxes.] The report contained information as of its date only, and neither the Developer nor any other party have any obligation to update the report as of any subsequent date.

The City is under no continuing obligation to provide to any holder or prospective holder of the Note, or to post to the Electronic Municipal Market Access System of the Municipal Securities Rulemaking Board or any other website, any current or updated information with respect to the Project Area, the Project, the historical and projected Available Incremental Taxes or the Note. The City does not prepare or have readily available any current or updated information about the Project Area, the Project or the Available Incremental Taxes.

Unavailability of City Funds

The City is not obligated to pay principal of or interest on the Note in any year in which there are inadequate Available Incremental Taxes. The City is obligated to pay the amount of any unpaid principal or accrued interest in any subsequent year but only to the extent of the availability of Available Incremental Taxes for those subsequent years. If, on the maturity date of the Note, any outstanding unpaid principal or interest on the Note exists for any reason, including without limitation the inadequacy of Available Incremental Taxes, such outstanding principal and/or interest will be forgiven in full and the City will have no further obligation to pay such outstanding amount. In such event, there would be no further payments of principal or interest in respect of the Note.

Risk of Failure to Maintain Levels of Assessed Valuation

[USE THE FOLLOWING PARAGRAPH IF THE NOTE PAYMENTS WILL BE PAID FROM PROJECT PINS ONLY]

There can be no assurance that the equalized assessed value of the Project property will remain the same throughout the term of the Note. Furthermore, the successful petition or application of any owner for the reduction of the assessed value of the Project property may cause the equalized assessed value of the [Property/Land/Project – USE THE RDA DEFINED TERM FOR

THE REAL PROPERTY] to be less than the originally projected equalized assessed value. If any time during the term of the Note the actual equalized assessed value is less than what was projected, the generation of Available Incremental Taxes for payment on the Note is likely to be significantly impaired.

[USE THE FOLLOWING PARAGRAPH IF THE NOTE PAYMENTS WILL BE PAID FROM PROJECT AREA-WIDE INCREMENT]

There can be no assurance that the equalized assessed value of the Project property will remain the same throughout the term of the Note. Furthermore, the successful petition or application of any owner for the reduction of the assessed value of the Project property or any other properties in the Project Area may cause the equalized assessed value of properties in the Project Area to be less than the originally projected equalized assessed value. If any time during the term of the Note the actual equalized assessed value is less than what was projected, the generation of Available Incremental Taxes for payment on the Note is likely to be significantly impaired.

Risk of Change in Incremental Taxes

Prospective holders of the Note should carefully consider, among other factors, the risks associated with the ultimate generation of Available Incremental Taxes in the Project Area. These risks include, but are not limited to, the following:

- 1. Property tax rates are calculated by the Cook County Clerk for numerous funds of a number of taxing districts that tax all or part of the property in the Project Area. A reduction in the tax levies by the affected taxing districts may have an adverse effect on the Available Incremental Taxes.
- 2. Further changes may be made in the real property tax system by the State of Illinois or Cook County. Such changes could include various property tax rollbacks, abatements, exemptions, changes in the ratio of assessment, or relief measures, limitations on the amount or percent of increase in tax levies by taxing districts, or other measures that would limit the tax levy amount that could be extended to the property within the Project Area and, consequently, the projected Available Incremental Taxes generated. For example, if Illinois adopted practices used in other states, the property tax system could be changed so that schools would be financed from a source other than property taxes. This type of change could have a significant adverse effect upon Available Incremental Taxes.
- 3. Cook County's methodology and procedures used to assess the value of property may be altered resulting in a potentially reduced or altered valuation in a particular year or succession of years.

FUTURE LEGISLATION, REGULATIONS, GOVERNMENTAL OR JUDICIAL INTERPRETATION OF REGULATIONS OR LEGISLATION OR PRACTICES AND PROCEDURES RELATED TO PROPERTY TAX ASSESSMENT, LEVY, COLLECTIONS OR DISTRIBUTION COULD HAVE A MATERIAL EFFECT ON THE CALCULATION OR AVAILABILITY OF INCREMENTAL TAXES COLLECTED OR DISTRIBUTED AND THEREFORE A MATERIAL EFFECT ON THE AMOUNT OF AVAILABLE INCREMENTAL TAXES FOR PAYMENT OF PRINCIPAL OF AND INTEREST ON THE NOTE.

Changes in Multiplier and Tax Rate

The equalization factor annually determined by the Illinois Department of Revenue for properties located within Cook County (commonly referred to as the "multiplier") may vary substantially in future years. A decrease in the multiplier would reduce the equalized assessed value of the taxable real property in the Project Area and, therefore, the Available Incremental Taxes available to pay principal of and interest on the Note. The future tax rates of the units of local government levying taxes in the Project Area either individually or on a composite basis, may differ from their historical levels. Any decrease in the composite tax rate of the governmental units would decrease the amount of Available Incremental Taxes available to pay principal of and interest on the Note. Any decrease in the composite tax rate of the governmental units could occur in future years as a result of various factors, including, but not limited to, one or more of the following: (a) reduced governmental costs; (b) constitutional or statutory spending or tax rate limitations; or (c) governmental reorganization or consolidation.

Economic Risks Affecting Incremental Taxes

Changing economic circumstances or events in the Project Area may result in reductions in Available Incremental Taxes available to pay principal of and interest on the Note. Relocations of major property owners to sites outside the Project Area or sales of major properties to tax-exempt entities could reduce the assessed valuation of the real properties in the Project Area. Substantial damage to or destruction of improvements within the Project Area could cause a material decline in assessed valuation and impair the ability of the taxpayers in the Project Area to pay their respective portions of real estate taxes. Similarly, there can be no assurance that the improvements in the Project Area will be sufficiently insured under fire and extended coverage insurance policies. Even if such insurance is sufficient, the proceeds thereof will not be assigned as security for the payment of real estate taxes or to secure payment of the Note. In addition, any insurance proceeds may not be sufficient to repair or rebuild said improvements. The restoration of such improvements may be delayed by other factors, or the terms of then-applicable mortgage financing could require the application of insurance proceeds to the reduction of mortgage balances. Any of the foregoing circumstances could result in the assessed valuation of property in the Project Area remaining depressed for an unknown period of time and decrease the amount of Available Incremental Taxes available to pay principal of and interest on the Note.

Results of operation of properties within the Project Area depend, in part, on sales, leases, rental rates and occupancy levels, which may be adversely affected by competition, suitability of the properties located in the Project Area, local unemployment, availability of transportation, neighborhood changes, crime levels in the Project Area, vandalism, rising operating costs and similar factors. Poor operating results of properties within the Project Area may cause delinquencies in the payment of real estate taxes, reduce assessed valuations and increase the risk of foreclosures. Successful petitions by taxpayers to reduce their assessed valuations could adversely affect Available Incremental Taxes available for payment of principal of and interest on the Note.

Failure to Sell or Lease Property

At the time of Note issuance, the redevelopment plan called for the Developer [to sell/lease to commercial or industrial enterprises/retailers prior to/in connection with] completion of the Project. The slowdown, stoppage or failure of the Developer to complete the Project and to successfully sell/lease the Project could delay or reduce the amount of Available Incremental Taxes generated in

the Project Area. Such delay or reduction could lead to a default in payments of the principal of and interest on the Note.

Reliance on Primary Taxpayers

If one or only a few property owners within the Project Area are responsible for generating a substantial amount of the Incremental Taxes, the generation of Available Incremental Taxes could be significantly adversely affected if such owner or owners and/or their tenants discontinue or curtail their businesses or terminate or default on their leases, and substitutes or replacements cannot be made on a timely basis.

Force Majeure Conditions

Riots, civil disturbances, vandalism, fires, and natural disasters or other "Acts of God" affecting the conditions and viability of the Project Area may reduce or eliminate the receipt of Incremental Taxes which would result in the reduction or elimination of Available Incremental Taxes to pay principal of and interest on the Note.

Contiguous Project Areas

The Project Area is, or may become, contiguous with other redevelopment areas designated by the City pursuant to the TIF Act. The TIF Act allows the City to expend Incremental Taxes collected from the Project Area which are in excess of the amounts required in each year to pay and secure obligations issued and project costs incurred with respect to the Project Area to pay for costs eligible for payment under the TIF Act which are incurred in such contiguous areas. In the event Incremental Taxes from the Project Area in excess of Available Incremental Taxes, along with the amounts required to (i) pay principal and interest coming due on the Note in any year, and (ii) be deposited in other funds and accounts maintained under the Redevelopment Agreement, are allocated to a contiguous project redevelopment area, such excess Incremental Taxes will not be available to remedy any future failure to pay principal of and interest on the Note.

Risk of Delay in Payment

The failure of current or future owners of real property in the Project Area to remit property taxes to Cook County when due or the failure of Cook County to timely remit Incremental Taxes to the City could result in insufficient Available Incremental Taxes being available to pay principal of or interest on the Note when due.

Delays in Exercising Remedies

The enforceability of the Note is subject to applicable bankruptcy laws, equitable principles affecting the enforcement of creditors' rights generally and of liens securing such rights, and the police powers of the State of Illinois and its political subdivisions. Because of delays inherent in obtaining judicial remedies, it should not be assumed that these remedies could be accomplished rapidly.

Remedies available to the Noteholder may be limited by a variety of factors and may be inadequate to assure the timely payment of principal of and interest on the Note, or to preserve the tax-exempt status of the Note. The Note is not subject to acceleration due to payment default. Lack of remedies may entail risks of delay, limitation, or modification of the rights of the Noteholder.

Judicial remedies, such as foreclosure and enforcement of covenants, are subject to exercise of judicial discretion.

Risk of Transferee Becoming a Debtor in Bankruptcy

If a transferee of the Note were to become a debtor under the United States Bankruptcy Code or applicable state laws, a creditor or trustee in bankruptcy of the transferee might argue that the sale of the Note by the transferee constituted a fraudulent conveyance or a pledge of the Note rather than a sale. If such positions were accepted by a court, then delays in principal and interest payments to the Noteholder could occur or reductions in the amounts of such payments could result. Additionally, if the transfer of the Note is re-characterized as a pledge, then a tax lien, governmental lien or other lien created by operation of law on the property of the transferee could have priority over the holder's interest in the Note.

Loss of Tax Exemption

Interest on the Note could become includible in gross income for federal income tax purposes retroactive to the date of issuance of the Note as a result of a failure of the City to comply with certain provisions of the Internal Revenue Code of 1986, as amended (the "Code"). An event of taxability does not trigger a mandatory redemption of the Note, and the Note will remain outstanding to maturity or until redeemed.

THE ABOVE IS NOT INTENDED TO BE A COMPREHENSIVE DISCUSSION OF ALL POTENTIAL RISKS ASSOCIATED WITH THE NOTE.

* * * * *

PRINCIPAL PAYMENT RECORD

DATE OF PAYMENT PRINCIPAL PAYMENT PRINCIPAL BALANCE DUE

(ASSIGNMENT)

FOR '	VALUE RECEIVED, the undersigned sells, assigns and transfers unto [NAME	OF		
ASSIGNEE]	the within Note and does hereby irrevocably constitute and app	ooint		
	attorney to transfer the said Note on the books kept for registra	ation		
thereof with fo	full power of substitution in the premises.			
Dated:				
	Registered Owner			
NOTICE:	The signature to this assignment must correspond with the name of the Registor Owner as it appears upon the face of the Note in every particular, without alterator enlargement or any change whatever.			
Signature	Guaranteed:			
	Notice: Signature(s) must be guaranteed by a member of the New York S Exchange or a commercial bank or trust company.	tock		
Consented to	b by:			
CITY OF CHICAGO DEPARTMENT OF PLANNING AND DEVELOPMENT				
BY:				
ITS:				
CITY OF CHI DEPARTMEN	ICAGO NT OF FINANCE			
BY:				
ITS:				

CERTIFICATION OF EXPENDITURE

(Closir	osing Date)				
To:	Registered Owner				
Re:	City of Chicago, Cook County, Illinois (the "City") \$ Tax Increment Allocation Revenue Note (Redevelopment Project, [Taxable] Series [A]) (the "Redevelopment Note")				
	This Certification is submitted to you, Registered Owner of the Redevelopment Note, pursuant to the Ordinance of the City authorizing the execution of the Redevelopment Note adopted by the City Council of the City on, (the AOrdinance"). All terms used herein shall have the same meaning as when used in the Ordinance.				
charge Ordina outstar	The City hereby certifies that \$ is advanced as principal under the Redevelopment Note as of the date hereof. Such amount has been properly incurred, is a proper charge made or to be made in connection with the redevelopment project costs defined in the Ordinance and has not been the basis of any previous principal advance. As of the date hereof, the outstanding principal balance under the Redevelopment Note is \$, including the amount of this Certificate and less payment made on the Redevelopment Note.				
IN WITNESS WHEREOF, the City has caused this Certification to be signed on its behalf as of (Closing Date).					
	CITY OF CH	ICAGO			
		Commissioner t of Planning and Development			
AUTHE	THENTICATED BY:				
REGIS	GISTRAR	·			

EXHIBIT N

FORM OF SUBORDINATION AGREEMENT

[SAMPLE – UPDATE BASED ON SPECIFIC PROJECT]

This document prepared by and after recording return to:
, Esq.
City of Chicago
Department of Law
121 North LaSalle Street, Room 600

Chicago, IL 60602

SUBORDINATION AGREEMENT

This Subordination Agreement ("Agreement") is made and entered into as of the day of _____, ____ between the City of Chicago by and through its Department of Planning and Development (the "City"), [Name Lender], a [national banking association] (the ALender").

WITNESSETH:

WHEREAS, [Describe Properties of the control of the	oject - us	e la	nguage	form Re	citals of	Redevelop	ment
agreement] the		an	Illinois	[limited	liability	company]	(the
ADeveloper"), has purchased certain prop	erty locate	ed w	ithin the			_Redevelop	ment
Project Area at,	Chicago,	, Illir	nois	and	legally of	described or	າ the
Exhibit hereto (the "Property"), in order to				located o	n the Pro	perty throug	h the
following activities:		(th	e "Proje	ect"); and			
			** 1				

WHEREAS, [describe financing and security documents - leave blanks as necessary if you do not have financing documents - see example below] as part of obtaining financing for the Project, Developer and American National Bank and Trust Company of Chicago, as trustee under Trust Agreement dated November 19, 1996 and known as Trust No. 122332-01 (the ALand Trustee") (Developer and the Land Trustee collectively referred to herein as the ABorrower"), have entered into a certain Construction Loan Agreement dated as of December 29, 1997 with the Lender pursuant to which the Lender has agreed to make a loan to the Borrower in an amount not to exceed \$44,000,000 (the ALoan"), which Loan is evidenced by a Mortgage Note and executed by the Borrower in favor of the Lender (the ANote"), and the repayment of the Loan is secured by, among other things, certain liens and encumbrances on the Property and other property of the Borrower pursuant to the following: (i) Mortgage dated December 29, 1997 and recorded January 2, 1998 as document number 98001840 made by the Borrower to the Lender; and (ii) Assignment of Leases and Rents recorded January 2, 1998 as document number 98001841 made by the Borrower to the Lender (all such agreements referred to above and otherwise relating to the Loan referred to herein collectively as the ALoan Documents");

WHEREAS, Developer desires to enter into a certain Redevelopment Agreement dated the date hereof with the City in order to obtain additional financing for the Project (the

"Redevelopment Agreement," referred to herein along with various other agreements and documents related thereto as the "City Agreements");

WHEREAS, pursuant to the Redevelopment Agreement, Developer will agree to be bound by certain covenants expressly running with the Property, as set forth in Sections [8.02, 8.06 and 8.19] [Note" Refer to Section 7.02 of the Agreement to confirm which covenants to list] of the Redevelopment Agreement (the "City Encumbrances");

WHEREAS, the City has agreed to enter into the Redevelopment Agreement with Developer as of the date hereof, subject, among other things, to (a) the execution by Developer of the Redevelopment Agreement and the recording thereof as an encumbrance against the Property; and (b) the agreement by the Lender to subordinate their respective liens under the Loan Documents to the City Encumbrances; and

NOW, THEREFORE, for good and valuable consideration, the receipt, adequacy and sufficiency of which are hereby acknowledged, the Lender and the City agree as hereinafter set forth:

- 1. <u>Subordination</u>. All rights, interests and claims of the Lender in the Property pursuant to the Loan Documents are and shall be subject and subordinate to the City Encumbrances. In all other respects, the Redevelopment Agreement shall be subject and subordinate to the Loan Documents. Nothing herein, however, shall be deemed to limit the Lender's right to receive, and Developer's ability to make, payments and prepayments of principal and interest on the Note, or to exercise its rights pursuant to the Loan Documents except as provided herein.
- 2. <u>Notice of Default</u>. The Lender shall use reasonable efforts to give to the City, and the City shall use reasonable efforts to give to the Lender, (a) copies of any notices of default which it may give to Developer with respect to the Project pursuant to the Loan Documents or the City Agreements, respectively, and (b) copies of waivers, if any, of Developer's default in connection therewith. Under no circumstances shall Developer or any third party be entitled to rely upon the agreement provided for herein.
- 3. <u>Waivers</u>. No waiver shall be deemed to be made by the City or the Lender of any of their respective rights hereunder, unless the same shall be in writing, and each waiver, if any, shall be a waiver only with respect to the specific instance involved and shall in no way impair the rights of the City or the Lender in any other respect at any other time.
- 4. <u>Governing Law; Binding Effect.</u> This Agreement shall be interpreted, and the rights and liabilities of the parties hereto determined, in accordance with the internal laws and decisions of the State of Illinois, without regard to its conflict of laws principles, and shall be binding upon and inure to the benefit of the respective successors and assigns of the City and the Lender.
- 5. <u>Section Titles; Plurals</u>. The section titles contained in this Agreement are and shall be without substantive meaning or content of any kind whatsoever and are not a part of the agreement between the parties hereto. The singular form of any word used in this Agreement shall include the plural form.
- 6. <u>Notices</u>. Any notice required hereunder shall be in writing and addressed to the party to be notified as follows:

If to the City:	If to Developer:
City of Chicago Department of Planning and Development 121 North LaSalle Street, Room 1000 Chicago, Illinois 60602 Attention: Commissioner	Attention:
With Copies To:	With Copies To:
City of Chicago Department of Law 121 North LaSalle Street, Room 600 Chicago, Illinois 60602 Attention: Finance and Economic Development Division	Attention:

or to such other address as either party may designate for itself by notice. Notice shall be deemed to have been duly given (i) if delivered personally or otherwise actually received, (ii) if sent by overnight delivery service, (iii) if mailed by first class United States mail, postage prepaid, registered or certified, with return receipt requested, or (iv) if sent by facsimile with facsimile confirmation of receipt (with duplicate notice sent by United States mail as provided above). Notice mailed as provided in clause (iii) above shall be effective upon the expiration of three (3) business days after its deposit in the United States mail. Notice given in any other manner described in this paragraph shall be effective upon receipt by the addressee thereof; provided, however, that if any notice is tendered to an addressee and delivery thereof is refused by such addressee, such notice shall be effective upon such tender.

7. <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall constitute an original and all of which, when taken together, shall constitute one instrument.

IN WITNESS WHEREOF, this Subordination Agreement has been signed as of the date first written above.

	[LENDER], [a national banking association]
•	Ву:
	Its:
	CITY OF CHICAGO
	Ву:
	Its: Commissioner, Department of Planning and Development
ACKNOWLEDGED AND AGREED TO THIS DAY OF,	
[Developer], a	
Ву:	
Its:	
Exhibit to Subordination Agreement – Legal Desc	cription

) SS	
COUNTY OF COOK)	
I, the undersigned, a notary public in and for the County and State aforesaid CERTIFY THAT, personally known to me to be the Common Department of Planning and Development of the City of Chicago, Illinois (the "City") known to me to be the same person whose name is subscribed to the foregoinappeared before me this day in person and acknowledged that as such (s) the signed and delivered the said instrument pursuant to authority, as his/her free act, and as the free and voluntary act and deed of said City, for the uses and purpose forth.	nissioner of the and personally ing instrument, Commissioner, and voluntary
GIVEN under my hand and notarial seal this day of,	·
Notary Public	
My Commission Expires	
(SEAL)	
STATE OF ILLINOIS)) SS COUNTY OF COOK)	
I,, a notary public in and for the said Count aforesaid, DO HEREBY CERTIFY THAT, personally known to of [Lender], a, and personally be the same person whose name is subscribed to the foregoing instrument, appear this day in person and acknowledged that he/she signed, sealed and delivered so pursuant to the authority given to him/her by Lender, as his/her free and voluntary free and voluntary act of the Lender, for the uses and purposes therein set forth.	known to me to ared before me aid instrument,
GIVEN under my hand and notarial seal this day of,	<u>-</u> ·
,	
Notary Public	`
My Commission Expires	
(SEAL)	

EXHIBIT O FORM OF PAYMENT BOND

EXHIBIT P

INVESTOR LETTER

	, 20
City of Chicago Department of Planning and Development 121 N. LaSalle Street, Suite 1000 Chicago, Illinois 60602 Attention: Commissioner	
Re: \$[] City of Chicago, Illinois Ta ([] Redevelopment Proje	ax Increment Allocation Revenue Note ect) [Taxable] Series [] issued to
Ladies and Gentlemen:	
The undersigned (the "Investor") is the acquirer undersigned acknowledges that the Note was issurprose of paying the costs of certain eligible recommendately [] (the "Developer") which were in approximately [] square foot [] Redevelopment Project Area Chicago.	sued by the City of Chicago (the "City") for the development project costs incurred by ncurred in connection with the development of the "Project") in the
The undersigned acknowledges that the Note was the City Council of the City on, 2	20 (the "Project Ordinance") and the Agreement dated as of, 20 nent Number in the Office of
In connection with the acquisition of the Note be following representations upon which the City ma	
1. The Investor is a duly under the laws of the State of are execute this letter and any other instruments and Investor in connection with the acquisition of the	documents required to be executed by the
[2. The Investor is an institutional "accredited inve (3) or (7) of the Securities and Exchange Commi- the Securities Act of 1933, as amended (the "Sec within the meaning of Rule 144A of the Commiss	ssion (the "Commission") promulgated under curities Act") or a "qualified institutional buyer"

¹ If the Investor is a broker or dealer and is purchasing the Note with a view toward any distribution, sale or resale of the Note or any beneficial interest therein, replace paragraph 2 with the following.

- [2. The Investor is a "dealer" meeting the requirements of Rule 144A(a)(1)(ii) or (iii) of the Commission promulgated under the Securities Act.]
- 3. The Investor has sufficient knowledge and experience in financial and business matters, including the acquisition and ownership of notes issued by municipalities, to be able to evaluate the merits and risks of its investment in the Note, and the Investor is able to bear any economic risk associated with its investment in the Note.
- [4. The Investor is acquiring the Note for its own account and not with a view toward any distribution, sale or resale of the Note. The Investor intends to hold the Note for an indefinite period of time. The Investor understands that it may need to bear the risks of this investment for an indefinite time, since any sale of the Note prior to maturity may not be possible. ²¹
- ² If the Investor is a broker or dealer and is purchasing the Note with a view toward any distribution, sale or resale of the Note or any beneficial interest therein, replace paragraph 4 with the following:
- [4. The Investor (i) has not solicited offers for, or offered or sold, and will not solicit offers for, or offer and sell the Note or any beneficial interest therein except to persons who it reasonably believes are "qualified institutional buyers" within the meaning of Rule 144A(a)(1) of the Commission promulgated under the Securities Act; (ii) has not solicited offers for, or offered or sold, and will not solicit offers for, or offer or sell Notes by, any form of general solicitation or general advertising or in any manner involving a public offering; and (iii) shall inform each prospective purchaser of the Note or any beneficial interest therein of the restrictions on resale of the Note or beneficial interests therein under the Agreement.]

[¹ If the Investor is a "dealer" meeting the requirements of Section 144A(a)(1) (ii), replace paragraph 4 with the following:

- 4. The Investor (i) has not solicited offers for, or offered or sold, and will not solicit offers for, or offer and sell the Note or any beneficial interest therein, except to persons who it reasonably believes are "qualified institutional buyers" within the meaning of Rule 144A(a)(1) under the Securities Act; (ii) has not solicited offers for, or offered or sold, and will not solicit offers for, or offer or sell Notes by, any form of general solicitation or general advertising or in any manner involving a public offering; and (iii) shall inform each prospective purchaser of the Note or any beneficial interest therein of the restrictions on resale of the Note or beneficial interests therein under the Agreement]
- 5. The Investor understands that the Note is not registered under the Securities Act and that such registration is not legally required as of the date hereof; and the Investor further understands that the Note (a) is not being registered or otherwise qualified for sale under the "Blue Sky" laws and regulations of any state, (b) will not be listed in any stock or other securities exchange, (c) will not carry a rating from any rating service, and (d) will be delivered in a form which is not to be readily marketable.
- 6 The Investor understands that (a) the Note is a limited obligation of the City, payable solely from moneys on deposit in the [_____] Account (as defined in the Project Ordinance); (b) the Note does not constitute an indebtedness of the City within the meaning of any constitutional or statutory provision or limitation; (c) no holder of the Note will have the right to compel the exercise of any taxing power of the City for payment of the principal of, or interest

premium, if any, on the Note; and (d) the Note does not and will not represent or constitute a general obligation or a pledge of the faith and credit of the City, the State of Illinois or any political subdivision thereof.

- 7. The Investor has read and considered the risk factors set forth in "Noteholder Risks" attached hereto.
- 8. The Investor has not relied upon the City for any information in connection with its acquisition of the Note. The Investor has either been supplied with or been given access to information, including financial statements and other financial information, to which a reasonable investor would attach significance in making investment decisions, and the Investor has had the opportunity to ask questions and receive answers from knowledgeable individuals concerning the Developer, the Project and the Note. The Investor is in possession of all the information and material necessary to evaluate the merits and risks of the acquisition of the Note.
- 9. The Investor has been furnished with and has examined the Agreement and other documents, certificates and the legal opinions delivered in connection with the issuance of the Note. The Investor acknowledges that neither the City nor the Developer has prepared an offering document with respect to the Note. The Investor has made its own inquiry and analysis with respect to the Note and material factors affecting the payment of the Note.
- 10. The Investor acknowledges that with respect to the Notes, the City has no obligation to provide any continuing disclosure to the Electronic Municipal Market Access System maintained by the Municipal Securities Rulemaking Board, any holder of the Note or any other person under Rule 15c2-12 of the Commission promulgated under the Securities Exchange Act of 1934 or otherwise, and shall have no liability with respect thereto.
- 11. The Investor understands that the City, the Developer, their respective counsel and Bond Counsel will rely upon the accuracy and truthfulness of the representations and warranties contained herein and hereby consents to such reliance.

Very truly yours,		
],
a[]	
Ву:		
Name:		_
Title:		

[NOTE: Use applicable definition: (Available Increment Taxes v. Incremental Taxes)]

NOTEHOLDER RISKS

The purchase of or investment in the Note involves certain risks. Each prospective holder or purchaser of the Note, or any interest therein, should make an independent evaluation of the financial and business risks associated with holding or having an investment interest in the Note. Certain of these risks are set forth below. The following summary is not intended to be complete and does not purport to identify all possible risks that should be considered by prospective holders of the Note or any interests therein. Capitalized terms used herein have the meanings set forth in the Note.

All prospective holders of the Note are urged to consult with their financial adviser and legal counsel before acquiring the Note or any interest therein.

Loss of Investment

Investment in the Note involves a high degree of risk. It is suitable only for persons who are able to bear the economic risks of the investment, including total loss. No assurance can be provided that prospective holders of the Note will not lose their entire investment in the Note.

Lack of Liquidity

The Note is suitable only for persons who have no need for liquidity. The transferability of the Note is restricted. The Note may only be transferred in the manner and subject to the limitations provided in the Redevelopment Agreement. Investors in the Note must be prepared to hold the Note until the maturity of the Note.

Reliance on Projections

The City does not endorse projections of any kind from any source as to the sufficiency of available incremental taxes to pay principal and interest on the Note. Investor who rely on any such projections do so at their own risk.

The City's Office of Budget and Management ("OBM") produces five-year District Projection Reports for each TIF district in the City for the purpose of evaluating resources and project balances. This information, which is currently publicly available, is used by the OBM to determine how much funding has been committed and how much funding is available for potential projects. The reports and the projections including therein are not audited and do not represent a final accounting of funds. The reports are not prepared for investors or as a basis for making investment decisions with respect to any notes, bonds or other debt obligations of the City that are payable from available incremental taxes, including the Note. Prospective investors in Note are cautioned not to rely on any of the information contained in the District Projection Reports.

Limited Obligations

THE NOTE IS A SPECIAL LIMITED OBLIGATION OF THE CITY PAYABLE SOLELY FROM THE AVAILABLE INCREMENTAL TAXES, AND SHALL BE A VALID CLAIM ONLY AGAINST SAID SOURCES. THE NOTE DOES NOT CONSTITUTE AN INDEBTEDNESS OR A

LOAN AGAINST THE GENERAL TAXING POWERS OR CREDIT OF THE CITY, WITHIN THE MEANING OF ANY CONSTITUTIONAL OR STATUTORY PROVISION. THE NOTEHOLDER HAS NO RIGHT TO COMPEL ANY EXERCISE OF THE TAXING POWER OF THE CITY, THE STATE OF ILLINOIS OR ANY POLITICAL SUBDIVISION THEREOF TO PAY THE PRINCIPAL OR INTEREST OF THE NOTE.

There can be no assurance that Available Incremental Taxes will be sufficient for payment of amounts due and owing on the Note.

Limited Information

The Note was issued to the Developer under the Redevelopment Agreement as part of a commercial transaction negotiated by the Developer and the City. [The Developer] engaged a [consultant] to deliver a [feasibility report][projection report] to the City in connection with the Project, which included certain information about the Project Area, the Project and historical and projected Available Incremental Taxes.] The report contained information as of its date only, and neither the Developer nor any other party have any obligation to update the report as of any subsequent date.

The City is under no continuing obligation to provide to any holder or prospective holder of the Note, or to post to EMMA or any website, any current or updated information with respect to the Project Area, the Project, the historical and projected Available Incremental Taxes or the Note. The City does not prepare or have readily available any current or updated information about the Project Area, the Project or the Available Incremental Taxes.

Unavailability of City Funds

The City is not obligated to pay principal of or interest on the Note in any year in which there are inadequate Available Incremental Taxes. The City is obligated to pay the amount of any unpaid principal or accrued interest in any subsequent year but only to the extent of the Available Incremental Taxes for those subsequent years. If, on the maturity date of the Note, any outstanding unpaid principal or interest on the Note exists for any reason, including without limitation the inadequacy of Available Incremental Taxes, such outstanding principal and/or interest will be forgiven in full and the City will have no further obligation to pay such outstanding amount. In such event, there would be no further payments of principal or interest in respect of the Note.

Risk of Failure to Maintain Levels of Assessed Valuation

There can be no assurance that the equalized assessed value of the Project property will remain the same throughout the term of the Note. Furthermore, the successful petition or application of any owner for the reduction of the assessed value of the Project property may cause the equalized assessed value of the Project Area to be less than the originally projected equalized assessed value of the property. If any time during the term of the Note the actual equalized assessed value is less than what was projected, the generation of Available Incremental Taxes for payment on the Note is likely to be significantly impaired.

Risk of Change in Available Incremental Taxes

Prospective holders of the Note should carefully consider, among other factors, the risks associated with the ultimate generation of Available Incremental Taxes in the Project Area. These risks include, but are not limited to, the following:

- 1. Property tax rates are calculated by the County Clerk for numerous funds of a number of taxing districts that tax all or part of the property in the Project Area. A reduction in the tax levies by the affected taxing districts may have an adverse effect on the Available Incremental Taxes.
- 2. Further changes may be made in the real property tax system by the State of Illinois or Cook County. Such changes could include various property tax rollbacks, abatements, exemptions, changes in the ratio of assessment, or relief measures, limitations on the amount or percent of increase in tax levies by taxing districts, or other measures that would limit the tax levy amount that could be extended to the property within the Project Area and, consequently, the projected Available Incremental Taxes generated. For example, if Illinois adopted practices used in other states, the property tax system could be changed so that schools would be financed from a source other than property taxes. This type of change could have a significant adverse effect upon Available Incremental Taxes.
- 3. Cook County's methodology and procedures used to assess the value of property may be altered resulting in a potentially reduced or altered valuation in a particular year or succession of years.
- 4. Failure by Cook County to remit property taxes to the City on a timely basis could result in insufficient Available Incremental Taxes being available to pay principal of or interest on the Note when due.

FUTURE LEGISLATION, REGULATIONS, GOVERNMENTAL OR JUDICIAL INTERPRETATION OF REGULATIONS OR LEGISLATION OR PRACTICES AND PROCEDURES RELATED TO PROPERTY TAX ASSESSMENT, LEVY, COLLECTIONS OR DISTRIBUTION COULD HAVE A MATERIAL EFFECT ON THE CALCULATION OR AVAILABILITY OF AVAILABLE INCREMENTAL TAXES COLLECTED OR DISTRIBUTED.

Changes in Multiplier and Tax Rate

The equalization factor annually determined by the Illinois Department of Revenue for properties located within Cook County (commonly referred to as the "multiplier") may vary substantially in future years. A decrease in the multiplier would reduce the equalized assessed value of the taxable real property in the Project Area and, therefore, the Available Incremental Taxes available to pay debt service on the Note. The future tax rates of the units of local government levying taxes in the Project Area either individually or on a composite basis, may differ from their historical levels. Any decrease in the composite tax rate of the governmental units would decrease the amount of Available Incremental Taxes available to pay debt service on the Note. Any decrease in the composite tax rate of the governmental units could occur in future years as a result of various factors, including, but not limited to, one or more of the following: (a) reduced governmental costs; (b) constitutional or statutory spending or tax rate limitations; or (c) governmental reorganization or consolidation.

Economic Risks Affecting Available Incremental Taxes

Changing economic circumstances or events in the Project Area may result in reductions in Available Incremental Taxes available to pay debt service on the Note. Relocations of major property owners to sites outside the Project Area or sales of major properties to tax-exempt entities could reduce the assessed valuation of the Project Area. Substantial damage to or destruction of improvements within the Project Area could cause a material decline in assessed valuation and impair the ability of the taxpayers in the Project Area to pay their respective portions of real estate taxes. There can be no assurance that the improvements in the Project Area are or will be insured under fire and extended coverage insurance policies, and, even if such insurance exists, the proceeds thereof will not be assigned as security for the payment of real estate taxes or to secure payment of the Note. In addition, any insurance proceeds may not be sufficient to repair or rebuild the improvements. The restoration of the improvements may be delayed by other factors, or the terms of then-applicable mortgage financing could require the application of insurance proceeds to the reduction of mortgage balances. Any of the foregoing circumstances could result in the assessed valuation of property in the Project Area remaining depressed for an unknown period of time and decrease the amount of Available Incremental Taxes available to pay debt service on The Note.

Results of operation of properties within the Project Area depend, in part, on sales, leases, rental rates and occupancy levels, which may be adversely affected by competition, suitability of the properties located in the Project Area in its local market, local unemployment, availability of transportation, neighborhood changes, crime levels in the Project Area, vandalism, rising operating costs and similar factors. Poor operating results of properties within the Project Area may cause delinquencies in the payment of real estate taxes, reduce assessed valuations and increase the risk of foreclosures. Successful petitions by taxpayers to reduce their assessed valuations could adversely affect available incremental Taxes available for payment of the Note.

Failure to Sell or Lease Property

At the time the Note was issued, the redevelopment plan called for the Developer [to sell/lease to commercial or industrial enterprises/retailers] prior to completion of the Project. The slowdown, stoppage or failure of the Developer to complete the Project and to successfully sell/lease the Project could delay or reduce the amount of Available Incremental Taxes generated in the Project Area. Such delay or reduction could lead to a default in payments of the principal of, and interest on, the Note.

Reliance on Primary Taxpayers

If one or only a few property owners within the Project Area are responsible for generating a substantial amount of the Available Incremental Taxes, the generation of Available Incremental Taxes could be significantly adversely affected if such owner or owners and/or their tenants discontinue or curtail their businesses, terminate or default on their leases and substitutes or replacements cannot be found or located on a timely basis.

Force Majeure Conditions

Riots, civil disturbances, vandalism, fires, and natural disasters or other "Acts of God" affecting the conditions and viability of the Project Area may reduce or eliminate the receipt of Available Incremental Property.

Contiguous Project Areas

The Project Area is contagious with other redevelopment areas designated by the City pursuant to the TIF Act and may become contiguous with others. The TIF Act allows the City to expend incremental taxes collected from the Project Area which are in excess of the amounts required in each year to pay and secure obligations issued and project costs incurred with respect to the Project Area to pay for costs eligible for payment under the TIF Act which are incurred in such contiguous areas. In the event Incremental Taxes from the Project Area in excess of Available Incremental Taxes and the amounts required to (i) pay principal and interest coming due on the Note in any year and (ii) be deposited in other funds and accounts maintained under the Redevelopment Agreement are allocated to a contiguous project redevelopment area, such excess incremental taxes will not be available to remedy any future failure to pay principal of and interest on the Note.

Risk of Delay in Payment of Available Incremental Taxes

The failure of current or future owners of property in the Project Area to remit property taxes to the City when due or the failure of the City to timely remit Available Incremental Taxes to the Noteholder could result in insufficient Available Incremental Taxes being available to pay principal of or interest on the Note when due.

Delays in Exercising Remedies

The enforceability of the Note is subject to applicable bankruptcy laws, equitable principles affecting the enforcement of creditors' rights generally and of liens securing such rights, and the police powers of the State of Illinois and its political subdivisions. Because of delays inherent in obtaining judicial remedies, it should not be assumed that these remedies could be accomplished rapidly.

Remedies available to holder of the Note may be limited by a variety of factors and may be inadequate to assure the timely payment of principal of and interest on the Note, or to preserve the tax-exempt status of The Note. The Note is not subject to acceleration due to payment default. Lack of remedies may entail risks of delay, limitation, or modification of the rights of the holders of the Note. Judicial remedies, such as foreclosure and enforcement of covenants, are subject to exercise of judicial discretion.

Risk of Transferee Becoming a Debtor in Bankruptcy

If a transferee of the Note were to become a debtor under the United States Bankruptcy Code or applicable state laws, a creditor or trustee in bankruptcy of the transferee might argue that the sale of the Note by the transferee constituted a fraudulent conveyance or a pledge of the Note rather than a sale. If such positions were accepted by a court, then delays in principal and interest payments to holder the Note could occur or reductions in the amounts of such payments could result. Additionally, if the transfer of the Note is re-characterized as a pledge, then a tax lien, governmental lien or other lien created by operation of law on the property of the transferee could have priority over the holder's interest in the Note.

Loss of Tax Exemption

Interest on the Note could become includible in gross income for federal income tax purposes retroactive to the date of issuance of the Note as a result of a failure of the City to comply

with certain provisions of the Internal Revenue Code of 1986, as amended (the "Code"). An event of taxability does not trigger a mandatory redemption of the Note, and the Note will remain outstanding to maturity or until redeemed.

THE ABOVE IS NOT INTENDED TO BE A COMPREHENSIVE DISCUSSION OF ALL POTENTIAL RISKS ASSOCIATED WITH THE NOTE.

* * * * *

43RD AND CALUMET PHASE I LP

02021-3265



CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT ----AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing P	arty submitt	ing this EDS	. Include d/b/a/ if applicable:
43RD AND CALUMET PHASE I L	.P		
Check ONE of the following thre	ee boxes:		
the contract, transaction or other un "Matter"), a direct or indirect interename: OR	nolding, or a ndertaking to est in excess ect or indirec	nticipated to which this of 7.5% in t	hold within six months after City action on EDS pertains (referred to below as the he Applicant. State the Applicant's legal
B. Business address of the Disclos	sing Party:	350 W. Hubba	ard St., Ste. 500
		Chicago, IL 60	354
C. Telephone: 312 527 5400	Fax: 312.52	27 7440	Email:Email:
D. Name of contact person: Jeff Hear	d		
E. Federal Employer Identification	n No. (if you	ı have one)	
			ns. (Include project number and location of
Appaicar on By Applicant for various forms of City financial assistance tappo Multi-Program Funds (probably HOME), purchase of City property in reduce	emately \$5,000 000 of Ta g price; to facilitate develor	k Increment Financing (TTIF) expent of property located at	, 775 Crears (LIHTUs and IAHTOs). approxima bly \$2,000 bln. City Lean of 43rd St. and S. Callur et Ave. (
G. Which City agency or department	ent is reques	ting this EDS	2) Department of Housing / Department of Planning and Development
If the Matter is a contract being has complete the following:	ndled by the	City's Depa	rtment of Procurement Services, please
Specification # N/A		_ and Contra	et # <u>N/A</u>
Ver.2018-1	Pa	ige 1 of 15	

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE	E DISCLOSING PART	Y
Person	p	Limited liability company Limited liability partnership Joint venture Not-for-profit corporation (Is the not-for-profit corporation also a 501(c)(3))? Yes No [] Other (please specify)
2. For legal entities, t	he state (or foreign cour	ntry) of incorporation or organization, if applicable:
Delaware		
-	ot organized in the State f Illinois as a forcign en	e of Illinois: Has the organization registered to do atity?
✓ Yes	No	Organized in Illinois
B. IF THE DISCLOS	ING PARTY IS A LEG	SAL ENTITY:
the entity; (ii) for not- are no such members, similar entities, the tr limited partnerships, each general partner, n	for-profit corporation write "no members whi ustee, executor, adminis limited liability comp	oplicable, of: (i) all executive officers and all directors of s, all members, if any, which are legal entities (if there ch are legal entities"); (iii) for trusts, estates or other strator, or similarly situated party; (iv) for general or anies, limited liability partnerships or joint ventures, ager or any other person or legal entity that directly or at of the Applicant.
NOTE: Each legal ent	ity listed below must su	ubmit an EDS on its own behalf.
Name 43rd and Calumet Phase I LLC	<u>;</u>	Title General Partner
Entity has no individuals as Ex	ecutive Officers or Directors	
		concerning each person or legal entity having a direct or nonths after City action) beneficial interest (including
ownership) in excess of	of 7.5% of the Applicant	t. Examples of such an interest include shares in a hip or joint venture, interest of a member or manager in a

Name Habitat 43 Green LLC*	Business Address 350 W. Hubbard St., Ste. 500, Chicago, IL 60654	Percentage Interest in the Applicant 99,99%
SEE EXHIBIT "A" FOR	OTHER PERSONS OR ENTITIES WITH INTEREST IN E	XCESS OF 7.5% OF APPLICANT
*It is anticipated that at	closing the limited partner will be replaced by Stratford 43	Green Investors Limited Partnership
OFFICIALS	•	OR OWNERSHIP BY, CITY ELECTE
	g Party provided any income or compension preceding the date of this EDS?	ation to any City elected official during the Yes Vo
	ng Party reasonably expect to provide any uring the 12-month period following the d	
•	of the above, please identify below the natione or compensation:	me(s) of such City elected official(s) and
inquiry, any City	ected official or, to the best of the Disclosic elected official's spouse or domestic partithe Municipal Code of Chicago ("MCC")	ner, have a financial interest (as defined in
1 68		

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained) Busin Address Address		Fees (indicate whether paid or estimated:) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
See attached Ex. B		
(Add sheets if necessary)		
Check here if the Disclosing	Party has not retained, nor expects to	retain, any such persons or entities.
SECTION V CERTIFICAT	IONS	
A. COURT-ORDERED CHILD	SUPPORT COMPLIANCE	
	ubstantial owners of business entities child support obligations throughout t	
* *	ndirectly owns 10% or more of the Di bligations by any Illinois court of com	
Yes No No pers	son directly or indirectly owns 10% or	more of the Disclosing Party.
If "Yes," has the person entered is the person in compliance with	into a court-approved agreement for path that agreement?	ayment of all support owed and
Yes No		
B. FURTHER CERTIFICATIO	NS	
Procurement Services.] In the 5-Party nor any Affiliated Entity [green performance of any public contrainspector general, or integrity coinvestigative, or other similar ski	ly if the Matter is a contract being hand-year period preceding the date of this see definition in (5) below] has engage act, the services of an integrity monito in the mpliance consultant (i.e., an individually, designated by a public agency to have as well as help the vendors reform	EDS, neither the Disclosing ed, in connection with the or, independent private sector all or entity with legal, auditing, nelp the agency monitor the

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

can be considered for agency contracts in the future, or continue with a contract in progress).

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity:
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below: N/A
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusivel presumed that the Disclosing Party certified to the above statements.
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the Cit of Chicago (if none, indicate with "N/A" or "none").
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointe official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient.
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
1. The Disclosing Party certifies that the Disclosing Party (check one) is is not
a "financial institution" as defined in MCC Section 2-32-455(b).
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to

believe has not provided or cannot provide truthful certifications.

Ver 2018-1

predatory lender may result in the loss of the privilege of doing business with the City."

MCC Section 2-32-455(b here (attach additional pa)) is a predatory lender wigges if necessary):	because it or any of its affiliates (as defined in thin the meaning of MCC Chapter 2-32, explain
		e appears on the lines above, it will be tified to the above statements.
D. CERTIFICATION RE	EGARDING FINANCIAL	INTEREST IN CITY BUSINESS
Any words or terms defin	ed in MCC Chapter 2-156	have the same meanings if used in this Part D.
after reasonable inquiry, of		the best of the Disclosing Party's knowledge yee of the City have a financial interest in his or entity in the Matter?
Yes	No	
	Yes" to Item D(1), procee D(2) and D(3) and proceed	d to Items D(2) and D(3). If you checked "No" to Part E.
official or employee shall other person or entity in the taxes or assessments, or ("City Property Sale"). Co	have a financial interest in the purchase of any propertion iii) is sold by virtue of legal compensation for property t	bidding, or otherwise permitted, no City elected his or her own name or in the name of any y that (i) belongs to the City, or (ii) is sold for all process at the suit of the City (collectively, taken pursuant to the City's eminent domain the meaning of this Part D.
Does the Matter involve a	City Property Sale?	
Yes	No	
		names and business addresses of the City officials names and business addresses of the City officials
Name	Business Address	Nature of Financial Interest

4. The Disclosing Party further certifies that no prohibited financial interest in the Matter will be acquired by any City official or employee.

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.
1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.
A. CERTIFICATION REGARDING LOBBYING
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)

by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Page 9 of 15

Ver.2018-1

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue; renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs $\Lambda(1)$ through $\Lambda(4)$ above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the Applicant?
✓ Yes No
If "Yes," answer the three questions below:
1. Have you developed and do you have on file affirmative action programs pursuant to applicable federal regulations? (See 41 CFR Part 60-2.) Yes No
2. Have you filed with the Joint Reporting Committee, the Director of the Office of Federal Contract Compliance Programs, or the Equal Employment Opportunity Commission all reports due under the applicable filing requirements? Yes □No □Reports not required
3. Have you participated in any previous contracts or subcontracts subject to the equal opportunity clause? ✓ Yes No
If you checked "No" to question (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities sceking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period. as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

(See Opposite Signature Block)	43RD AND CALUMET PHASE I LP
(Print or type exact legal name of Disclosing Party	By: 43rd and Calumet Phase I LLC, its General Partner
By:	By: 43 Green JV LLC, its Manager/Sole Member
(Sign here)	By. Habitat 43 Green LLC, its Managing Member
(Print or type game of person signing)	By: Habitat Acquisitions Company LLC, its Manager
(Print or type name of person signing)	By: The Habitat Company LLC, its Manager
(Print or type title of person signing)	By Matthew G. Fiascone Its: President
Signed and sworn to before me on (date) May	11,2021,
at COOK County, TLL/NO/s (state	de).
There Thereine Chalds	;·····
Notary Public	OFFICIAL SEAL LORI FRANCINE CHACOS
0	NOTARY PUBLIC - STATE OF ILLINOIS MY COMMISSION EXPIRES 10/21/24
Commission expires: $\frac{10}{21}\frac{24}{24}$	

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

	g Party or any "Applicable Party" or any Spouse or Domestic Partner thereof ial relationship" with an elected city official or department head?
Yes	No
which such person is c	fy below (1) the name and title of such person, (2) the name of the legal entity to onnected; (3) the name and title of the elected city official or department head to a familial relationship, and (4) the precise nature of such familial relationship.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

1. Pursuant to MCC Section 2-154-010, is the Applicant or any Owner identified as a building conscious or problem landlord pursuant to MCC Section 2-92-416?	ode
☐ Yes	
2. If the Applicant is a legal entity publicly traded on any exchange, is any officer or director of the Applicant identified as a building code scofflaw or problem landlord pursuant to MCC Sectio 2-92-416?	n
Yes No The Applicant is not publicly traded on any exchange	ţe.
3. If yes to (1) or (2) above, please identify below the name of each person or legal entity identifias a building code scofflaw or problem landlord and the address of each building or buildings to the pertinent code violations apply.	
· · · · · · · · · · · · · · · · · · ·	

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

Yes
No
$\sqrt{N/A}$ – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

EXHIBIT "A"

City of Chicago Economic Disclosure and Affidavit The Habitat Company LLC SECTION II: DISCLOSURE OF OWNERSHIP INTERESTS

B. 2 Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None"

NAME	BUSINESS ADDRESS	PERCENTAGE INTEREST IN APPLICANT
HABITAT 43 GREEN LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	99.99%
HABITAT ACQUISITIONS COMPANY LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	89.99%
GENERATIONS HOUSING INITIATIVES	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	9.99%
THE HABITAT COMPANY LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	89.99%
Daniel E. Levin	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	
Matthew G. Fiascone	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	

EXHIBIT "B"

43rd and Calumet Phase I LP City of Chicago Economic Disclosure and Affidavit

SECTION IV: DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

Name (indicated whether retained or anticipated to be retained	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response
Landon Bone Baker Architects (retained)	1625 W. Carroll Avenue Chicago IL 60612	Architect	\$1,260,000.00 (estimated) \$435,461 (paid)
Neal and Leroy, LLC (retained)	120 N. LaSalle Street Suite 2600 Chicago, IL 60602	Attorney	\$110,000 (estimated) \$105,344 (paid)
Carnow, Conibear & Assoc., Ltd. (retained)	600 W. Van Buren Street Suite 500 Chicago, IL 60607	Environmental	\$20,000 (estimated) \$15,460 (paid)
Earnest R. Sawyer Enterprises, Inc. (retained)	100 N. LaSalle Street Suite 1515 Chicago, IL 60602	TIF Consulting	\$30,000 (estimated)
IRR Chicago (retained)	400 E. Randolph Street Chicago, IL 60601	Appraisal and Market Study	\$14,500 (estimated) \$6,500 (paid)
TERRA Engineering Ltd. (retained)	225 W. Erie Street Chicago, IL 60654	Survey, Civil Engineering	\$8,380 (paid)
Bowa/McHugh JV (retained)	1040 W. Adams Street, Suite 605 Chicago, IL 60607	General Contractor	\$27,841,790 (estimated)
Building Permits, Inc. (retained)	55 E. Monroe Suite 3800 Chicago, IL 60603	Permit Expeditor	\$15,400 (estimated)
Katten Muchin Rosenman LLP (retained)	525 W. Monroe Street Chicago, IL 60661-3693	Legal – Partnership	\$100,000 (estimated)

43RD AND CALUMET PHASE I LLC

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclos	ing Party submitt	ing this EDS	S. Include d/b/a/ if applicable:
43RD AND CALUMET PHAS	SEILLC		
Check ONE of the following	three boxes:		
the contract, transaction or of "Matter"), a direct or indirect name: OR	ntly holding, or and an er undertaking to interest in excess and direct or indirect	nticipated to which this of 7.5% in the tright of co	hold within six months after City action on EDS pertains (referred to below as the the Applicant. State the Applicant's legal and the Applicant (see Section II(B)(1))
- Add MARKAMAP - ADD	inclusion Doutes	350 W. Hubb	ard St., Ste. 500
B. Business address of the D	isciosing Party:	Chicago, IL 60	
C. Telephone: 312 527 5400	Fax: 312 52	27 7440	Email:Email:
D. Name of contact person:	eff Head		
E. Federal Employer Identifi	cation No. (if you	have one):	· · · · · · · · · · · · · · · · · ·
F. Brief description of the M property, if applicable):	atter to which this	s EDS pertai	ins. (Include project number and location of
Application by Applicant for various forms of Cry financial assists Muti-Program Fories (probable FOME), purchase of City proper	ince (approximately \$5,000,000 of Tax y at reduced price) to facilitate develop	Increment Financing (*TIF iment of property located a	'], Tax Credis (LIHTOs and IAHTOs) approximately \$3,000,000 City Loan of at 43rd St, and S. Calumet Ave.)
G. Which City agency or dep	artment is request	ting this ED	\$2 Department of Housing / Department of Planning and Development
If the Matter is a contract being complete the following:	ng handled by the	City's Depa	artment of Procurement Services, please
Specification # N/A		and Contra	act # N/A
Ver.2018-1		ge 1 of 15	

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY 1. Indicate the nature of the Disclosing Party: Person Limited liability company Limited liability partnership Publicly registered business corporation Privately held business corporation Joint venture Sole proprietorship Not-for-profit corporation $\overline{\text{US}}$ the not-for-profit corporation also a 501(c)(3))? General partnership Limited partnership Yes No Trust Other (please specify) 2. For legal entities, the state (or foreign country) of incorporation or organization, if applicable: Illinois 3. For legal entities not organized in the State of Illinois: Has the organization registered to do business in the State of Illinois as a foreign entity? Organized in Illinois Yes No B. IF THE DISCLOSING PARTY IS A LEGAL ENTITY: List below the full names and titles, if applicable, of: (i) all executive officers and all directors of 1. the entity; (ii) for not-for-profit corporations, all members, if any, which are legal entities (if there are no such members, write "no members which are legal entities"); (iii) for trusts, estates or other similar entities, the trustee, executor, administrator, or similarly situated party; (iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant. **NOTE**: Each legal entity listed below must submit an EDS on its own behalf. Name Title 43 Green JV LLC Manager/Sole Member Entity has no individuals as Executive Officers or Directors

2. Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

state "None."	company, or interest of a beneficiary	of a trust, estate or other similar entity. If none,
NOTE: Each leg	gal entity listed below may be require	ed to submit an EDS on its own behalf.
Name	Business Address	Percentage Interest in the Applicant
NONE		
SECTION III - OFFICIALS	- INCOME OR COMPENSATION	N TO, OR OWNERSHIP BY, CITY ELECTED
	ng Party provided any income or cord preceding the date of this EDS?	npensation to any City elected official during the Yes No
	sing Party reasonably expect to providuring the 12-month period following	de any income or compensation to any City the date of this EDS? Yes No
	of the above, please identify below to come or compensation:	the name(s) of such City elected official(s) and
, ,	· · · · · · · · · · · · · · · · · · ·	isclosing Party's knowledge after reasonable c partner, have a financial interest (as defined in
	f the Municipal Code of Chicago ("M	
,	identity below the name(s) of such C escribe the financial interest(s).	ity elected official(s) and/or spouse(s)/domestic

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

•	usiness .ddress	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(Add sheets if necessary)			
Check here if the Disclos SECTION V CERTIFIC		has not retained, nor expects to re	tain, any such persons or entities.
A. COURT-ORDERED CH	IILD SUP	PORT COMPLIANCE	
		ntial owners of business entities th support obligations throughout the	
• •		ctly owns 10% or more of the Disc ions by any Illinois court of compe	•
☐ Yes ✓ No ☐ No	person di	rectly or indirectly owns 10% or m	nore of the Disclosing Party.
If "Yes," has the person enter is the person in compliance v		court-approved agreement for pay agreement?	ment of all support owed and
Yes No			
B. FURTHER CERTIFICA	TIONS	•	
Procurement Services.] In the Party nor any Affiliated Entiperformance of any public coinspector general, or integrity investigative, or other similar	he 5-year ity <u>[see</u> de ontract, th y complia ir skills, d	ne Matter is a contract being handle period preceding the date of this E finition in (5) below] has engaged, he services of an integrity monitor, ance consultant (i.e., an individual of esignated by a public agency to he s well as help the vendors reform t	DS, neither the Disclosing, in connection with the independent private sector or entity with legal, auditing. Ip the agency monitor the

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

can be considered for agency contracts in the future, or continue with a contract in progress).

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- c. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government. including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity:
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1 Page 6 of 15

11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below: N/A
If the letters "NA." the word "None," or no response appears on the lines above, it will be conclusivel presumed that the Disclosing Party certified to the above statements.
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the Cit of Chicago (if none, indicate with "N/A" or "none").
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointe official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient.
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
1. The Disclosing Party certifies that the Disclosing Party (check one) is is not
a "financial institution" as defined in MCC Section 2-32-455(b).
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a predatory lender may result in the loss of the privilege of doing business with the City."

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to believe has not provided or cannot provide truthful certifications.

) is a predatory lender with	because it or any of its affiliates (as defined in nin the meaning of MCC Chapter 2-32, explain
		appears on the lines above, it will be ified to the above statements.
D. CERTIFICATION RE	GARDING FINANCIAL	INTEREST IN CITY BUSINESS
Any words or terms define	ed in MCC Chapter 2-156 l	have the same meanings if used in this Part D.
after reasonable inquiry, d		the best of the Disclosing Party's knowledge see of the City have a financial interest in his or entity in the Matter?
Yes	✓No	
	Yes" to Item D(1), proceed 10(2) and D(3) and proceed 1	to Items D(2) and D(3). If you checked "No" to Part E.
official or employee shall other person or entity in the taxes or assessments, or (in "City Property Sale"). Co	have a financial interest in ne purchase of any property ii) is sold by virtue of legal	bidding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, ken pursuant to the City's eminent domain he meaning of this Part D.
Does the Matter involve a	City Property Sale?	
Yes	No	
		nmes and business addresses of the City officials ify the nature of the financial interest:
Name	Business Address	Nature of Financial Interest

4. The Disclosing Party further certifies that no prohibited financial interest in the Matter will be acquired by any City official or employee.

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Ver.2018-1

the City and proceeds of debt obligations of the City are not federal funding. A. CERTIFICATION REGARDING LOBBYING 1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Dis Party with respect to the Matter: (Add sheets if necessary): (If no explanation appears or begins on the lines above, or if the letters "NA" or if the word appear, it will be conclusively presumed that the Disclosing Party means that NO persons or registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying co	sclosing "None" r entities
the City and proceeds of debt obligations of the City are not federal funding. A. CERTIFICATION REGARDING LOBBYING 1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Discourse Act of 1995.	·
the City and proceeds of debt obligations of the City are not federal funding.	ated by
	area by
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits alloc	ated by
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) ab Disclosing Party has found records of investments or profits from slavery or slaveholder inspolicies. The Disclosing Party verifies that the following constitutes full disclosure of all strecords, including the names of any and all slaves or slaveholders described in those records.	urance ich
1. The Disclosing Party verifies that the Disclosing Party has searched any and all receive the Disclosing Party and any and all predecessor entities regarding records of investments of from slavery or slaveholder insurance policies during the slavery era (including insurance prissued to slaveholders that provided coverage for damage to or injury or death of their slave the Disclosing Party has found no such records.	r profits olicies
	y in

Page 9 of 15

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the Applicant? Yes No
If "Yes," answer the three questions below:
1. Have you developed and do you have on file affirmative action programs pursuant to applicable federal regulations? (See 41 CFR Part 60-2.) Yes No
2. Have you filed with the Joint Reporting Committee, the Director of the Office of Federal Contract Compliance Programs, or the Equal Employment Opportunity Commission all reports due under the applicable filing requirements? Yes Reports not required
3. Have you participated in any previous contracts or subcontracts subject to the equal opportunity clause? Yes No
If you checked "No" to question (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

43RD AND CALUMET PHASE ILLC

(See Opposite Signature Block)	By. 43 Green JV LLC, its Manager/Sole Member
(Print or type exact legal name of Disclosing Party)	By Habitat 43 Green LLC, its Managing Member
By:(Sign here)	By Habitat Acquisitions Company LLC, its Manager By: The Habitat Company LLC, its Manager
(Print or type name of person signing)	Name Matthew G. Fiascone Its: President
(Print or type title of person signing)	·
Signed and sworn to before me on (date) Nyus at County, The Nois (state) Notary Public	OFFICIAL SEAL LORI FRANCINE CHACOS
Commission expires: $\frac{10}{31}$	NOTARY PUBLIC - STATE OF ILLINOIS MY COMMISSION EXPIRES 10/21/24

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?

Yes

You

If yes, please identify below (1) the name and title of such person, (2) the name of the legal entity to which such person is connected; (3) the name and title of the elected city official or department head to whom such person has a familial relationship, and (4) the precise nature of such familial relationship.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

1. Pursuant to MCC Section 2-154-010, is the Applicant or any Owner identified as a building code scofflaw or problem landlord pursuant to MCC Section 2-92-416?
☐Yes ✓ No
2. If the Applicant is a legal entity publicly traded on any exchange, is any officer or director of the Applicant identified as a building code scofflaw or problem landlord pursuant to MCC Section 2-92-416?
☐ Yes ☐ No ☐ The Applicant is not publicly traded on any exchange.
3. If yes to (1) or (2) above, please identify below the name of each person or legal entity identified as a building code scotflaw or problem landlord and the address of each building or buildings to which the pertinent code violations apply.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

Yes
No
$\sqrt{N/A}$ – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

43 GREEN JV LLC

02021-3265

(3)

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing Party submitti	ng this EDS. Include d/b/a/ if applicable:
43 GREEN JV LLC	
Check ONE of the following three boxes:	
the contract, transaction or other undertaking to "Matter"), a direct or indirect interest in excess name: OR	right of control of the Applicant (see Section II(B)(1))
B. Business address of the Disclosing Party:	350 W. Hubbard St., Ste. 500
B. Business ductiess of the Disclosing Farty.	Chicago, IL 60654
C. Telephone: 312 527 5400 Fax: 312 527	7 7440 Email: jhead1@habitat.com
D. Name of contact person: Jeff Head	
E. Federal Employer Identification No. (if you	have one):
F. Brief description of the Matter to which this property, if applicable):	EDS pertains. (Include project number and location of
Application (iv) Applicant for various forms of City Brancial assistance (approximately \$5,000,000 of Fax libit to the form Funds rarebably HOME), purchase of City property at request, priority fundating development.	nor makint Financing (TIIF1), To+ Credits (LIHTOs and IAHTOs), approximately \$3,000,000 City Lipn of tend of coroniz location at 4,00 St. and S. Callinn (Avk.)
G. Which City agency or department is requesti	ing this EDS? Department of Housing / Department of Planning and Development
If the Matter is a contract being handled by the complete the following:	City's Department of Procurement Services, please
Specification # N/A	and Contract # N/A
	ge 1 of 15

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY	Ý
1. Indicate the nature of the Disclosing Pa Person Publicly registered business corporation Privately held business corporation Sole proprietorship General partnership Limited partnership Trust	Limited liability company Limited liability partnership Joint venture Not-for-profit corporation (Is the not-for-profit corporation also a 501(c)(3))? Yes No Other (please specify)
2. For legal entities, the state (or foreign cour	ntry) of incorporation or organization, if applicable:
Illinois	
3. For legal entities not organized in the State business in the State of Illinois as a foreign entity Yes	e of Illinois: Has the organization registered to do tity? Organized in Illinois
B. IF THE DISCLOSING PARTY IS A LEG	AL ENTITY:
the entity; (ii) for not-for-profit corporations are no such members, write "no members whice similar entities, the trustee, executor, administ limited partnerships, limited liability comparency general partner, managing member, managind indirectly controls the day-to-day management.	• •
NOTE: Each legal entity listed below must su	bmit an EDS on its own behalf.
Name Habitat 43 Green LLC	Title Managing Member
indirect, current or prospective (i.e. within 6 m	oncerning each person or legal entity having a direct or nonths after City action) beneficial interest (including . Examples of such an interest include shares in a

Ver.2018-1

corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

limited liability state "None."	company, or interest of a beneficiary	of a trust, estate or other similar entity. If none,
NOTE: Each le	egal entity listed below may be require	ed to submit an EDS on its own behalf.
Name	Business Address	Percentage Interest in the Applicant
NONE		
SECTION III OFFICIALS	INCOME OR COMPENSATION	TO, OR OWNERSHIP BY, CITY ELECTED
	sing Party provided any income or cond preceding the date of this EDS?	npensation to any City elected official during the Yes No
	osing Party reasonably expect to providuring the 12-month period following	de any income or compensation to any City the date of this EDS? Yes No
•	r of the above, please identify below to	he name(s) of such City elected official(s) and
inquiry, any Cit Chapter 2-156 of Yes If "yes," please	ty elected official's spouse or domestic of the Municipal Code of Chicago ("Mo No	sclosing Party's knowledge after reasonable partner, have a financial interest (as defined in ICC")) in the Disclosing Party? ty elected official(s) and/or spouse(s)/domestic

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(Add sheets if necessary)			
Check here if the Disc	closing Part	y has not retained, nor expects to re	tain, any such persons or entities.
SECTION V CERTII	FICATION	S ·	
A. COURT-ORDERED	CHILD SU	PPORT COMPLIANCE	
		antial owners of business entities the d support obligations throughout the	•
• •	•	ectly owns 10% or more of the Disc ations by any Illinois court of compo	•
Yes No 🗸	No person o	directly or indirectly owns 10% or n	nore of the Disclosing Party.
If "Yes," has the person e is the person in complian		a court-approved agreement for pay agreement?	ment of all support owed and
Yes No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] I Party nor any Affiliated I performance of any publi inspector general, or integinvestigative, or other sin activity of specified agen	In the 5-year Entity [see decontract, to grity complinitar skills, of cy vendors	the Matter is a contract being handler period preceding the date of this Edefinition in (5) below] has engaged the services of an integrity monitor, iance consultant (i.e., an individual designated by a public agency to he as well as help the vendors reform tests in the future, or continue with a cont	DS, neither the Disclosing, in connection with the independent private sector or entity with legal, auditing, lp the agency monitor the heir business practices so they

Ver 2018-1

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared incligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1

	If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further fications), the Disclosing Party must explain below:
	e letters "NA," the word "None," or no response appears on the lines above, it will be conclusively amed that the Disclosing Party certified to the above statements.
comp mont	to the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a plete list of all current employees of the Disclosing Party who were, at any time during the 12-th period preceding the date of this EDS, an employee, or elected or appointed official, of the City nicago (if none, indicate with "N/A" or "none").
the I office made the copoliti	o the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a olete list of all gifts that the Disclosing Party has given or caused to be given, at any time during 2-month period preceding the execution date of this EDS, to an employee, or elected or appointed ial, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything a generally available to City employees or to the general public, or (ii) food or drink provided in ourse of official City business and having a retail value of less than \$25 per recipient, or (iii) a cal contribution otherwise duly reported as required by law (if none, indicate with "N/A" or et"). As to any gift listed below, please also list the name of the City recipient.
C. C	ERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
1. T	he Disclosing Party certifies that the Disclosing Party (check one)
a	"financial institution" as defined in MCC Section 2-32-455(b).
2. It	the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
pledg	are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further that none of our affiliates is, and none of them will become, a predatory lender as defined in Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to

believe has not provided or cannot provide truthful certifications.

Ver.2018-1

predatory lender may result in the loss of the privilege of doing business with the City."

If the Disclosing Party is unable to make this pledge because it or any of its affiliates (as defined in MCC Section 2-32-455(b)) is a predatory lender within the meaning of MCC Chapter 2-32, explain here (attach additional pages if necessary):		
		ppears on the lines above, it will be
	ed that the Disclosing Party certifold	Ted to the above statements. NTEREST IN CITY BUSINESS
		ave the same meanings if used in this Part D.
In accordance with after reasonable inqu	h MCC Section 2-156-110: To tl	ne best of the Disclosing Party's knowledge e of the City have a financial interest in his or
Yes	No	
	xed "Yes" to Item D(1), proceed $ms D(2)$ and $D(3)$ and proceed to	to Items D(2) and D(3). If you checked "No" o Part E.
official or employee other person or entity taxes or assessments, "City Property Sale")	shall have a financial interest in I in the purchase of any property or (iii) is sold by virtue of legal	idding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, en pursuant to the City's eminent domain he meaning of this Part D.
Does the Matter invo	lve a City Property Sale?	
Yes	No	
		mes and business addresses of the City officials fy the nature of the financial interest:
Name	Business Address	Nature of Financial Interest

4. The Disclosing Party further certifies that no prohibited financial interest in the Matter will be acquired by any City official or employee.

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.
1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.
A. CERTIFICATION REGARDING LOBBYING
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)
2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined

by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Page **9** of **15**

Ver.2018-1

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the Applicant? Yes No
If "Yes," answer the three questions below:
1. Have you developed and do you have on file affirmative action programs pursuant to applicable federal regulations? (See 41 CFR Part 60-2.) Yes No
2. Have you filed with the Joint Reporting Committee, the Director of the Office of Federal Contra Compliance Programs, or the Equal Employment Opportunity Commission all reports due under the applicable filing requirements? Yes Reports not required
3. Have you participated in any previous contracts or subcontracts subject to the equal opportunity clause? Yes No
If you checked "No" to question (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

43 GREEN JV LLC

(See Opposite Signature Block)	By: Habitat 43 Green LLC, its Managing Member
(Print or type exact legal name of Disclosing Party)	By. Habitat Acquisitions Company LLC, its Manager
By:(Sign here)	By: The Habitat Company LLC, its Manager By
(Print or type name of person signing)	Its [.] President
(Print or type title of person signing)	
Signed and sworn to before me on (date)	
at GOOK County, TLLINOI'S (state) THE MANNETE CHALLS Notary Public	OFFICIAL SEAL LORI FRANCINE CHACOS NOTARY PUBLIC - STATE OF ILLINOIS MY COMMISSION EXPIRES: 10/21/24
Commission expires: $\frac{10}{21}$	

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?
Yes No
If yes, please identify below (1) the name and title of such person, (2) the name of the legal entity which such person is connected; (3) the name and title of the elected city official or department head twhom such person has a familial relationship, and (4) the precise nature of such familial relationship.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

 Pursuant to MCC Secti scofflaw or problem landle 		is the Applicant or any Owner identified as a building code MCC Section 2-92-416?
Yes	No	
	~ ,	cly traded on any exchange, is any officer or director of de scofflaw or problem landlord pursuant to MCC Section
Yes	No	The Applicant is not publicly traded on any exchange.
, , , ,	w or problem la	tify below the name of each person or legal entity identified andlord and the address of each building or buildings to which

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

Yes
No
$\sqrt{N/A}$ – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

HABITAT 43 GREEN LLC

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing Party submitt	ing this EDS. Include d/b/a/ if applicable:	
HABITAT 43 GREEN LLC		
Check ONE of the following three boxes:		
the contract, transaction or other undertaking to "Matter"), a direct or indirect interest in excess name: OR	nticipated to hold within six months after City action on which this EDS pertains (referred to below as the of 7.5% in the Applicant. State the Applicant's legal	
B. Business address of the Disclosing Party:	350 W. Hubbard St., Ste. 500	
	Chicago, IL 60654	
C. Telephone: 312 527 5400 Fax: 312 52	27 7440 Email: jhead1@habitat.com	
D. Name of contact person: Jeff Head		
E. Federal Employer Identification No. (if you	have one):	
F. Brief description of the Matter to which this property, if applicable):	s EDS pertains. (Include project number and location of	
Approach on by Approach for various forms of City financial assistance (approximately \$5,000,000 of 1 ax Maiti-Program Funds (proposity HOME), purchase of City property at reduced postal to facilitate development.	Theorement Financing (TTP), Tuy Credits (LIHTICs and IAHTCs), approximately \$3,000,000,000,000, and intermediately, of properly located at 43rd St. and S. Calumet Ave.)	
G. Which City agency or department is request	ting this EDS? Department of Housing / Department of Planning and Development	
If the Matter is a contract being handled by the complete the following:	City's Department of Procurement Services, please	
Specification # N/A	and Contract # N/A	
	ge 1 of 15	

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY 1. Indicate the nature of the Disclosing Party: Person Limited liability company Limited liability partnership Publicly registered business corporation Privately held business corporation Joint venture Sole proprietorship Not-for-profit corporation General partnership (Is the not-for-profit corporation also a 501(c)(3))? Limited partnership Yes No Trust Other (please specify) 2. For legal entities, the state (or foreign country) of incorporation or organization, if applicable: Illinois 3. For legal entities not organized in the State of Illinois: Has the organization registered to do business in the State of Illinois as a foreign entity? Organized in Illinois Yes B. IF THE DISCLOSING PARTY IS A LEGAL ENTITY: 1. List below the full names and titles, if applicable, of: (i) all executive officers and all directors of the entity; (ii) for not-for-profit corporations, all members, if any, which are legal entities (if there are no such members, write "no members which are legal entities"); (iii) for trusts, estates or other similar entities, the trustee, executor, administrator, or similarly situated party; (iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant. **NOTE**: Each legal entity listed below must submit an EDS on its own behalf. Name Title Habitat Acquisitions Company LLC Manager 2. Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including

ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a

corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

Ver 2018-1 Page 2 of 15

limited liability state "None."	company, or interest of a beneficiary	of a trust, estate or other similar entity. If none,
NOTE: Each le	gal entity listed below may be require	ed to submit an EDS on its own behalf.
Name	Business Address	Percentage Interest in the Applicant
NONE		
SECTION III - OFFICIALS	INCOME OR COMPENSATION	TO, OR OWNERSHIP BY, CITY ELECTED
	ing Party provided any income or cond preceding the date of this EDS?	npensation to any City elected official during the Yes No
	sing Party reasonably expect to providuring the 12-month period following	de any income or compensation to any City the date of this EDS? Yes No
•	of the above, please identify below t come or compensation:	he name(s) of such City elected official(s) and
inquiry, any Cit		isclosing Party's knowledge after reasonable partner, have a financial interest (as defined in ICC")) in the Disclosing Party?
	identify below the name(s) of such Ci escribe the financial interest(s).	ty elected official(s) and/or spouse(s)/domestic

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(Add sheets if necessary)			
Check here if the Disc	closing Part	y has not retained, nor expects to re	tain, any such persons or entities.
SECTION V CERTII	FICATION	S	
A. COURT-ORDERED	CHILD SU	PPORT COMPLIANCE	
		antial owners of business entities the support obligations throughout the	
* •	•	ectly owns 10% or more of the Disc ations by any Illinois court of comp	
Yes No	No person o	lirectly or indirectly owns 10% or n	nore of the Disclosing Party.
If "Yes," has the person e is the person in complian		a court-approved agreement for pay agreement?	ment of all support owed and
Yes No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] I Party nor any Affiliated I performance of any publi inspector general, or integ investigative, or other sin activity of specified agen	In the 5-year Entity <u>[see</u> die contract, t grity complinilar skills, o cy vendors	the Matter is a contract being handler period preceding the date of this Elefinition in (5) below has engaged the services of an integrity monitor, iance consultant (i.e., an individual designated by a public agency to he as well as help the vendors reform that in the future, or continue with a contin	DS, neither the Disclosing, in connection with the independent private sector or entity with legal, auditing, lp the agency monitor the heir business practices so they

Ver 2018-1

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Ver 2018-1 Page 5 of 15

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

	. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further extifications), the Disclosing Party must explain below:
	the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively esumed that the Disclosing Party certified to the above statements.
co.	To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a implete list of all current employees of the Disclosing Party who were, at any time during the 12-onth period preceding the date of this EDS, an employee, or elected or appointed official, of the City Chicago (if none, indicate with "N/A" or "none").
the off ma	To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a implete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed ficial, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything a de generally available to City employees or to the general public, or (ii) food or drink provided in a course of official City business and having a retail value of less than \$25 per recipient, or (iii) a litical contribution otherwise duly reported as required by law (if none, indicate with "N/A" or one"). As to any gift listed below, please also list the name of the City recipient.
<u> </u>	CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
1.	The Disclosing Party certifies that the Disclosing Party (check one) is is not
	a "financial institution" as defined in MCC Section 2-32-455(b).
2.	If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
ple	de are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further dge that none of our affiliates is, and none of them will become, a predatory lender as defined in CC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to

believe has not provided or cannot provide truthful certifications.

Ver 2018-1 Page 7 of 15

predatory lender may result in the loss of the privilege of doing business with the City."

If the Disclosing Party is unable to make this pledge because it or any of its affiliates (as defined in MCC Section 2-32-455(b)) is a predatory lender within the meaning of MCC-Chapter 2-32, explain here (attach additional pages if necessary):			
	the word "None," or no response a	appears on the lines above, it will be fied to the above statements.	
D. CERTIFICATI	ON REGARDING FINANCIAL I	NTEREST IN CITY BUSINESS	
Any words or term	s defined in MCC Chapter 2-156 h	ave the same meanings if used in this Part D.	
after reasonable in		he best of the Disclosing Party's knowledge e of the City have a financial interest in his or entity in the Matter?	
Yes	✓ No		
	ecked "Yes" to Item D(1), proceed Items D(2) and D(3) and proceed to	to Items D(2) and D(3). If you checked "No" o Part E.	
official or employed other person or ent taxes or assessmen "City Property Sale	te shall have a financial interest in ity in the purchase of any property ts, or (iii) is sold by virtue of legal	bidding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, ten pursuant to the City's eminent domain he meaning of this Part D.	
Does the Matter in	volve a City Property Sale?		
Yes	No		
		mes and business addresses of the City officials ify the nature of the financial interest:	
Name	Business Address	Nature of Financial Interest	

4. The Disclosing Party further certifies that no prohibited financial interest in the Matter will be acquired by any City official or employee.

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Ver.2018-1

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.				
1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.				
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:				
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS				
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.				
A. CERTIFICATION REGARDING LOBBYING				
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):				
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)				
2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee				

Page 9 of **15**

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the Applicant? Yes No
If "Yes," answer the three questions below:
1. Have you developed and do you have on file affirmative action programs pursuant to applicable federal regulations? (See 41 CFR Part 60-2.) Yes No
2. Have you filed with the Joint Reporting Committee, the Director of the Office of Federal Contract Compliance Programs, or the Equal Employment Opportunity Commission all reports due under the applicable filing requirements? Yes Reports not required
3. Have you participated in any previous contracts or subcontracts subject to the equal opportunity clause? No
If you checked "No" to question (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, 1L 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

HABITAT 43 GREEN LLC

(See Opposite Signature Block)	Ву:	Habitat	Acquisitions Company LLC, its Manager
(Print or type exact legal name of Disclosing Party) By:	·	Ву.	The Habitat Company LLC, its Manager By: M Up.
(Sign here)			Name: Matthew G. Fiascone Its: President
(Print or type name of person signing)			
(Print or type title of person signing)			
Signed and sworn to before me on (date) May	,	<u> 21</u> ,	
at GOK County ILLINOIS (state).			
Notary Public Commission expires: $\frac{10}{31/34}$	NOTARY	PUBLIC - S	L SEAL NE CHACOS STATE OF ILLINOIS EXPIRES 10/21/24

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?
☐ Yes ✓ No
If yes, please identify below (1) the name and title of such person, (2) the name of the legal entity to which such person is connected; (3) the name and title of the elected city official or department head to whom such person has a familial relationship, and (4) the precise nature of such familial relationship.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		10, is the Applicant or any Owner identified as a building code at to MCC Section 2-92-416?
Yes	No	
		ablicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
Yes	No	The Applicant is not publicly traded on any exchange.
	offlaw or proble	dentify below the name of each person or legal entity identified m landlord and the address of each building or buildings to which
		

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

Yes
No
$\sqrt{N/A}$ – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

HABITAT ACQUISITIONS COMPANY LLC

02021-3265

(5)

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing Party submitti	ing this EDS. Inc	lude d/b/a/ if applicable:
HABITAT ACQUISITIONS COMPANY LLC		
Check ONE of the following three boxes:		
Indicate whether the Disclosing Party submitting 1. the Applicant OR 2. a legal entity currently holding, or any the contract, transaction or other undertaking to "Matter"), a direct or indirect interest in excess name: OR 3. a legal entity with a direct or indirect State the legal name of the entity in which the I 43RD AND CALUMET PHASE ILP	nticipated to hold which this EDS of 7.5% in the Ap	of the Applicant (see Section II(B)(1))
B. Business address of the Disclosing Party:	350 WEST HUBBAR	RD STREET SUITE 500
	CHICAGO, IL 60654	
C. Telephone: (312) 527-5400 Fax: (312) 5	27-7440	Email: jhead1@habitat.com
D. Name of contact person: Jeff Head		_
E. Federal Employer Identification No. (if you	have one):	
F. Brief description of the Matter to which this property, if applicable):	EDS pertains. (1	Include project number and location of
Appikation by Apoteant for various forms of Chy financial assistance (approximately \$5,000 000 of Tax Must-Program Funds (probably HOME), purchase of City property at reduced price) to facilitate develope	Ingrement Financing ("TIF"). Tax Credi men, of property located at 43rd St. an	ts (DHTCs and IAFFCs) Epuroximatery \$3,000,000, City Leon of d S. Calumet Avv.)
G. Which City agency or department is request	ing this EDS? Dep	artment of Housing / Department of Planned Development
If the Matter is a contract being handled by the complete the following:	City's Departmer	nt of Procurement Services, please
Specification # N/A	and Contract # _	
	ge I of 15	

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY	
1. Indicate the nature of the Disclosing Party Person Publicly registered business corporation Privately held business corporation Sole proprietorship General partnership Limited partnership Trust	Limited liability company Limited liability partnership Joint venture Not-for-profit corporation The not-for-profit corporation also a 501(c)(3))? Yes No Other (please specify)
2. For legal entities, the state (or foreign country	y) of incorporation or organization, if applicable:
ILLINOIS	
3. For legal entities not organized in the State of business in the State of Illinois as a foreign entity	
Yes No	✓ Organized in Illinois
B. IF THE DISCLOSING PARTY IS A LEGAL	ENTITY:
the entity; (ii) for not-for-profit corporations, a are no such members, write "no members which similar entities, the trustee, executor, administra limited partnerships, limited liability compani	cable, of: (i) all executive officers and all directors of all members, if any, which are legal entities (if there are legal entities"); (iii) for trusts, estates or other tor, or similarly situated party; (iv) for general or es, limited liability partnerships or joint ventures, or or any other person or legal entity that directly or if the Applicant.
NOTE: Each legal entity listed below must subm	nit an EDS on its own behalf.
Name The Habitat Company LLC	Title Manager
indirect, current or prospective (i.e. within 6 mon ownership) in excess of 7.5% of the Applicant. If	cerning each person or legal entity having a direct or this after City action) beneficial interest (including Examples of such an interest include shares in a or joint venture, interest of a member or manager in a

limited liability state "None."	company, or interest of a beneficiary	of a trust, estate or other similar entity. If none,		
NOTE: Each le	NOTE: Each legal entity listed below may be required to submit an EDS on its own behalf.			
Name	Business Address	Percentage Interest in the Applicant		
NONE				
SECTION III - OFFICIALS	INCOME OR COMPENSATION	TO, OR OWNERSHIP BY, CITY ELECTED		
	ing Party provided any income or cond preceding the date of this EDS?	npensation to any City elected official during the Yes VNo		
	sing Party reasonably expect to providuring the 12-month period following	de any income or compensation to any City, the date of this EDS? Yes No		
•	of the above, please identify below to come or compensation:	he name(s) of such City elected official(s) and		
inquiry, any City	-	isclosing Party's knowledge after reasonable partner, have a financial interest (as defined in ICC")) in the Disclosing Party?		
•	identify below the name(s) of such Ci escribe the financial interest(s).	ity elected official(s) and/or spouse(s)/domestic		

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(Add sheets if necessary)			
		y has not retained, nor expects to re	tain, any such persons or entities.
SECTION V CERTII	ICATION	S	
A. COURT-ORDERED	CHILD SU	PPORT COMPLIANCE	
		antial owners of business entities the d support obligations throughout the	
· .		ectly owns 10% or more of the Disc ations by any Illinois court of compe	•
Yes No	No person (lirectly or indirectly owns 10% or m	nore of the Disclosing Party.
If "Yes," has the person e is the person in complian		a court-approved agreement for pay agreement?	ment of all support owed and
Yes No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] I Party nor any Affiliated I performance of any publi inspector general, or integ investigative, or other sin activity of specified agen	In the 5-year Entity [see do c contract, to grity complianilar skills, of cy vendors	the Matter is a contract being handler period preceding the date of this E lefinition in (5) below] has engaged, the services of an integrity monitor, iance consultant (i.e., an individual designated by a public agency to he as well as help the vendors reform that in the future, or continue with a contribute with a c	DS, neither the Disclosing, in connection with the independent private sector or entity with legal, auditing, lp the agency monitor the heir business practices so they

Ver 2018-1

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party:
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:		
N/A		
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.		
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with "N/A" or "none").		
None		
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a		
political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient.		
None		
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION		
1. The Disclosing Party certifies that the Disclosing Party (check one) is is not		
a "financial institution" as defined in MCC Section 2-32-455(b).		
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:		
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a		

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to

believe has not provided or cannot provide truthful certifications.

Ver 2018-1

predatory lender may result in the loss of the privilege of doing business with the City."

) is a predatory lender within th	nuse it or any of its affiliates (as defined in ne meaning of MCC Chapter 2-32, explain
	rd "None," or no response appe t the Disclosing Party certified	ears on the lines above, it will be to the above statements.
D. CERTIFICATION RE	GARDING FINANCIAL INTI	EREST IN CITY BUSINESS
Any words or terms define	ed in MCC Chapter 2-156 have	the same meanings if used in this Part D.
after reasonable inquiry, de		est of the Disclosing Party's knowledge the City have a financial interest in his or y in the Matter?
Yes	✓No	
	es" to Item D(1), proceed to It (2) and D(3) and proceed to Pa	tems D(2) and D(3). If you checked "No" art E.
official or employee shall other person or entity in the taxes or assessments, or (ii "City Property Sale"). Con	nave a financial interest in his or e purchase of any property that i) is sold by virtue of legal pro-	ing, or otherwise permitted, no City elected or her own name or in the name of any t (i) belongs to the City, or (ii) is sold for cess at the suit of the City (collectively, pursuant to the City's eminent domain neaning of this Part D.
Does the Matter involve a	City Property Sale?	
Yes	No	
		and business addresses of the City officials he nature of the financial interest:
Name	Business Address	Nature of Financial Interest

4. The Disclosing Party further certifies that no prohibited financial interest in the Matter will be acquired by any City official or employee.

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.			
1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.			
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:			
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS			
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.			
A. CERTIFICATION REGARDING LOBBYING			
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):			
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)			
2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined			

by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Page 9 of 15

Ver.2018-1

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the Applicant? Yes No	
If "Yes," answer the three questions below:	
Have you developed and do you have on file affirmative action programs pursuant to applicable federal regulations? (Sec 41 CFR Part 60-2.) Yes No	ole
2. Have you filed with the Joint Reporting Committee, the Director of the Office of Federal Con Compliance Programs, or the Equal Employment Opportunity Commission all reports due under applicable filing requirements? [Yes	
3. Have you participated in any previous contracts or subcontracts subject to the equal opportunity clause? Yes No	
If you checked "No" to question (1) or (2) above, please provide an explanation:	

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

(See Opposite Signature Block)	HABITAT ACQUISITIONS COMPANY LLC
(Print or type exact legal name of Disclosing Party)	
By:	By: The Habitat Company LLC, its Manage
(Sign here)	By:
(Print or type name of person signing)	
(Print or type title of person signing)	
Signed and sworn to before me on (date) Music	11,2021,
at Cook County, Illinois (state).	
Notary Public Commission expires: 10 / 21 / 21/	OFFICIAL SEAL LORI FRANCINE CHACOS NOTARY PUBLIC - STATE OF ILLINOIS

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?
Yes No
If yes, please identify below (1) the name and title of such person, (2) the name of the legal entity which such person is connected; (3) the name and title of the elected city official or department head twhom such person has a familial relationship, and (4) the precise nature of such familial relationship.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		10, is the Applicant or any Owner identified as a building code at to MCC Section 2-92-416?
Yes	No	
		ablicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
Yes	No	The Applicant is not publicly traded on any exchange.
	fflaw or probler	dentify below the name of each person or legal entity identified m landlord and the address of each building or buildings to which

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

Yes	
No ,	
$\sqrt{N/A}$ – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.	
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).	
If you checked "no" to the above, please explain.	
·	_

THE HABITAT COMPANY LLC

02021-3265

(6)

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing Party submitt	ing this EDS. Include d/b/a/ if applicable:
THE HABITAT COMPANY LLC	
Check ONE of the following three boxes:	
the contract, transaction or other undertaking to "Matter"), a direct or indirect interest in excess name: OR	nticipated to hold within six months after City action on which this EDS pertains (referred to below as the of 7.5% in the Applicant. State the Applicant's legal
B. Business address of the Disclosing Party:	350 W. HUBBARD ST STE 500
b. Business address of the Disclosing Party.	CHICAGO, IL 60654
C. Telephone: (312) 527-5400 Fax: (312) 5	Email: JHEAD1@HABITAT COM
D. Name of contact person: JEFF HEAD	
E. Federal Employer Identification No. (if you	have one):
F. Brief description of the Matter to which this property, if applicable):	s EDS pertains. (Include project number and location of
Appinal on by Applicant for various forms of Chy Francial assistence (approximately \$5,000,000 of Tax Mulli-Program Funds (procedby (IOME), purchase of Chy property at reduced price) to facilitate develop	k Increment Financing ["TIF"), Tax Credits (LIHTCs and IAHTCs) approximately \$3,990,000 City Loan of properly located at 43rd St, and S. Calumet Ave.)
G. Which City agency or department is reques	ting this EDS? Department of Housing / Department of Planning and Development
If the Matter is a contract being handled by the complete the following:	City's Department of Procurement Services, please
Specification # N/A	and Contract #
	ge 1 of 15

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY

1. Indicate the nature of the Disclosing Party: Person Limited liability company Publicly registered business corporation Limited liability partnership Privately held business corporation Joint venture Sole proprietorship Not-for-profit corporation (Is the not-for-profit corporation also a 501(c)(3))? General partnership Limited partnership Yes No Trust Other (please specify) 2. For legal entities, the state (or foreign country) of incorporation or organization, if applicable: ILLINOIS 3. For legal entities not organized in the State of Illinois: Has the organization registered to do business in the State of Illinois as a foreign entity? Organized in Illinois No Yes B. IF THE DISCLOSING PARTY IS A LEGAL ENTITY: 1. List below the full names and titles, if applicable, of: (i) all executive officers and all directors of the entity; (ii) for not-for-profit corporations, all members, if any, which are legal entities (if there are no such members, write "no members which are legal entities"); (iii) for trusts, estates or other similar entities, the trustee, executor, administrator, or similarly situated party; (iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant. NOTE: Each legal entity listed below must submit an EDS on its own behalf. Name Title SEE EXHIBIT "A" ATTACHED HERETO AND MADE A PART HEREOF 2. Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None." NOTE: Each legal entity listed below may be required to submit an EDS on its own behalf. Name **Business Address** Percentage Interest in the Applicant NONE SECTION III -- INCOME OR COMPENSATION TO, OR OWNERSHIP BY, CITY ELECTED **OFFICIALS** Has the Disclosing Party provided any income or compensation to any City elected official during the 12-month period preceding the date of this EDS? √ No Yes Does the Disclosing Party reasonably expect to provide any income or compensation to any City elected official during the 12-month period following the date of this EDS? If "yes" to either of the above, please identify below the name(s) of such City elected official(s) and describe such income or compensation: Does any City elected official or, to the best of the Disclosing Party's knowledge after reasonable inquiry, any City elected official's spouse or domestic partner, have a financial interest (as defined in Chapter 2-156 of the Municipal Code of Chicago ("MCC")) in the Disclosing Party? Yes If "yes," please identify below the name(s) of such City elected official(s) and/or spouse(s)/domestic partner(s) and describe the financial interest(s).

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist. etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(Add sheets if necessary)			
Check here if the Disc	closing Part	y has not retained, nor expects to re	tain, any such persons or entities.
SECTION V CERTII	FICATION	S	
A. COURT-ORDERED	CHILD SU	PPORT COMPLIANCE	
		antial owners of business entities the support obligations throughout the	
		ectly owns 10% or more of the Disc ations by any Illinois court of compe	<u> </u>
Yes No	No person c	lirectly or indirectly owns 10% or n	nore of the Disclosing Party.
If "Yes," has the person e is the person in complian-		a court-approved agreement for pay agreement?	ment of all support owed and
Yes No			7
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] I Party nor any Affiliated E performance of any publi inspector general, or integinvestigative, or other sin activity of specified agent	in the 5-year Entity <u> see</u> d c contract, t grity compli nilar skills, c cy vendors	the Matter is a contract being handler period preceding the date of this Exterinition in (5) below] has engaged the services of an integrity monitor, ance consultant (i.e., an individual designated by a public agency to he as well as help the vendors reform that in the future, or continue with a conti	DS, neither the Disclosing, in connection with the independent private sector or entity with legal, auditing, lp the agency monitor the heir business practices so they

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party:
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

	11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:		
N//	A		
	the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively esumed that the Disclosing Party certified to the above statements.		
co mo	. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a mplete list of all current employees of the Disclosing Party who were, at any time during the 12-onth period preceding the date of this EDS, an employee, or elected or appointed official, of the City Chicago (if none, indicate with "N/A" or "none").		
No	ne		
the of the po	To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a mplete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed ficial, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything ade generally available to City employees or to the general public, or (ii) food or drink provided in a course of official City business and having a retail value of less than \$25 per recipient, or (iii) a litical contribution otherwise duly reported as required by law (if none, indicate with "N/A" or one"). As to any gift listed below, please also list the name of the City recipient.		
C.	CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION		
1.	The Disclosing Party certifies that the Disclosing Party (check one) is is not		
	a "financial institution" as defined in MCC Section 2-32-455(b).		
2.	If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:		
	/e are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further edge that none of our affiliates is, and none of them will become, a predatory lender as defined in		

MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a

predatory lender may result in the loss of the privilege of doing business with the City."

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to

believe has not provided or cannot provide truthful certifications.

Ver.2018-1 Page 7 of 15

MCC Section 2-32-		necause it or any of its affiliates (as defined in the meaning of MCC Chapter 2-32, explain
	the word "None," or no response a ned that the Disclosing Party certif	ppears on the lines above, it will be lied to the above statements.
D. CERTIFICATIO	ON REGARDING FINANCIAL II	NTEREST IN CITY BUSINESS
Any words or terms	defined in MCC Chapter 2-156 h	ave the same meanings if used in this Part D.
after reasonable inq		ne best of the Disclosing Party's knowledge e of the City have a financial interest in his or ntity in the Matter?
Yes	No	
	cked "Yes" to Item $D(1)$, proceed tems $D(2)$ and $D(3)$ and proceed to	to Items D(2) and D(3). If you checked "No" Part E.
official or employed other person or enti- taxes or assessment "City Property Sale"	e shall have a financial interest in he ty in the purchase of any property s, or (iii) is sold by virtue of legal	idding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, en pursuant to the City's eminent domain he meaning of this Part D.
Does the Matter inv	olve a City Property Sale?	
Yes	No	
		nes and business addresses of the City officials fy the nature of the financial interest:
Name	Business Address	Nature of Financial Interest
		· · · · · · · · · · · · · · · · · · ·

4. The Disclosing Party further certifies that no prohibited financial interest in the Matter will be acquired by any City official or employee.

Ver.2018-1

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.
1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.
A. CERTIFICATION REGARDING LOBBYING
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)
2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined

by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee

Page 9 of 15

Ver.2018-1

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the Applicant?
Yes
If "Yes," answer the three questions below:
1. Have you developed and do you have on file affirmative action programs pursuant to applicable federal regulations? (See 41 CFR Part 60-2.) Yes No
2. Have you filed with the Joint Reporting Committee, the Director of the Office of Federal Contract Compliance Programs, or the Equal Employment Opportunity Commission all reports due under the applicable filing requirements? Yes Reports not required
3. Have you participated in any previous contracts or subcontracts subject to the equal opportunity clause? No
If you checked "No" to question (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

THE HABITAT COMPANY LLC
(Print or type exact legal name of Disclosing Party)
By: M. Ur-
(Sign here)
MATTHEW G FIASCONE
(Print or type name of person signing)
PRESIDENT
(Print or type title of person signing)
Signed and sworn to before me on (date)
OFFICIAL SEAL LORI FRANCINE CHACOS NOTARY PUBLIC - STATE OF ILLINOIS MY COMMISSION EXPIRES: 10/21/24

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT " APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

	ing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof nilial relationship" with an elected city official or department head?
Yes	No
which such person is	ntify below (1) the name and title of such person, (2) the name of the legal entity to connected; (3) the name and title of the elected city official or department head to as a familial relationship, and (4) the precise nature of such familial relationship.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

 Pursuant to MCC Secti scofflaw or problem landle 		o MCC Section 2-92-416?
Yes	No	
		licly traded on any exchange, is any officer or director of de scofflaw or problem landlord pursuant to MCC Section
Yes	No	The Applicant is not publicly traded on any exchange.
	w or problem l	ntify below the name of each person or legal entity identified landlord and the address of each building or buildings to which

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

Yes
No
N/A − I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

EXHIBIT "A"

City of Chicago Economic Disclosure and Affidavit The Habitat Company LLC SECTION II: DISCLOSURE OF OWNERSHIP INTERESTS

- B. 1(i) all executive officers and all directors of the entity
- B. I(iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant.

DANIEL E. LEVIN	CHAIRMAN / MANAGER
MATTHEW G. FIASCONE	PRESIDENT / MANAGER
STEPHEN F. GALLER	EXECUTIVE VICE PRESIDENT / GENERAL COUNSEL /
	MANAGER
SHEILA BYRNE	EXECUTIVE VICE PRESIDENT / PROPERTY
	MANAGEMENT / MANAGER
THOMAS BLACK	SENIOR VICE PRESIDENT, ARCHITECT &
	DIRECTOR OF PROJECT MANAGEMENT
MICHAEL CARSON	SENIOR VICE PRESIDENT / CHIEF INFORMATION
	OFFICER
CYNTHIA DIETZ	SENIOR VICE PRESIDENT / CORPORATE
	CONTROLLER
CHARLTON HAMER	SENIOR VICE PRESIDENT / PUBLIC HOUSING
	PORTFOLIO
LORI CHACOS	VICE PRESIDENT / ASSISTANT GENERAL COUNSEL.
SHYRLDEN CARMICHAEL	VICE PRESIDENT / BUSINESS OPERATIONS
JEFFREY D. HEAD	VICE PRESIDENT / COMMUNITY DEVELOPMENT
	GROUP
LORI FLASKA	VICE PRESIDENT / HUMAN RESOURCES
GINA FORTUNE-HARMON	VICE PRESIDENT / MARKET RATE
MAUREEN VAUGHN	VICE PRESIDENT / MARKETING &
	COMMUNICATIONS
HAGOP (JACK) DEVEDJIAN	VICE PRESIDENT / FACILITIES MANAGEMENT AND
	ENGINEERING
DAVID BARNHART	VICE PRESIDENT / CONDOMINIUM MANAGEMENT
KIM SISNEY	VICE PRESIDENT / HABITAT AFFORDABLE GROUP
ZACK ZALAR	VICE PRESIDENT / ACQUISITIONS & INVESTMENTS

HABITAT 43 GREEN LLC

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

•	
A. Legal name of the Disclosing Party submitt	ing this EDS. Include d/b/a/ if applicable:
HABITAT 43 GREEN LLC	
Check ONE of the following three boxes:	
the contract, transaction or other undertaking to "Matter"), a direct or indirect interest in excess name: 43RD AND CALUMET PHASE I LP OR	nticipated to hold within six months after City action on which this EDS pertains (referred to below as the of 7.5% in the Applicant. State the Applicant's legal
3. a legal entity with a direct or indirect State the legal name of the entity in which the I	t right of control of the Applicant (see Section II(B)(1)) Disclosing Party holds a right of control: ———————————————————————————————————
B. Business address of the Disclosing Party:	350 W. Hubbard St., Ste. 500
	Chicago, IL 60654
C. Telephone: 312 527 5400 Fax: 312 52	7 7440 Email: jhead1@habitat.com
D. Name of contact person: Jeff Head	
E. Federal Employer Identification No. (if you	have one):
F. Brief description of the Matter to which this property, if applicable):	EDS pertains. (Include project number and location of
Application by Applicant for various forms of City financial assistance (approximately \$5,000,000 of Tax Marti-Program Funds (crossibly (IGME)) purchase of City property at reduced price) to footblate develop	Increment Financing [TIFT], Tax Credits (LIHTCs and IAHTCs), approximately \$3,000,000 City Loan of men, of properly located at 4,7to St, and S. Calimati Ave.)
G. Which City agency or department is request	ing this EDS? Department of Housing / Department of Planning and Development
If the Matter is a contract being handled by the complete the following:	City's Department of Procurement Services, please
Specification # N/A	and Contract # N/A
Ver.2018-1 Pa	ge 1 of 15

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY

1. Indicate the nature of the Disclosing Par Person Publicly registered business corporation Privately held business corporation Sole proprietorship General partnership Limited partnership Trust	Limited liability company Limited liability partnership Joint venture Not-for-profit corporation (Is the not-for-profit corporation also a 501(c)(3))? Yes No Other (please specify)
2. For legal entities, the state (or foreign count	try) of incorporation or organization, if applicable:
3. For legal entities not organized in the State business in the State of Illinois as a foreign entities.	of Illinois: Has the organization registered to do ity?
Yes No	✓ Organized in Illinois
B. IF THE DISCLOSING PARTY IS A LEGA	AL ENTITY:
the entity; (ii) for not-for-profit corporations are no such members, write "no members whic similar entities, the trustee, executor, administ limited partnerships, limited liability compa	olicable, of: (i) all executive officers and all directors of all members, if any, which are legal entities (if there have legal entities"); (iii) for trusts, estates or other rator, or similarly situated party; (iv) for general or nies, limited liability partnerships or joint ventures, ger or any other person or legal entity that directly or of the Applicant.
NOTE: Each legal entity listed below must sub	omit an EDS on its own behalf.
Name Habitat Acquisitions Company LLC	Title Manager
indirect, current or prospective (i.e. within 6 me	oncerning each person or legal entity having a direct or onths after City action) beneficial interest (including Examples of such an interest include shares in a

Ver 2018-1

corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None."

NOTE: Each legal entity listed below may be required to submit an EDS on its own behalf.

Name Habitat Acquisitions Company LLC	Business Address 350 W. Hubbard St., Ste. 500, Chicago, IL 60654	Percentage Interest in the Applicant
Generations Housing Initiatives	350 W. Hubbard St., Ste. 500, Chicago, IL 60654	9.99%
SEE EXHIBIT "A" FOR OTHER I	PERSONS OR ENTITIES WITH INTEREST IN EXC	CESS OF 7.5% OF APPLICANT
SECTION III INCO OFFICIALS	OME OR COMPENSATION TO, C	OR OWNERSHIP BY, CITY ELECTED
	y provided any income or compensating the date of this EDS?	ion to any City elected official during the Yes No
_	ty reasonably expect to provide any ine 12-month period following the dat	
If "yes" to either of the a describe such income or		e(s) of such City elected official(s) and
inquiry, any City elected Chapter 2-156 of the Mi Yes If "yes," please identify	unicipal Code of Chicago ("MCC")) i ✓ No	r, have a financial interest (as defined in

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Ver.2018-1

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(Add sheets if necessary)			
Check here if the Disc	closing Part	y has not retained, nor expects to re	tain, any such persons or entities.
SECTION V CERTII	FICATION	S	
A. COURT-ORDERED	CHILD SU	PPORT COMPLIANCE	
		antial owners of business entities th d support obligations throughout the	<u> </u>
- 1	•	ectly owns 10% or more of the Disc ations by any Illinois court of compe	<u> </u>
Yes No	No person o	directly or indirectly owns 10% or m	nore of the Disclosing Party.
If "Yes," has the person e is the person in complian		a court-approved agreement for pay agreement?	ment of all support owed and
Yes No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] I Party nor any Affiliated E performance of any publi inspector general, or integinvestigative, or other sin	In the 5-year Entity (<u>see</u> dic contract, to grity complinity skills, on the compliant of the compliant of the compliant skills, on the compliant	the Matter is a contract being handler period preceding the date of this E lefinition in (5) below] has engaged, the services of an integrity monitor, iance consultant (i.e., an individual designated by a public agency to he as well as help the vendors reform t	DS, neither the Disclosing, in connection with the independent private sector or entity with legal, auditing, lp the agency monitor the

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

can be considered for agency contracts in the future, or continue with a contract in progress).

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees: or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity:
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Ver.2018-1

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1

11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below: N/A
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with "N/A" or "none").
complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient. N/A
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
1. The Disclosing Party certifies that the Disclosing Party (check one) is is not
a "financial institution" as defined in MCC Section 2-32-455(b).
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to

believe has not provided or cannot provide truthful certifications.

Ver.2018-1

predatory lender may result in the loss of the privilege of doing business with the City."

MCC Section 2-32-	455(b)) is a predatory lender with nal pages if necessary):	because it or any of its affiliates (as defined in in the meaning of MCC Chapter 2-32, explain
	the word "None," or no response a ned that the Disclosing Party certi	appears on the lines above, it will be fied to the above statements.
D. CERTIFICATION	ON REGARDING FINANCIAL I	NTEREST IN CITY BUSINESS
Any words or terms	s defined in MCC Chapter 2-156 h	ave the same meanings if used in this Part D.
after reasonable inc		he best of the Disclosing Party's knowledge e of the City have a financial interest in his or entity in the Matter?
Yes	No	
	cked "Yes" to Item D(1), proceed tems D(2) and D(3) and proceed to	to Items D(2) and D(3). If you checked "No" o Part E.
official or employed other person or enti- taxes or assessment "City Property Sale	e shall have a financial interest in ty in the purchase of any property s, or (iii) is sold by virtue of legal	bidding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, ten pursuant to the City's eminent domain he meaning of this Part D.
Does the Matter inv	volve a City Property Sale?	
Yes	No	
		mes and business addresses of the City officials ify the nature of the financial interest:
Name	Business Address	Nature of Financial Interest

4. The Disclosing Party further certifies that no prohibited financial interest in the Matter will be acquired by any City official or employee.

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Ver.2018-1

must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.
1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.
A. CERTIFICATION REGARDING LOBBYING
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)
2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph $A(1)$ above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee

Page 9 of 15

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the Applicant? Yes No
If "Yes," answer the three questions below:
1. Have you developed and do you have on file affirmative action programs pursuant to applicable federal regulations? (See 41 CFR Part 60-2.) Yes No
2. Have you filed with the Joint Reporting Committee, the Director of the Office of Federal Contract Compliance Programs, or the Equal Employment Opportunity Commission all reports due under the applicable filing requirements? Yes Reports not required
3. Have you participated in any previous contracts or subcontracts subject to the equal opportunity clause? Yes No
If you checked "No" to question (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

HABITAT 43 GREEN LLC

(See Opposite Signature Block)	Ву:	Habitat	t Acquisitions Company LLC, its Manager
(Print or type exact legal name of Disclosing Party) By:	,	Ву	The Habitat Company LLC, its Manager
(Sign here)			Name: Matthew G Fiascone Its. President
(Print or type name of person signing)	,		
(Print or type title of person signing)			
Signed and sworn to before me on (date) May	•	021	,
at COOK County, ILINO'S (state).			
Notary Public	LORI	OFFICIAL FRANCII	NE CHACOO
Commission expires: $\frac{10}{21/24}$	NOTARY MY CO	PUBLIC S MMISSION	EXPIRES.10/21/24

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?
☐Yes ✓No
If yes, please identify below (1) the name and title of such person, (2) the name of the legal entity to which such person is connected; (3) the name and title of the elected city official or department head to whom such person has a familial relationship, and (4) the precise nature of such familial relationship.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

1. Pursuant to MCC Section 2-154-010, is the Applicant scofflaw or problem landlord pursuant to MCC Section 2-	,
☐ Yes	
2. If the Applicant is a legal entity publicly traded on any the Applicant identified as a building code scofflaw or pro 2-92-416?	•
Yes No The Applica	nt is not publicly traded on any exchange.
3. If yes to (1) or (2) above, please identify below the na as a building code scofflaw or problem landlord and the a the pertinent code violations apply.	· · · · · · · · · · · · · · · · · · ·

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (<u>www.amlegal.com</u>), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

Yes
No
N/A − I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

EXHIBIT "A"

City of Chicago Economic Disclosure and Affidavit Habitat 43 Green LLC

SECTION II: DISCLOSURE OF OWNERSHIP INTERESTS

B. 2 Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None"

NAME	BUSINESS ADDRESS	PERCENTAGE INTEREST IN APPLICANT
HABITAT 43 GREEN LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	99.99%
HABITAT ACQUISITIONS COMPANY LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	89.99%
GENERATIONS HOUSING INITIATIVES	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	9.99%
THE HABITAT COMPANY LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	89.99%
Daniel E. Levin	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	
Matthew G. Fiascone	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	

HABITAT ACQUISITIONS COMPANY LLC

02021-3265

(8)

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION 1 -- GENERAL INFORMATION

A. Legal name of the Disclosing Party submitti	ng this EDS. Incl	ude d/b/a/ if applicable:	
HABITAT ACQUISITIONS COMPANY LLC			
Check ONE of the following three boxes:			
the contract, transaction or other undertaking to "Matter"), a direct or indirect interest in excess name: OR 3. a legal entity with a direct or indirect	ticipated to hold which this EDS pof 7.5% in the Ap	f the Applicant (see Section II(B)(1))	
State the legal name of the entity in which the D	Disclosing Party ho	olds a right of control:	
B. Business address of the Disclosing Party:	350 WEST HUBBARI	D STREET SUITE 500	
	CHICAGO, IL 60654		
C. Telephone: (312) 527-5400 Fax: (312) 52	27-7440	Email: jhead1@habitat.com	
D. Name of contact person: Jeff Head		-	
E. Federal Employer Identification No. (if you	have one)		
F. Brief description of the Matter to which this property, if applicable):	EDS pertains. (In	nclude project number and location of	
Approprian by Approach for various forms of Gry Francial assistance (approximately \$5,000,000 or This In Multi-Program Funds (prohably HOME), purchase of City property at reduced prical to facilitate developm	norsment Financing ("TIF"), Tax Credis ien, of property located at 43rd St. and	(LIHTCs and IAHTCs) zueroxima.oly \$3,900,000 Circlitian of Si Calumet Ave)	
G. Which City agency or department is requesti	ing this EDS? Depar	tment of Housing / Department of Planning and Development	
If the Matter is a contract being handled by the complete the following:	City's Department	t of Procurement Services, please	
Specification # N/A	and Contract # _		
	ge 1 of 15		

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY	,			
1. Indicate the nature of the Disclosing Par Person Publicly registered business corporation Privately held business corporation Sole proprietorship General partnership Limited partnership Trust	Limited liability company Limited liability partnership Joint venture Not-for-profit corporation (Is the not-for-profit corporation also a 501(c)(3))? Yes No Other (please specify)			
2. For legal entities, the state (or foreign coun	try) of incorporation or organization, if applicable:			
	of Illinois: Has the organization registered to do ity?			
Yes No	✓ Organized in Illinois			
B. IF THE DISCLOSING PARTY IS A LEGA	AL ENTITY:			
1. List below the full names and titles, if applicable, of: (i) all executive officers and all directors of the entity; (ii) for not-for-profit corporations, all members, if any, which are legal entities (if there are no such members, write "no members which are legal entities"); (iii) for trusts, estates or other similar entities, the trustee, executor, administrator, or similarly situated party; (iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant.				
NOTE: Each legal entity listed below must submit an EDS on its own behalf.				
Name The Habitat Company LLC	Title Manager			
Entity has no executive officers or directors				
indirect, current or prospective (i.e. within 6 mownership) in excess of 7.5% of the Applicant.	oncerning each person or legal entity having a direct or onths after City action) beneficial interest (including Examples of such an interest include shares in a p or joint venture, interest of a member or manager in a			

limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None." **NOTE**: Each legal entity listed below may be required to submit an EDS on its own behalf. **Business Address** Percentage Interest in the Applicant Name The Habitat Company LLC 350 W. Hubbard St., Ste., 500 Chicago, IL 60654 89 99% SEE EXHIBIT "A" FOR OTHER PERSONS OR ENTITIES WITH INTEREST IN EXCESS OF 7.5% OF APPLICANT SECTION III -- INCOME OR COMPENSATION TO, OR OWNERSHIP BY, CITY ELECTED **OFFICIALS** Has the Disclosing Party provided any income or compensation to any City elected official during the 12-month period preceding the date of this EDS? Yes Does the Disclosing Party reasonably expect to provide any income or compensation to any City elected official during the 12-month period following the date of this EDS? Yes If "yes" to either of the above, please identify below the name(s) of such City elected official(s) and describe such income or compensation: Does any City elected official or, to the best of the Disclosing Party's knowledge after reasonable inquiry, any City elected official's spouse or domestic partner, have a financial interest (as defined in Chapter 2-156 of the Municipal Code of Chicago ("MCC")) in the Disclosing Party? Yes **√** No If "yes," please identify below the name(s) of such City elected official(s) and/or spouse(s)/domestic partner(s) and describe the financial interest(s).

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(Add sheets if necessary)			
Check here if the Disc	closing Part	y has not retained, nor expects to re	tain, any such persons or entities.
SECTION V CERTII	FICATION	S	
A. COURT-ORDERED	CHILD SU	PPORT COMPLIANCE	
		antial owners of business entities the d support obligations throughout the	
<i>y</i> .	•	ectly owns 10% or more of the Disc ations by any Illinois court of compe	
Yes No	No person o	lirectly or indirectly owns 10% or n	nore of the Disclosing Party.
If "Yes," has the person e is the person in complian		a court-approved agreement for pay agreement?	ment of all support owed and
Yes No	,		
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] I Party nor any Affiliated E performance of any publi inspector general, or integ investigative, or other sin	In the 5-year Entity [see dec contract, to grity complinition of the contract, the compliant of the contract o	the Matter is a contract being handler period preceding the date of this Elefinition in (5) below] has engaged the services of an integrity monitor, iance consultant (i.e., an individual designated by a public agency to he as well as help the vendors reform to	DS, neither the Disclosing, in connection with the independent private sector or entity with legal, auditing, lp the agency monitor the

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

can be considered for agency contracts in the future, or continue with a contract in progress).

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties"):
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe. a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1

11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:
N/A
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with "N/ Λ " or "none").
None
the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient.
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
1. The Disclosing Party certifies that the Disclosing Party (check one) is is not
a "financial institution" as defined in MCC Section 2-32-455(b).
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to

believe has not provided or cannot provide truthful certifications.

Ver.2018-1 Page 7 of 15

predatory lender may result in the loss of the privilege of doing business with the City."

If the Disclosing Party is unable to make this pledge because it or any of its affiliates (as defined in MCC Section 2-32-455(b)) is a predatory lender within the meaning of MCC Chapter 2-32, explain here (attach additional pages if necessary): N/A
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.
D. CERTIFICATION REGARDING FINANCIAL INTEREST IN CITY BUSINESS
Any words or terms defined in MCC Chapter 2-156 have the same meanings if used in this Part D.
1. In accordance with MCC Section 2-156-110: To the best of the Disclosing Party's knowledge after reasonable inquiry, does any official or employee of the City have a financial interest in his or her own name or in the name of any other person or entity in the Matter?
☐ Yes
NOTE: If you checked "Yes" to Item D(1), proceed to Items D(2) and D(3). If you checked "No" to Item D(1), skip Items D(2) and D(3) and proceed to Part E.
2. Unless sold pursuant to a process of competitive bidding, or otherwise permitted, no City elected official or employee shall have a financial interest in his or her own name or in the name of any other person or entity in the purchase of any property that (i) belongs to the City, or (ii) is sold for taxes or assessments, or (iii) is sold by virtue of legal process at the suit of the City (collectively, "City Property Sale"). Compensation for property taken pursuant to the City's eminent domain power does not constitute a financial interest within the meaning of this Part D.
Does the Matter involve a City Property Sale?
Yes No
3. If you checked "Yes" to Item D(1), provide the names and business addresses of the City officials or employees having such financial interest and identify the nature of the financial interest:
Name Business Address Nature of Financial Interest

4. The Disclosing Party further certifies that no prohibited financial interest in the Matter will be acquired by any City official or employee.

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.			
1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.			
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:			
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS			
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.			
A. CERTIFICATION REGARDING LOBBYING			
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):			
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)			
2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Ver.2018-1			

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations. Is the Disclosing Party the Applicant? If "Yes," answer the three questions below: 1. Have you developed and do you have on file affirmative action programs pursuant to applicable federal regulations? (See 41 CFR Part 60-2.) Yes 2. Have you filed with the Joint Reporting Committee, the Director of the Office of Federal Contract Compliance Programs, or the Equal Employment Opportunity Commission all reports due under the applicable filing requirements? Yes Reports not required 3. Have you participated in any previous contracts or subcontracts subject to the equal opportunity clause? Yes No If you checked "No" to question (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

(See Opposite Signature Block)	HABITAT ACQUISITIONS COMPANY LLC
(Print or type exact legal name of Disclosing Party)	
By:	By: The Habitat Company LLC, its Manage
(Sign here)	By:
(Print or type name of person signing)	
(Print or type title of person signing)	
Signed and sworn to before me on (date) Music	11,2021,
at Cook County, Illinois (state).	
Notary Public Commission expires: 10/31/34	OFFICIAL SEAL LORI FRANCINE CHACOS NOTARY PUBLIC - STATE OF ILLINOIS MY COMMISSION EXPIRES: 10/21/24

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

•	Party or any "Applicable Party" or any Spouse or Domestic Partner thereof l relationship" with an elected city official or department head?
Yes	✓ No
which such person is cor	below (1) the name and title of such person, (2) the name of the legal entity to meeted; (3) the name and title of the elected city official or department head to familial relationship, and (4) the precise nature of such familial relationship.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		10, is the Applicant or any Owner identified as a building code to MCC Section 2-92-416?
Yes	✓ No	
• •	<u> </u>	blicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
Yes	No	The Applicant is not publicly traded on any exchange.
• • • • • •	offlaw or problen	entify below the name of each person or legal entity identified a landlord and the address of each building or buildings to which

Ver.2018-1 Page 14 of 15

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

Yes
No
$\sqrt{N/A}$ – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.
·

EXHIBIT "A"

City of Chicago Economic Disclosure and Affidavit Habitat Aquisitions Company LLC SECTION II: DISCLOSURE OF OWNERSHIP INTERESTS

B. 2 Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None"

NAME	BUSINESS ADDRESS	PERCENTAGE INTEREST IN APPLICANT
HABITAT 43 GREEN LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	99.99%
HABITAT ACQUISITIONS COMPANY LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	89.99%
GENERATIONS HOUSING INITIATIVES	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	9.99%
THE HABITAT COMPANY LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	89.99%
Daniel E. Levin	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	
Matthew G. Fiascone	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	

THE HABITAT COMPANY LLC

02021-3265

(9)

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION 1 - GENERAL INFORMATION

A. Legal name of the Disclosing Party submitting	ng this EDS. Include d/b/a/ if applicable:	
THE HABITAT COMPANY LLC		
Check ONE of the following three boxes:		
the contract, transaction or other undertaking to "Matter"), a direct or indirect interest in excess on name: 43RD AND CALUMET PHASE LP OR	ticipated to hold within six months after City action on which this EDS pertains (referred to below as the of 7.5% in the Applicant. State the Applicant's legal right of control of the Applicant (see Section II(B)(1))	
B. Business address of the Disclosing Party: 350 W. HUBBARD ST STE 500		
	CHICAGO, IL 60654	
C. Telephone: (312) 527-5400 Fax: (312) 52	7-7440 Email: JHEAD1@HABITAT.COM	
D. Name of contact person: JEFF HEAD		
E. Federal Employer Identification No. (if you	have one):	
F. Brief description of the Matter to which this property, if applicable):	EDS pertains. (Include project number and location of	
Appical on the Appicant for various forms of City financial assistance (approximately \$5,000,000 of Tax in Multi-Program Funds (probably rIOME), purchase of City property at reduced price) to facilitate developm	crement Financing [TTF1], Tax Credks (UHTCs and IASTOs) Epuroximately \$3,000,000 City Loan of ent of property located at 43rd St, and S. Calun at Ave.)	
G. Which City agency or department is requesti	ng this EDS? Department of Housing / Department of Planning and Development	
If the Matter is a contract being handled by the Complete the following:	City's Department of Procurement Services, please	
Specification # N/A	and Contract #	
	e 1 of 15	

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY 1. Indicate the nature of the Disclosing Party: Person Limited liability company Limited liability partnership Publicly registered business corporation Privately held business corporation Joint venture Sole proprietorship Not-for-profit corporation General partnership fts the not-for-profit corporation also a 501(c)(3))? Limited partnership Yes No Trust Other (please specify) 2. For legal entities, the state (or foreign country) of incorporation or organization, if applicable: ILLINOIS 3. For legal entities not organized in the State of Illinois: Has the organization registered to do business in the State of Illinois as a foreign entity? Organized in Illinois Yes B. IF THE DISCLOSING PARTY IS A LEGAL ENTITY: 1. List below the full names and titles, if applicable, of: (i) all executive officers and all directors of the entity; (ii) for not-for-profit corporations, all members, if any, which are legal entities (if there are no such members, write "no members which are legal entities"); (iii) for trusts, estates or other similar entities, the trustee, executor, administrator, or similarly situated party; (iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant. **NOTE**: Each legal entity listed below must submit an EDS on its own behalf. Name Title SEE EXHIBIT "A" ATTACHED HERETO AND MADE A PART HEREOF

2. Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

Name Daniel E. Levin	Business Address 350 W. Hubbard St., Ste. 500, Chicago, IL 60654	Percentage Interest in the Applicant
Matthew G Fiascone	350 W. Hubbard St., Ste. 500. Chicago, IL 60654	
SEE EXHIBIT "B" FOF	ROTHER PERSONS OR ENTITIES WITH INTEREST I	N EXCESS OF 7.5% OF APPLICANT
OFFICIALS		O, OR OWNERSHIP BY, CITY ELECTE Insation to any City elected official during the
	preceding the date of this EDS?	Yes Vo
	sing Party reasonably expect to provide a luring the 12-month period following the	·
=	of the above, please identify below the acome or compensation:	name(s) of such City elected official(s) and
inquiry, any City	•	osing Party's knowledge after reasonable rtner, have a financial interest (as defined in "")) in the Disclosing Party?
		elected official(s) and/or spouse(s)/domestic

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(A 11 al			
(Add sheets if necessary)			
Check here if the Disc	closing Part	y has not retained, nor expects to re	tain, any such persons or entities.
SECTION V CERTII	FICATION	S	
A. COURT-ORDERED	CHILD SU	PPORT COMPLIANCE	
	•	antial owners of business entities the support obligations throughout the	•
	•	ectly owns 10% or more of the Disc ations by any Illinois court of compe	_ ·
Yes No	No person o	directly or indirectly owns 10% or n	nore of the Disclosing Party.
If "Yes," has the person e is the person in complian		a court-approved agreement for pay agreement?	ment of all support owed and
Yes No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] I Party nor any Affiliated I	n the 5-yea Entity [<u>see</u> c	the Matter is a contract being handler period preceding the date of this Elefinition in (5) below] has engaged	DS, neither the Disclosing, in connection with the

- 1. [This paragraph 1 applies only if the Matter is a contract being handled by the City's Department of Procurement Services.] In the 5-year period preceding the date of this EDS, neither the Disclosing Party nor any Affiliated Entity [see definition in (5) below] has engaged, in connection with the performance of any public contract, the services of an integrity monitor, independent private sector inspector general, or integrity compliance consultant (i.e., an individual or entity with legal, auditing, investigative, or other similar skills, designated by a public agency to help the agency monitor the activity of specified agency vendors as well as help the vendors reform their business practices so they can be considered for agency contracts in the future, or continue with a contract in progress).
- 2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor-is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver 2018-1

11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:			
N//	N/A .		
	the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively esumed that the Disclosing Party certified to the above statements.		
co mo	12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with "N/A" or "none").		
No	ne .		
the of the po	mplete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed ficial, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything ade generally available to City employees or to the general public, or (ii) food or drink provided in a course of official City business and having a retail value of less than \$25 per recipient, or (iii) a litical contribution otherwise duly reported as required by law (if none, indicate with "N/A" or one"). As to any gift listed below, please also list the name of the City recipient.		
 C.	CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION		
١.	The Disclosing Party certifies that the Disclosing Party (check one) is is not		
	a "financial institution" as defined in MCC Section 2-32-455(b).		
2.	If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:		
	/e are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further edge that none of our affiliates is, and none of them will become, a predatory lender as defined in		

MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a

predatory lender may result in the loss of the privilege of doing business with the City."

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to

believe has not provided or cannot provide truthful certifications.

Ver.2018-1 Page 7 of 15

_ •) is a predatory lender withi	ecause it or any of its affiliates (as defined in the meaning of MCC Chapter 2-32, explain
	rd "None," or no response a t the Disclosing Party certif	ppears on the lines above, it will be ied to the above statements.
D. CERTIFICATION RE	GARDING FINANCIAL IN	NTEREST IN CITY BUSINESS
Any words or terms define	ed in MCC Chapter 2-156 ha	ave the same meanings if used in this Part D.
after reasonable inquiry, d		e best of the Disclosing Party's knowledge of the City have a financial interest in his or atity in the Matter?
Yes	No	
	Yes" to Item D(1), proceed to (2) and D(3) and proceed to	o Items D(2) and D(3). If you checked "No" Part E.
official or employee shall other person or entity in the taxes or assessments, or (if "City Property Sale"). Con	have a financial interest in he e purchase of any property i ii) is sold by virtue of legal p	dding, or otherwise permitted, no City elected is or her own name or in the name of any hat (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, en pursuant to the City's eminent domain e meaning of this Part D.
Does the Matter involve a	City Property Sale?	
Yes	No	
•	• •	nes and business addresses of the City officials by the nature of the financial interest:
Name	Business Address	Nature of Financial Interest

4. The Disclosing Party further certifies that no prohibited financial interest in the Matter will be acquired by any City official or employee.

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.			
1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.			
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:			
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS			
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.			
A. CERTIFICATION REGARDING LOBBYING			
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):			
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)			
2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined			

by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee

Page 9 of 15

Ver.2018-1

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the Applicant? Yes No
If "Yes," answer the three questions below:
1. Have you developed and do you have on file affirmative action programs pursuant to applicable federal regulations? (See 41 CFR Part 60-2.) Yes No
2. Have you filed with the Joint Reporting Committee, the Director of the Office of Federal Contract Compliance Programs, or the Equal Employment Opportunity Commission all reports due under the applicable filing requirements? Yes Reports not required
3. Have you participated in any previous contracts or subcontracts subject to the equal opportunity clause? Yes No
If you checked "No" to question (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

THE HABITAT COMPANY LLC	
(Print or type exact legal name of Disclosing Party)	•
Ву:	
(Sign here)	
MATTHEW G FIASCONE	
(Print or type name of person signing)	
PRESIDENT	
(Print or type title of person signing)	
Signed and sworn to before me on (date) August COOK County, ILLINOIS (state) White August County Public	11,2021,
Commission expires: $\frac{10/31/24}{}$	OFFICIAL SEAL LORI FRANCINE CHACOS IOTARY PUBLIC - STATE OF ILLINOIS

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?
☐ Yes
If yes, please identify below (1) the name and title of such person, (2) the name of the legal entity to which such person is connected; (3) the name and title of the elected city official or department head to whom such person has a familial relationship, and (4) the precise nature of such familial relationship.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT -APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		110, is the Applicant or any Owner identified as a building code at to MCC Section 2-92-416?
Yes	No	
	• • •	ablicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
Yes	No	The Applicant is not publicly traded on any exchange.
	flaw or probler	dentify below the name of each person or legal entity identified m landlord and the address of each building or buildings to which

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

Yes
No
$\sqrt{N/A}$ – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

EXHIBIT "A"

City of Chicago Economic Disclosure and Affidavit The Habitat Company LLC SECTION II: DISCLOSURE OF OWNERSHIP INTERESTS

- B. 1(i) all executive officers and all directors of the entity
- B. I(iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant.

DANIEL E. LEVIN	CHAIRMAN / MANAGER
MATTHEW G. FIASCONE	PRESIDENT / MANAGER
STEPHEN F. GALLER	EXECUTIVE VICE PRESIDENT / GENERAL COUNSEL /
	MANAGER
SHEILA BYRNE	EXECUTIVE VICE PRESIDENT / PROPERTY
	MANAGEMENT / MANAGER
THOMAS BLACK	SENIOR VICE PRESIDENT, ARCHITECT &
	DIRECTOR OF PROJECT MANAGEMENT
MICHAEL CARSON	SENIOR VICE PRESIDENT / CHIEF INFORMATION
	OFFICER
CYNTHIA DIETZ	SENIOR VICE PRESIDENT / CORPORATE
	CONTROLLER
CHARLTON HAMER	SENIOR VICE PRESIDENT / PUBLIC HOUSING
	PORTFOLIO
LORI CHACOS	VICE PRESIDENT / ASSISTANT GENERAL COUNSEL
SHYRLDEN CARMICHAEL	VICE PRESIDENT / BUSINESS OPERATIONS
JEFFREY D. HEAD	VICE PRESIDENT / COMMUNITY DEVELOPMENT
	GROUP
LORI FLASKA	VICE PRESIDENT / HUMAN RESOURCES
GINA FORTUNE-HARMON	VICE PRESIDENT / MARKET RATE
MAUREEN VAUGHN	VICE PRESIDENT / MARKETING &
	COMMUNICATIONS
HAGOP (JACK) DEVEDJIAN	VICE PRESIDENT / FACILITIES MANAGEMENT AND
	ENGINEERING
DAVID BARNHART	VICE PRESIDENT / CONDOMINIUM MANAGEMENT
KIM SISNEY	VICE PRESIDENT / HABITAT AFFORDABLE GROUP
ZACK ZALAR	VICE PRESIDENT / ACQUISITIONS & INVESTMENTS

EXHIBIT "B"

City of Chicago Economic Disclosure and Affidavit The Habitat Company LLC SECTION II: DISCLOSURE OF OWNERSHIP INTERESTS

B. 2 Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None"

NAME	BUSINESS ADDRESS	PERCENTAGE INTEREST IN APPLICANT
HABITAT 43 GREEN LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	99.99%
HABITAT ACQUISITIONS COMPANY LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	89.99%
GENERATIONS HOUSING INITIATIVES	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	9,99%
THE HABITAT COMPANY LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	89.99%
Daniel E. Levin	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	
Matthew G. Fiascone	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	

GENERATIONS HOUSING INITIATIVES

02021-3265

(10)

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION 1 -- GENERAL INFORMATION

•		
A. Legal name of the Disclosing Party submitting	ng this EDS. Inclu	.ide d/b/a/ if applicable:
GENERATIONS HOUSING INITIATIVES		
Check ONE of the following three boxes:		
Indicate whether the Disclosing Party submittin 1. the Applicant OR 2. a legal entity currently holding, or an the contract, transaction or other undertaking to "Matter"), a direct or indirect interest in excess on name: 43RD AND CALUMET PHASE ILP	ticipated to hold w	ertains (referred to below as the
OR 3. a legal entity with a direct or indirect State the legal name of the entity in which the D		
B. Business address of the Disclosing Party:	350 W. Hubbard St., S	ite. 500
	Chicago, IL 60654	
C. Telephone: (312) 527-5400 Fax: (312) 52	27-7440	Email: jhead1@habitat.com
D. Name of contact person: Jeff Head		
E. Federal Employer Identification No. (if you	have one):	
F. Brief description of the Matter to which this property, if applicable):	EDS pertains. (In	clude project number and location of
Application by Applicant for various forms of City Francial assistance (approximately \$5,000,000 of Tax In Mail-Program Funds (probably (IOME), purchase of City property at recurred most to facilitate davelopm	ncrement Financing ("TIF"), Tax Credits (nent of properly tocated at 43rd Stilland S	LIHTCs and IAHTCs), sporoximately \$3,000,000 City Lean of Calumet Ave.)
G. Which City agency or department is requesti	ing this EDS? Departs	ment of Housing / Department of Planning and Development
If the Matter is a contract being handled by the complete the following:	City's Department	of Procurement Services, please
Specification #	and Contract #	
Ver.2018-1 Pag	ge 1 of 15	·

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY

SEE ATTACHED EXHIBIT "A"	
Name	Title
NOTE: Each legal entity listed below must so	ubmit an EDS on its own behalf.
the entity; (ii) for not-for-profit corporation are no such members, write "no members wh similar entities, the trustee, executor, admini limited partnerships, limited liability comp	pplicable, of: (i) all executive officers and all directors of as, all members, if any, which are legal entities (if there ich are legal entities"); (iii) for trusts, estates or other strator, or similarly situated party; (iv) for general or canies, limited liability partnerships or joint ventures, tager or any other person or legal entity that directly or not of the Applicant.
B. IF THE DISCLOSING PARTY IS A LEG	GAL ENTITY:
3. For legal entities not organized in the State business in the State of Illinois as a foreign entities. Yes No	e of Illinois: Has the organization registered to dontity? Organized in Illinois
Illinois	
2. For legal entities, the state (or foreign cou	ntry) of incorporation or organization, if applicable:
Person Publicly registered business corporation Privately held business corporation Sole proprietorship General partnership Limited partnership Trust	Limited liability company Limited liability partnership Joint venture Not-for-profit corporation (Is the not-for-profit corporation also a 501(c)(3))? Yes No Other (please specify)

corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

limited liability state "None."	company, or interest of a beneficiary	of a trust, estate or other	similar entity. If no	one,
NOTE: Each le	gal entity listed below may be require	ed to submit an EDS on it	s own behalf.	
Name SEE EXHIBIT "B" FOR	Business Address R OTHER PERSONS OR ENTITIES WITH INTERES	Percentage Interest in the Applicant ST IN EXCESS OF 7.5% OF APPLICANT		nt
SECTION III - OFFICIALS	- INCOME OR COMPENSATION	TO, OR OWNERSHI	P BY, CITY ELEC	CTEE
	ing Party provided any income or cond preceding the date of this EDS?	npensation to any City el		g the No
	sing Party reasonably expect to providuring the 12-month period following			No
•	of the above, please identify below to	he name(s) of such City	elected official(s) ar	nd
inquiry, any City	lected official or, to the best of the Di y elected official's spouse or domestic of the Municipal Code of Chicago ("M	c partner, have a financia	l interest (as defined	d in
	identify below the name(s) of such Ciescribe the financial interest(s).	ity elected official(s) and	or spouse(s)/domes	stic

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
	-i		
(Add sheets if necessary)			
Check here if the Disc	closing Part	y has not retained, nor expects to re	tain, any such persons or entities.
SECTION V CERTIF	FICATION	S	
A. COURT-ORDERED	CHILD SU	PPORT COMPLIANCE	
	•	antial owners of business entities th d support obligations throughout the	•
· ·	•	ectly owns 10% or more of the Disc ations by any Illinois court of compe	•
Yes No	No person (directly or indirectly owns 10% or m	nore of the Disclosing Party.
If "Yes," has the person e is the person in compliance		a court-approved agreement for pay agreement?	ment of all support owed and
Yes No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] I Party nor any Affiliated E performance of any publi inspector general, or integ investigative, or other sin	n the 5-yea Entity <u>[see</u> d c contract, t grity compli tilar skills,	the Matter is a contract being handler period preceding the date of this E lefinition in (5) below] has engaged, the services of an integrity monitor, iance consultant (i.e., an individual designated by a public agency to he as well as help the vendors reform t	DS, neither the Disclosing, in connection with the independent private sector or entity with legal, auditing, lp the agency monitor the

can be considered for agency contracts in the future, or continue with a contract in progress).

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

Ver 2018-1 Page 4 of 15

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- c. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity:
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Ver.2018-1 Page 5 of 15

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1

11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Fu Certifications), the Disclosing Party must explain below:	rther
N/A	
If the letters "NA," the word "None," or no response appears on the lines above, it will be concluded that the Disclosing Party certified to the above statements.	usively
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the month period preceding the date of this EDS, an employee, or elected or appointed official, of to of Chicago (if none, indicate with "N/A" or "none").	
N/A	
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time d the 12-month period preceding the execution date of this EDS, to an employee, or elected or ap official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) any made generally available to City employees or to the general public, or (ii) food or drink provide the course of official City business and having a retail value of less than \$25 per recipient, or (in political contribution otherwise duly reported as required by law (if none, indicate with "N/A" of "none"). As to any gift listed below, please also list the name of the City recipient.	pointed /thing ed in i) a
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION	
1. The Disclosing Party certifies that the Disclosing Party (check one) is is not	
a "financial institution" as defined in MCC Section 2-32-455(b).	
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:	

"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a

predatory lender may result in the loss of the privilege of doing business with the City."

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to

believe has not provided or cannot provide truthful certifications.

Ver.2018-1 Page 7 of 15

MCC Section 2-32-4		because it or any of its affiliates (as defined in in the meaning of MCC Chapter 2-32, explain
	ne word "None," or no response a ed that the Disclosing Party certi	appears on the lines above, it will be fied to the above statements.
D. CERTIFICATIO	N REGARDING FINANCIAL I	NTEREST IN CITY BUSINESS
Any words or terms	defined in MCC Chapter 2-156 h	ave the same meanings if used in this Part D.
after reasonable inqu		the best of the Disclosing Party's knowledge to of the City have a financial interest in his or entity in the Matter?
Yes	✓No	
-	ked "Yes" to Item D(1), proceed ems D(2) and D(3) and proceed to	to Items D(2) and D(3). If you checked "No" o Part E.
official or employee other person or entity taxes or assessments "City Property Sale"	shall have a financial interest in a in the purchase of any property or (iii) is sold by virtue of legal	bidding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, sen pursuant to the City's eminent domain the meaning of this Part D.
Does the Matter invo	lve a City Property Sale?	
Yes	No	
		mes and business addresses of the City officials ify the nature of the financial interest:
Name	Business Address	Nature of Financial Interest

4. The Disclosing Party further certifies that no prohibited financial interest in the Matter will be acquired by any City official or employee.

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Ver.2018-1

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.
I. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.
A. CERTIFICATION REGARDING LOBBYING
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)
2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee

Page 9 of **15**

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the Applicant?
Yes No
If "Yes," answer the three questions below:
1. Have you developed and do you have on file affirmative action programs pursuant to applicable federal regulations? (See 41 CFR Part 60-2.) Yes No
2. Have you filed with the Joint Reporting Committee, the Director of the Office of Federal Contrac Compliance Programs, or the Equal Employment Opportunity Commission all reports due under the applicable filing requirements? Yes Reports not required
3. Have you participated in any previous contracts or subcontracts subject to the equal opportunity clause? Yes No
If you checked "No" to question (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

GENERATIONS HOUSING INITIATIVES
(Print on type exact legal name of Disclosing Party)
By: 109/1
(Sign Merch V
JEFFREY HEAD
(Print or type name of person signing)
PRESIDENT, TREASURER
(Print or type title of person signing)

Notary Public

Commission expires: $\frac{10/21/34}{}$

OFFICIAL SEAL
LORI FRANCINE CHACOS
NOTARY PUBLIC - STATE OF ILLINOIS
MY COMMISSION EXPIRES 10/21/24

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?

Yes

No

If yes, please identify below (1) the name and title of such person, (2) the name of the legal entity to which such person is connected; (3) the name and title of the elected city official or department head to whom such person has a familial relationship, and (4) the precise nature of such familial relationship.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		10, is the Applicant or any Owner identified as a building code to MCC Section 2-92-416?
Yes	✓ No	
		ablicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
Yes	No	The Applicant is not publicly traded on any exchange.
•	offlaw or problen	lentify below the name of each person or legal entity identified in landlord and the address of each building or buildings to which

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

Yes
No
$\sqrt{N/A} - I$ am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

EXHIBIT "A"

City of Chicago Economic Disclosure and Affidavit Generations Housing Initiatives NFP SECTION II: DISCLOSURE OF OWNERSHIP INTERESTS

B. (i) all executive officers and all directors of the entity

JEFFREY D. HEAD	PRESIDENT, TREASURER
SHYRLDEN CARMICHAEL	SECRETARY
AMY KASPER	VICE PRESIDENT (ABSENT)
LAWRENCE GRISHAM	SECRETARY
CRISTINA VERA-HUNT	EXECUTIVE DIRECTOR

B. (ii) for not-for-profit corporations, all members, if any, which are legal entities (if there are no such members, write "no members which are legal entities");

NO MEMBERS WHICH ARE LEGAL ENTITIES

EXHIBIT "B"

City of Chicago Economic Disclosure and Affidavit Generation Housing Initiatives SECTION II: DISCLOSURE OF OWNERSHIP INTERESTS

B. 2 Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None"

NAME	BUSINESS ADDRESS	PERCENTAGE INTEREST IN APPLICANT
HABITAT 43 GREEN LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	99.99%
HABITAT ACQUISITIONS COMPANY LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	89.99%
GENERATIONS HOUSING INITIATIVES	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	9.99%
THE HABITAT COMPANY LLC	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	89.99%
Daniel E. Levin	350 W. HUBBARD ST. STE 500, CHICAGO, IL 60654	
Matthew G. Fiascone	350 W. HUBBARD ST. STE 500, CHICAGO, II. 60654	

STRATFORD 43 GREEN INVESTORS LIMITED PARTNERSHIP

02021-3265

O2021-3238 – Amendment of Municipal Code Chapters 10-20, 10-21 and 10-32 regarding renaming of and procedures, exceptions and requirements for "811 Chicago" underground facilities damage prevention system (fka "DIGGER")

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing	; Party submitting this EDS	S. Include d/b	/a/ if applicable:
Stratford 43 Green Investors Limited Partnership			
Check ONE of the following th	ree boxes:		
Indicate whether the Disclosing I 1. [] the Applicant OR	Party submitting this EDS i	is:	
2. [x] a legal entity currently the contract, transaction or other "Matter"), a direct or indirect into name: 43rd and Calumet Phase I, I	undertaking to which this lerest in excess of 7.5% in the	EDS pertains	(referred to below as the State the Applicant's legal
OR [] a legal entity with a di State the legal name of the entity			oplicant (see Section II(B)(1)) ght of control:
B. Business address of the Discle	osing Party: 100 Cor	porate Place, S	Suite 404
	Peabody,	, MA 01960	- 11 11 an annia.
C. Telephone: (978) 535-5600 ext. 1	12 Fax:	Email:	bdm@stratfordcapitalgroup.com
D. Name of contact person:B	enjamin Motolla	 	
E. Federal Employer Identificati	on No. (if you have one):		**
F. Brief description of the Matte property, if applicable):	r to which this EDS pertain	ns. (Include p	roject number and location of
Application by Applicant for various for Financing ["TIF"]; Tax Credits (LITITO (probably HOME); purchase of City probably HOME);	Is and IAHTCs), approximatel	<u>ly \$3,000,000 Ci</u>	ty Loan of Multi-Program Funds
and S. Calumet Ave.);G. Which City agency or departn	nent is requesting this EDS	Department of P	lanning and Development ("DPD") & ousing ("DOH") If the
Matter is a contract being handled			
complete the following:			•
Specification #	and Contrac	ct #	. Now that the second s
Ver.2018-1	Page 1 of 15		

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY 1. Indicate the nature of the Disclosing Party: [] Limited liability company [] Person [] Limited liability partnership [] Publicly registered business corporation [] Privately held business corporation [] Joint venture [] Not-for-profit corporation [] Sole proprietorship [] General partnership (Is the not-for-profit corporation also a 501(c)(3))? [x] Limited partnership [] Yes []No [] Other (please specify) [] Trust 2. For legal entities, the state (or foreign country) of incorporation or organization, if applicable: Massachusetts 3. For legal entities not organized in the State of Illinois: Has the organization registered to do business in the State of Illinois as a foreign entity? [] Organized in Illinois []Yes [x] No B. IF THE DISCLOSING PARTY IS A LEGAL ENTITY: List below the full names and titles, if applicable, of: (i) all executive officers and all directors of the entity; (ii) for not-for-profit corporations, all members, if any, which are legal entities (if there are no such members, write "no members which are legal entities"); (iii) for trusts, estates or other similar entities, the trustee, executor, administrator, or similarly situated party; (iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant. **NOTE**: Each legal entity listed below must submit an EDS on its own behalf. Name Title SCG Capital Corp. General Partner 2. Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including

Ver.2018-1

ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

Name Stratford Interim Ho	Business Address ousing LLC 100 Corporate Place #404, Peabody, Ma	Percentage Interest in the App A 01960 99.97%	olicant
SECTION III - OFFICIALS	INCOME OR COMPENSATION TO,	OR OWNERSHIP BY, CITY E	LECTE
	ing Party provided any income or compensate of this EDS?	• •	uring the [x] No
	sing Party reasonably expect to provide any during the 12-month period following the d		City [x] No
•	r of the above, please identify below the national come or compensation:	me(s) of such City elected official	(s) and
inquiry, any Cit	elected official or, to the best of the Disclosicy elected official's spouse or domestic partion of the Municipal Code of Chicago ("MCC") [x] No	ner, have a financial interest (as de	
	identify below the name(s) of such City ele escribe the financial interest(s).	cted official(s) and/or spouse(s)/do	omestic

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Ver.2018-1

retained or anticipated to be retained)	Address	(subcontractor, attorney, lobbyist, etc.)	paid or estimated.) NOTE: "hourly rate" or "t.b.d." is	
Holland & Knight LLP	10 St. Jame	es Ave., 11th Floor, Boston MA 02116	not an acceptable response. \$100,000 (est.)	
, which believe amounts and summing summings and the second secon	okulların o Albanın .	. we say the the the second se	udani a seni senirentendani i i i	
(Add sheets if necessary		responsable and annual of a company of the second of the s		
[] Check here if the Dis	sclosing Party	y has not retained, nor expects to re	etain, any such persons or entities	
SECTION V CERTI	FICATION	S		
A. COURT-ORDERED	CHILD SUI	PPORT COMPLIANCE		
	•	antial owners of business entities the support obligations throughout the	•	
		ectly owns 10% or more of the Disc tions by any Illinois court of comp		
[] Yes [x] No []	No person d	irectly or indirectly owns 10% or n	nore of the Disclosing Party.	
If "Yes," has the person is the person in complian		a court-approved agreement for pay agreement?	ment of all support owed and	
[] Yes [] No				
B. FURTHER CERTIF	CATIONS			
Procurement Services.] Party nor any Affiliated performance of any publ	In the 5-year Entity [see do ic contract, the	he Matter is a contract being handle period preceding the date of this Experimental period preceding the date of this Experimental period preceding the services of an integrity monitor, ance consultant (i.e., an individual)	DS, neither the Disclosing, in connection with the independent private sector	

Relationship to Disclosing Party

Fees (indicate whether

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

investigative, or other similar skills, designated by a public agency to help the agency monitor the activity of specified agency vendors as well as help the vendors reform their business practices so they

can be considered for agency contracts in the future, or continue with a contract in progress).

Name (indicate whether Business

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to believe has not provided or cannot provide truthful certifications.				
11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:				
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.				
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with "N/A" or "none"). None				
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient. None				
C. OF DETICATION OF CELETIC AS FINANCIAL DISCUSSION				
 C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION 1. The Disclosing Party certifies that the Disclosing Party (check one) [] is [x] is not 				
a "financial institution" as defined in MCC Section 2-32-455(b).				
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:				
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a predatory lender may result in the loss of the privilege of doing business with the City."				

MCC Section 2-3		because it or any of its affiliates (as defined in in the meaning of MCC Chapter 2-32, explain
	" the word "None," or no response amed that the Disclosing Party certi	appears on the lines above, it will be fied to the above statements.
D. CERTIFICAT	ION REGARDING FINANCIAL I	NTEREST IN CITY BUSINESS
Any words or tern	ns defined in MCC Chapter 2-156 h	ave the same meanings if used in this Part D.
after reasonable in		he best of the Disclosing Party's knowledge te of the City have a financial interest in his or entity in the Matter?
[] Yes	[x] No	
	ecked "Yes" to Item D(1), proceed Items D(2) and D(3) and proceed to	to Items D(2) and D(3). If you checked "No" o Part E.
official or employed other person or entaxes or assessment "City Property Sal	ee shall have a financial interest in tity in the purchase of any property its, or (iii) is sold by virtue of legal	his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, ten pursuant to the City's eminent domain the meaning of this Part D.
Does the Matter in	volve a City Property Sale?	
[] Yes	[] No	
		mes and business addresses of the City officials fy the nature of the financial interest:
Name	Business Address	Nature of Financial Interest
		* ************************************
	g Party further certifies that no proh ty official or employee.	ibited financial interest in the Matter will be

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Ver.2018-1

Page 9 of 15

behalf of the Disclosing Party with respect to the Matter.)

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the	Applicant?
[] Yes	[X] No
If "Yes," answer the three	questions below:
Have you developed an federal regulations? (See 4	d do you have on file affirmative action programs pursuant to applicable II CFR Part 60-2.) [] No
Compliance Programs, or tapplicable filing requirement	Joint Reporting Committee, the Director of the Office of Federal Contrac he Equal Employment Opportunity Commission all reports due under the ents? [] No [] Reports not required
	n any previous contracts or subcontracts subject to the [] No
If you checked "No" to que	stion (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

Stratford 43 Green Investors Limited Partnership

(Print or type exactlegal name) of Disclosing Party)	
By:	
(Sign here)	
Benjamin D. Mottola	
(Print or type name of person signing)	
Benjamin D. Mottolla, President of SCG Capital Corp., General Partn (Print or type title of person signing)	er of the Disclosing Party
Signed and sworn to before mc on (date) 6212021,	
at ESSEX County, WA (state).	Medith B Medital
MR Mano S Nótary Public	***************************************
Commission expires: $5/5/2028$	MAY PURE TO STATE OF THE PROPERTY OF THE PROPE

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

. FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?

If yes, please identify below (1) the name and title of such person, (2) the name of the legal entity which such person is connected; (3) the name and title of the elected city official or department head twhom such person has a familial relationship, and (4) the precise nature of such familial relationship.	[] Yes	[x] No				
	which	such person is conn	ected; (3) the name	and title of the e	lected city officia	al or department	t head to
The second secon	·	er j er reden			<u></u>		
				* ************************************	represents traditional and recommendation of the second		
		080000 011 00000 0 X 1000 11 0 0 0 0 0 1					

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		10, is the Applicant or any Owner identified as a building code to MCC Section 2-92-416?
[] Yes	[x] No	
• •		ablicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
[] Yes	[] No	[x] The Applicant is not publicly traded on any exchange.
	offlaw or probler	entify below the name of each person or legal entity identified in landlord and the address of each building or buildings to which

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

[] Yes
[] No
[x] N/A – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

SCG CAPITAL CORP.

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Discle	sing Party submitting this EDS. Include d/b/a/ if applicable:
SCG Capital Corp.	
Check ONE of the following	g three boxes:
1. [] the Applicant OR 2. [] a legal entity curre the contract, transaction or o "Matter"), a direct or indirect name: OR 3. [x] a legal entity with State the legal name of the o	and Party submitting this EDS is: Only holding, or anticipated to hold within six months after City action on ther undertaking to which this EDS pertains (referred to below as the interest in excess of 7.5% in the Applicant. State the Applicant's legal a direct or indirect right of control of the Applicant (see Section II(B)(1)) antity in which the Disclosing Party holds a right of control: Limited Partnership; Stratford Capital Group, LLC
B. Business address of the l	
C. Telephone: (978) 535-56	00 x112 Fax: Email: bdm@stratfordcapitalgroup.com
D. Name of contact person:	Ben Mottola
E. Federal Employer Identi	cation No. (if you have one)
F. Brief description of the N property, if applicable):	latter to which this EDS pertains. (Include project number and location of
	ns of City financial <u>assistance</u> (approximately \$5,000,000 of Tax Increment Financing ["TIF"]; Tax Credits 6,000,000 City Loan of Multi-Program Funds (probably HQME); purchase of City property at reduced price) to - at 43 rd St and S. Calumet Aye.)
G. Which City agency or de	partment is requesting this EDS? <u>DOH and DPD</u>
If the Matter is a contract be complete the following:	ng handled by the City's Department of Procurement Services, please
Specification #	and Contract #
Ver.2018-1	Page 1 of 15

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY 1. Indicate the nature of the Disclosing Party: [] Limited liability company 1 1 Person [] Limited liability partnership [] Publicly registered business corporation [x] Privately held business corporation [] Joint venture [] Sole proprietorship [] Not-for-profit corporation [] General partnership (Is the not-for-profit corporation also a 501(c)(3))? [] Limited partnership [] Yes [] No [] Other (please specify) [] Trust 2. For legal entities, the state (or foreign country) of incorporation or organization, if applicable: Delaware 3. For legal entities not organized in the State of Illinois: Has the organization registered to do business in the State of Illinois as a foreign entity? []Yes [] Organized in Illinois [x] No B. IF THE DISCLOSING PARTY IS A LEGAL ENTITY: List below the full names and titles, if applicable, of: (i) all executive officers and all directors of the entity; (ii) for not-for-profit corporations, all members, if any, which are legal entities (if there are no such members, write "no members which are legal entities"); (iii) for trusts, estates or other similar entities, the trustee, executor, administrator, or similarly situated party; (iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant. **NOTE**: Each legal entity listed below must submit an EDS on its own behalf. Title Name Benjamin D. Mottola. President. Chief Operating Officer. Secretary and Treasurer

2. Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

John M. Nelson, IV. Chief Executive Officer

Kyle F. Wolff, Assistant Secretary and Director of Acquisitions

Stephen P. Wilson, President of Virginia Office

limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None."

NOTE: Each legal entity listed below may be required to submit an EDS on its own behalf.

Name	Business Address	F	ercentage	Interest in the A	Applicant
NJ Enterprises LP	100 Corporate Place #404, Peab	oody, MA 01960		24.995%	
Benjamin D. Mottola	100 Corporate Place #404, Peab	ody, MA 01960			
Kyle F. Wolff	100 Corporate Place #404, Peab	ody, MA 01960			
Stephen P. Wilson	100 Corporate Place #404, Peab	oody, MA 01960			
OFFICIALS	ME OR COMPENSATION	·	,	·	
	provided any income or c ing the date of this EDS?	compensation	to any City	elected officia	l during the [x] No
elected official during th	ty reasonably expect to prone 12-month period following the 12-month period following the pove, please identify below compensation:	ing the date of	this EDS?	Yes [] Yes	[x] No
inquiry, any City elected	ficial or, to the best of the official's spouse or domes nicipal Code of Chicago ("[x] No	tic partner, ha	ve a financ	ial interest (as	
If "yes," please identify by partner(s) and describe the	below the name(s) of such he financial interest(s).	City elected of	official(s) a	nd/or spouse(s))/domestic

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	rees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(Add sheets if necessary)			
[x] Check here if the Disc	closing Party	y has not retained, nor expects to ret	tain, any such persons or entities.
SECTION V CERTIF	TICATION	S	
A. COURT-ORDERED	CHILD SUI	PPORT COMPLIANCE	
		antial owners of business entities the support obligations throughout the	
	•	ectly owns 10% or more of the Disc tions by any Illinois court of compe	•
[] Yes [x] No [] N	No person di	rectly or indirectly owns 10% or mo	ore of the Disclosing Party.
If "Yes," has the person er is the person in compliance		court-approved agreement for paying agreement?	nent of all support owed and
[] Yes [] No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] In Party nor any Affiliated E performance of any public inspector general, or integ investigative, or other sim activity of specified agend	n the 5-year Entity [see decontract, the grity compliant is a contract, the grity compliant is a contract and a contract is a contract a contra	he Matter is a contract being handle period preceding the date of this ED efinition in (5) below has engaged, he services of an integrity monitor, ance consultant (i.e., an individual of designated by a public agency to hel as well as help the vendors reform the ts in the future, or continue with a contract	OS, neither the Disclosing in connection with the independent private sector or entity with legal, auditing, p the agency monitor the neir business practices so they

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

Ver.2018-1 Page 4 of 15

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use. any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1 Page 6 of 15

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to believe has not provided or cannot provide truthful certifications.
11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with "N/A" or "none"). None
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient. None
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
 The Disclosing Party certifies that the Disclosing Party (check one) [] is [x] is not
a "financial institution" as defined in MCC Section 2-32-455(b).
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a

predatory lender may result in the loss of the privilege of doing business with the City."

MCC Section 2-32-		because it or any of its affiliates (as defining the meaning of MCC Chapter 2-32.	
	the word "None," or no response ned that the Disclosing Party cert	appears on the lines above, it will be ified to the above statements.	<i>:</i>
D. CERTIFICATIO	ON REGARDING FINANCIAL	INTEREST IN CITY BUSINESS	
Any words or terms	s defined in MCC Chapter 2-156 l	have the same meanings if used in this Pa	art D.
after reasonable inq		he best of the Disclosing Party's knowled ee of the City have a financial interest in entity in the Matter?	
[] Yes	[x] No		
	eked "Yes" to Item D(1), proceed tems D(2) and D(3) and proceed t	to Items D(2) and D(3). If you checked to Part E.	"No"
official or employed other person or enti- taxes or assessment "City Property Sale	e shall have a financial interest in ty in the purchase of any property s, or (iii) is sold by virtue of legal	bidding, or otherwise permitted, no City of his or her own name or in the name of any that (i) belongs to the City, or (ii) is sole process at the suit of the City (collective ten pursuant to the City's eminent domain the meaning of this Part D.	ny d for ely,
Does the Matter inv	volve a City Property Sale?		
[] Yes	[] No		
	· · · •	ames and business addresses of the City of ify the nature of the financial interest:	officials
Name	Business Address	Nature of Financial Interest	
		m and a second s	
4. The Disclosing		nibited financial interest in the Matter wi	ll be

Ver.2018-1 Page **8** of **15**

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.

the Disclosing Party and from slavery or slavehol	Party verifies that the Disclosing Party has searched any and all records of any and all predecessor entities regarding records of investments or profit der insurance policies during the slavery era (including insurance policies at provided coverage for damage to or injury or death of their slaves), and found no such records.		
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:			
NOTE: If the Matter is federally funded, procee	IFICATIONS FOR FEDERALLY FUNDED MATTERS federally funded, complete this Section VI. If the Matter is not ed to Section VII. For purposes of this Section VI, tax credits allocated by debt obligations of the City are not federal funding.		
,	EGARDING LOBBYING		
List below the nar Disclosure Act of 1995, a	mes of all persons or entities registered under the federal Lobbying as amended, who have made lobbying contacts on behalf of the Disclosing Matter: (Add sheets if necessary):		
rarry with respect to the			

2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Ver.2018-1

Page 9 of 15

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs $\Lambda(1)$ through $\Lambda(4)$ above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

is the Disclosing Pa	y the Applicant?
[] Yes	[x] No
If "Yes," answer the	aree questions below:
	sed and do you have on file affirmative action programs pursuant to applicable See 41 CFR Part 60-2.) [] No
•	th the Joint Reporting Committee, the Director of the Office of Federal Contracts, or the Equal Employment Opportunity Commission all reports due under the irements?
[] Yes	[] No [] Reports not required
3. Have you partici equal opportunity cl	ated in any previous contracts or subcontracts subject to the use?
[] Yes	[] No
If you checked "No"	question (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City-s Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE**: With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

SCG Capital Corp., a Delaware corporation	
(Print or type exact legal name of Disclosing Party)	
By:	
(Sign here)	
Benjamin D. Mottola	
(Print or type name of person signing)	
President	
(Print or type title of person signing)	
Signed and sworn to before me on (date) $\frac{1}{15}$	1021.
at Millesex County, Massachusetts- (state).	ENIC R MELLIN
Notary Public	O NOTANIO
Commission expires: 1//1/2024	OF MASSE

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?

[] Yes	[x] No		•	
which such persor	dentify below (1) the nar n is connected; (3) the na n has a familial relationsh	ume and title of the ele	ected city official or	department head to
			W W W W W	
			AMMIN .	

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		10, is the Applicant or any Owner identified as a building code to MCC Section 2-92-416?
[]Yes	[x] No	
		ablicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
[] Yes	[] No	[x] The Applicant is not publicly traded on any exchange.
	offlaw or problen	dentify below the name of each person or legal entity identified in landlord and the address of each building or buildings to which

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (<u>www.amlegal.com</u>), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

[] Yes
[] No
[x] N/A – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

STRATFORD INTERIM HOUSING LLC

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing Party submi	itting this EDS. Include d/b/a/ if applicable:
Stratford Interim Housing LLC	
Check ONE of the following three boxes:	
the contract, transaction or other undertaking	anticipated to hold within six months after City action on to which this EDS pertains (referred to below as the ss of 7.5% in the Applicant. State the Applicant's legal
	ect right of control of the Applicant (see Section II(B)(1)) e Disclosing Party holds a right of control:
B. Business address of the Disclosing Party:	100 Corporate Place, Suite 404
	Peabo dy,MA 01960
C. Telephone: (978) 535-5600 ext. 112 Fax:	Email: bdm@stratfordcapitalgroup.com
D. Name of contact person: Ben Mo tola	
E. Federal Employer Identification No. (if yo	
F. Brief description of the Matter to which th property, if applicable):	is EDS pertains. (Include project number and location of
Tax Increment Financing ["TIF"], Tax Credits Loan of Multi-Program Funds (probably HOM	of City financial assistance (approximately \$5,000,000 of s (LIHTCs and IAHTCs), approximately \$3,000,000 City IE), purchase of City property at reduced price) to 43rd St. and S. Calumet Ave.).
G. Which City agency or department is required the Matter is a contract being handled by the complete the following:	esting this EDS? DPD and DOH ne City's Department of Procurement Services, please
Specification #	and Contract #
Ver.2018-1	Page 1 of 15

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY	^r
 Indicate the nature of the Disclosing Pa Person Publicly registered business corporation Privately held business corporation Sole proprietorship General partnership Limited partnership Trust 	[x] Limited liability company [] Limited liability partnership [] Joint venture [] Not-for-profit corporation (Is the not-for-profit corporation also a 501(c)(3))? [] Yes [] No [] Other (please specify)
2. For legal entities, the state (or foreign coun Delaware	atry) of incorporation or organization, if applicable:
3. For legal entities not organized in the State business in the State of Illinois as a foreign ent	of Illinois: Has the organization registered to do tity?
[] Yes [x] No	[] Organized in Illinois
B. IF THE DISCLOSING PARTY IS A LEGA	AL ENTITY:
the entity; (ii) for not-for-profit corporations are no such members, write "no members whice similar entities, the trustee, executor, administ limited partnerships, limited liability compa	plicable, of: (i) all executive officers and all directors of s, all members, if any, which are legal entities (if there ch are legal entities"); (iii) for trusts, estates or other trator, or similarly situated party; (iv) for general or mies, limited liability partnerships or joint ventures, ager or any other person or legal entity that directly or of the Applicant.
NOTE: Each legal entity listed below must sub	omit an EDS on its own behalf.
Name Stratford Capital Group, LLC	Title Manager
2. Please provide the following information coindirect, current or prospective (i.e. within 6 mo	oncerning each person or legal entity having a direct or on onths after City action) beneficial interest (including Examples of such an interest include shares in a

corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

limited liability company, state "None."	or interest of a beneficiary of	a trust, estate or otl	her similar enti	ity. If none,
NOTE: Each legal entity	isted below may be required t	o submit an EDS o	n its own beha	lf.
Name Stratford Capital Group, LLC	Business Address 100 Corporate Place #404, Peaboo		Interest in the 2 99.97%	Applicant
SECTION III INCOM OFFICIALS	E OR COMPENSATION T	O, OR OWNERS	нір ву, сіт	Y ELECTE
Has the Disclosing Party p 12-month period preceding	rovided any income or compeg the date of this EDS?	nsation to any City	elected officia	al during the [x] No
	reasonably expect to provide a			ny City [x] No
If "yes" to either of the abodescribe such income or co	ove, please identify below the ompensation:	name(s) of such Ci	ty elected offic	cial(s) and
inquiry, any City elected of	cial or, to the best of the Disclefficial's spouse or domestic pacipal Code of Chicago ("MCC [x] No	rtner, have a finan	cial interest (as	
	low the name(s) of such City e	elected official(s) a	nd/or spouse(s)/domestic

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(Add sheets if necessary)			
[x] Check here if the Disc	closing Party	y has not retained, nor expects to rel	ain, any such persons or entities.
SECTION V CERTIF	FICATION	s	
A. COURT-ORDERED	CHILD SUI	PPORT COMPLIANCE	
		antial owners of business entities the support obligations throughout the	•
- ·	•	ectly owns 10% or more of the Disc tions by any Illinois court of compe	
[] Yes [x] No []	No person d	irectly or indirectly owns 10% or m	ore of the Disclosing Party.
If "Yes," has the person e is the person in compliance		agreement?	ment of all support owed and
[] Yes [] No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] If Party nor any Affiliated Experiormance of any public inspector general, or integral.	n the 5-year Intity [<u>see</u> de c contract, th grity complia	he Matter is a contract being handled period preceding the date of this Election in (5) below has engaged, the services of an integrity monitor, in the consultant (i.e., an individual of the esignated by a public agency to helection.	OS, neither the Disclosing in connection with the independent private sector or entity with legal, auditing,

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

activity of specified agency vendors as well as help the vendors reform their business practices so they

can be considered for agency contracts in the future, or continue with a contract in progress).

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1 Page 6 of 15

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to believe has not provided or cannot provide truthful certifications.
11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with "N/A" or "none"). None
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient. None
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
 The Disclosing Party certifies that the Disclosing Party (check one) is [x] is not
a "financial institution" as defined in MCC Section 2-32-455(b).
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a predatory lender may result in the loss of the privilege of doing business with the City."

MCC Section 2-32-		because it or any of its affiliates (as defined in in the meaning of MCC Chapter 2-32, explain
	Allendaria de la companya del companya de la companya del companya de la companya del la companya de la company	The second secon
•	the word "None," or no response a ned that the Disclosing Party certif	ppears on the lines above, it will be fied to the above statements.
D. CERTIFICATION	ON REGARDING FINANCIAL I	NTEREST IN CITY BUSINESS
Any words or terms	s defined in MCC Chapter 2-156 h	ave the same meanings if used in this Part D.
after reasonable inq		ne best of the Disclosing Party's knowledge of the City have a financial interest in his or ntity in the Matter?
[] Yes	[x] No	·
	cked "Yes" to Item D(1), proceed tems D(2) and D(3) and proceed to	to Items D(2) and D(3). If you checked "No" Part E.
official or employed other person or enti taxes or assessment "City Property Sale	e shall have a financial interest in he ty in the purchase of any property s, or (iii) is sold by virtue of legal	idding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, en pursuant to the City's eminent domain e meaning of this Part D.
Does the Matter inv	olve a City Property Sale?	
[] Yes	[] No	
		nes and business addresses of the City officials by the nature of the financial interest:
Name	Business Address	Nature of Financial Interest
	-	
~	Party further certifies that no prohi y official or employee.	bited financial interest in the Matter will be

Page 8 of 15

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.			
			SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.			
A. CERTIFICATION REGARDING LOBBYING			
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):			
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)			
2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined			

by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Page 9 of 15

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs $\Lambda(1)$ through $\Lambda(4)$ above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the	Applicant?
[] Yes	[X] No
If "Yes," answer the three	questions below:
Have you developed ar federal regulations? (See 4) [] Yes	ad do you have on file affirmative action programs pursuant to applicable 41 CFR Part 60-2.) [] No
Compliance Programs, or applicable filing requirement	e Joint Reporting Committee, the Director of the Office of Federal Contract the Equal Employment Opportunity Commission all reports due under the ents? [] No [] Reports not required
	n any previous contracts or subcontracts subject to the
If you checked "No" to que	estion (1) or (2) above, please provide an explanation:

Page 10 of 15

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. NOTE: With respect to Matters subject to MCC Chapter 1-23, Article I (imposing PERMANENT INELIGIBILITY for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

Stratford Interim Housing LLC	
(Print or type exact legal name of Disclosing Party)	
By: (Sign here)	
Benjamin D. Mottola	
(Print or type name of person signing)	
President of SCG Capital Corp., Manager of Stratford Capital Group, (Print or type title of person signing)	LLC, Manager of Disclosing Party
Signed and sworn to before me on (date) 6/16/2021,	
at <u>F55FX</u> County, <u>W</u> (state).	William B Marin
MR Mayo 5 Notary Public	***
Commission expires: 5/5/2028	SACHUSEIL

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?

Yes	[x] NO	
which such person	is connected; (3) the nam	e and title of such person, (2) the name of the legal entity to and title of the elected city official or department head to p, and (4) the precise nature of such familial relationship.
	## V ##	
	delikum /antu takihabarah . Vandinggan	
		1

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

 Pursuant to MCC Section 2-154-010, is the Applicant or any Owner identified as a building code scofflaw or problem landlord pursuant to MCC Section 2-92-416? 		
[] Yes	[x] No	
• •	~ .	ablicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
[] Yes	[] No	[x] The Applicant is not publicly traded on any exchange.
	offlaw or problen	lentify below the name of each person or legal entity identified in landlord and the address of each building or buildings to which

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

[] Yes
[] No
[x] N/A – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

STRATFORD CAPITAL GROUP LLC

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclos	ing Party submitting this El	DS. Include d/b/a/ if applicable:
Stratford Capital Group LLC		
Check ONE of the following	three boxes:	
Indicate whether the Disclosin 1. [] the Applicant OR	ng Party submitting this ED	OS is:
2. [x] a legal entity currer the contract, transaction or oth "Matter"), a direct or indirect name: 43rd and Calumet Phase	ner undertaking to which the interest in excess of 7.5% is	to hold within six months after City action on his EDS pertains (referred to below as the in the Applicant. State the Applicant's legal
OR 3. [] a legal entity with a State the legal name of the ent	direct or indirect right of c	control of the Applicant (see Section II(B)(1)) g Party holds a right of control:
B. Business address of the Di	sclosing Party: 100 C	Corporate Place, Suite 404
	Peabo	ody, MA 01960
C. Telephone: (978) 535-5600 es	d. 112 Fax:	Email: bdm@stratfordcapitalgroup.com
D. Name of contact person:	Ben Mottola	
E. Federal Employer Identific	cation No. (if you have one)	e):
property, if applicable): Application by Applicant for various Financing ["TIF"], Tax Credits (LIH")	s forms of City financial assistan TCs and IAHTCs); approximatel	rtains. (Include project number and location of nce (approximately \$5,000,000 of Tax Increment say \$3,000,000 City Loan of Multi-Program Funds facilitate development of property located at 437d St.
and S. Calumet Ave.).	property acreadoca price, to te	domate development of property located at 15th of.
G. Which City agency or depart of the Matter is a contract bein complete the following:	artment is requesting this Eg handled by the City's De	EDS? DPD and DOH epartment of Procurement Services, please
Specification #	and Con	ntract #
Ver.2018-1		

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY 1. Indicate the nature of the Disclosing Party: [] Person [x] Limited liability company [] Publicly registered business corporation 1 Limited liability partnership [] Privately held business corporation [] Joint venture [] Sole proprietorship [] Not-for-profit corporation [] General partnership (Is the not-for-profit corporation also a 501(c)(3))? [] Limited partnership [] Yes [] No [] Trust Other (please specify) 2. For legal entities, the state (or foreign country) of incorporation or organization, if applicable: Delaware 3. For legal entities not organized in the State of Illinois: Has the organization registered to do business in the State of Illinois as a foreign entity? [] Yes [x] No [] Organized in Illinois B. IF THE DISCLOSING PARTY IS A LEGAL ENTITY: List below the full names and titles, if applicable, of: (i) all executive officers and all directors of the entity; (ii) for not-for-profit corporations, all members, if any, which are legal entities (if there are no such members, write "no members which are legal entities"); (iii) for trusts, estates or other similar entities, the trustee, executor, administrator, or similarly situated party; (iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant. **NOTE**: Each legal entity listed below must submit an EDS on its own behalf. Name Title SCG Capital Corp. Manager 2. Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a

corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None."

NOTE: Each legal entity listed below may be required to submit an EDS on its own behalf.

Name NJ Enterprises LP	Business Address 100 Corporate Place #404,		ge Interest in the A	applicant
Milo Investments LLC	100 Corporate Place #404,	Peabody, MA 01960	24.99%)
Wilson Ventures II LLC	100 Corporate Place #404, 1	Peabody, MA 01960	24.99%)
Doublewide LLC	100 Corporate Place #404,	Peabody, MA 01960	24.99%	,
SECTION III INCO OFFICIALS	ME OR COMPENSATI	ON TO, OR OWNER	RSHIP ВУ, СІТУ	ELECTED
	provided any income or one the provided any income or one of the end of the e	compensation to any C	ity elected officia	l during the [x] No
-	y reasonably expect to pro e 12-month period follow	-	-	y City [x] No
If "yes" to either of the a describe such income or	bove, please identify belo compensation:	w the name(s) of such	City elected offic	ial(s) and
inquiry, any City elected	ficial or, to the best of the official's spouse or dome nicipal Code of Chicago ([x] No	stic partner, have a fin	ancial interest (as	
If "yes," please identify to partner(s) and describe the	oclow the name(s) of such ne financial interest(s).	City elected official(s) and/or spouse(s)	/domestic
	 			

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(Add sheets if necessary)		•	
SECTION V CERTII		y has not retained, nor expects to re	tain, any such persons or entities.
A. COURT-ORDERED	·		
Under MCC Section 2-92	2-415, subst	antial owners of business entities the support obligations throughout the	
		ectly owns 10% or more of the Disc ations by any Illinois court of compe	
[] Yes [x] No [] :	No person d	lirectly or indirectly owns 10% or m	nore of the Disclosing Party.
If "Yes," has the person e is the person in compliance		a court-approved agreement for pay agreement?	ment of all support owed and
[]Yes []No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] I Party nor any Affiliated E performance of any publi inspector general, or integ investigative, or other sim	n the 5-year Entity [see d c contract, t grity compli tilar skills, o	the Matter is a contract being handle period preceding the date of this E efinition in (5) below] has engaged, he services of an integrity monitor, ance consultant (i.c., an individual of designated by a public agency to help as well as help the vendors reform the	DS, neither the Disclosing in connection with the independent private sector or entity with legal, auditing, up the agency monitor the

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

can be considered for agency contracts in the future, or continue with a contract in progress).

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1

11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:
<u> </u>
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with "N/A" or "none"). None
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient. None
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
 The Disclosing Party certifies that the Disclosing Party (check one) is [x] is not
a "financial institution" as defined in MCC Section 2-32-455(b).
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a predatory lender may result in the loss of the privilege of doing business with the City."

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to believe has not provided or cannot provide truthful certifications.

MCC Section 2-32		occause it or any of its affiliates (as defined in in the meaning of MCC Chapter 2-32, explain
	" the word "None," or no response a amed that the Disclosing Party certi	appears on the lines above, it will be fied to the above statements.
D. CERTIFICAT	ION REGARDING FINANCIAL I	NTEREST IN CITY BUSINESS
Any words or term	ns defined in MCC Chapter 2-156 h	ave the same meanings if used in this Part D.
after reasonable in		ne best of the Disclosing Party's knowledge e of the City have a financial interest in his or ntity in the Matter?
[] Yes	[x] No	
•	ecked "Yes" to Item D(1), proceed Items D(2) and D(3) and proceed to	to Items D(2) and D(3). If you checked "No" o Part E.
official or employed other person or ent taxes or assessment "City Property Sal	ee shall have a financial interest in latity in the purchase of any property ats, or (iii) is sold by virtue of legal	idding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, en pursuant to the City's eminent domain he meaning of this Part D.
Does the Matter in	volve a City Property Sale?	
[] Yes	[] No	
		mes and business addresses of the City officials fy the nature of the financial interest:
Name	Business Address	Nature of Financial Interest
4 79 D. L.		
-	g Party further certifies that no prohity official or employee.	ibited financial interest in the Matter will be

Ver.2018-1

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated the City and proceeds of debt obligations of the City are not federal funding. A. CERTIFICATION REGARDING LOBBYING 1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosin Party with respect to the Matter: (Add sheets if necessary):
federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated to the City and proceeds of debt obligations of the City are not federal funding. A. CERTIFICATION REGARDING LOBBYING
federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated to the City and proceeds of debt obligations of the City are not federal funding.
federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by
Nome to be a local to the second of the seco
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS
policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, to Disclosing Party has found records of investments or profits from slavery or slaveholder insurance.
\underline{x} 1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or prof from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.
Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.

(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)

2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Ver.2018-1

Page 9 of 15

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs $\Lambda(1)$ and $\Lambda(2)$ above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

	he Disclosing Party the	
	[] Yes	[x] No
If "	Yes," answer the three of	lestions below:
fed	Have you developed an eral regulations? (See 4	do you have on file affirmative action programs pursuant to applicable CFR Part 60-2.)] No
Coi app	mpliance Programs, or tolicable filing requireme	Joint Reporting Committee, the Director of the Office of Federal Contract e Equal Employment Opportunity Commission all reports due under the ts? [] No [] Reports not required
	į j 103	[] No [] Reports not required
	Have you participated in all opportunity clause?	any previous contracts or subcontracts subject to the
	[] Yes] No
lf y	ou checked "No" to que	tion (1) or (2) above, please provide an explanation:
	~	

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that hc/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

Stratford Capital Group, LLC
(Print or type exact legal name of Disclosing Party)
By:
(Sign here)
Benjamin D. Mottola
(Print or type name of person signing)
President of SCG Capital Corp., Manager of the Disclosing Party
(Print or type title of person signing)
Signed and sworn to before me on (date) 6 16 2021,
at ESSEX County, MA (state).
Notary Public
Commission expires: 5 5 2028

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?

[] Yes	[x] No	
which such person	is connected; (3) the na	me and title of such person, (2) the name of the legal entity to me and title of the elected city official or department head to hip, and (4) the precise nature of such familial relationship.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		t to MCC Section 2-92-416?
[] Yes	[x] No	
• •	~ .	ablicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
[] Yes	[] No	[x] The Applicant is not publicly traded on any exchange.
	cofflaw or problen	entify below the name of each person or legal entity identified n landlord and the address of each building or buildings to which

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (<u>www.amlegal.com</u>), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

^
If you checked "no" to the above, please explain.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
[x] N/A – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
[] No
[] Yes

MILO INVESTMENTS LLC

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing Party submit	tting this EDS. Include d/b/a/ if applicable:
Milo Investments LLC	
Check ONE of the following three boxes:	
the contract, transaction or other undertaking to "Matter"), a direct or indirect interest in excess name: 43rd and Calumet Phase I I OR	anticipated to hold within six months after City action on to which this EDS pertains (referred to below as the s of 7.5% in the Applicant. State the Applicant's legal LP
B. Business address of the Disclosing Party:	46 Pilgrim Road
	Marblehead, MA 01945
C. Telephone: (978) 535-5600 Fax:	Email: bdm@stratfordcapitalgroup.com
D. Name of contact person: Ben Mottola	
E. Federal Employer Identification No. (if yo	
property, if applicable): Application by Applicant for various forms of City financial assista (LIHTCs and IAHTCs): approximately \$3,000,000 City Loan of	s EDS pertains. (Include project number and location of nee (approximately \$5.000,000 of Tax Increment Financing ["TIF"]; Tax Credits Multi-Program Funds (probably HOME), purchase of City property at reduced openty located at 43rd St. and S. Calumet Ave.).
G. Which City agency or department is reques	sting this EDS? DPD and DOH
If the Matter is a contract being handled by the complete the following:	e City's Department of Procurement Services, please
Specification #	and Contract #
Ver.2018-1 P	age 1 of 15

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY 1. Indicate the nature of the Disclosing Party: [] Person [x] Limited liability company [] Publicly registered business corporation [] Limited liability partnership [] Joint venture Privately held business corporation [] Sole proprietorship [] Not-for-profit corporation [] General partnership (Is the not-for-profit corporation also a 501(c)(3))? [] Limited partnership []Yes [] No [] Other (please specify) [] Trust 2. For legal entities, the state (or foreign country) of incorporation or organization, if applicable: Massachusetts 3. For legal entities not organized in the State of Illinois: Has the organization registered to do business in the State of Illinois as a foreign entity? [] Yes [] Organized in Illinois [x] No B. IF THE DISCLOSING PARTY IS A LEGAL ENTITY: List below the full names and titles, if applicable, of: (i) all executive officers and all directors of the entity; (ii) for not-for-profit corporations, all members, if any, which are legal entities (if there are no such members, write "no members which are legal entities"); (iii) for trusts, estates or other similar entities, the trustee, executor, administrator, or similarly situated party; (iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant. NOTE: Each legal entity listed below must submit an EDS on its own behalf. Name Title Benjamin D. Mottola Manager 2. Please provide the following information concerning each person or legal entity having a direct or

indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

Ver.2018-1 Page 2 of 15

state "None." NOTE: Each legal entity listed below may be required to submit an EDS on its own behalf. Business Address Name Percentage Interest in the Applicant Benjamin D. Mottola Living Trust 46 Pilgrim Road, Marblehead, MA 01945 SECTION III -- INCOME OR COMPENSATION TO, OR OWNERSHIP BY, CITY ELECTED **OFFICIALS** Has the Disclosing Party provided any income or compensation to any City elected official during the 12-month period preceding the date of this EDS? [] Yes |x| No Does the Disclosing Party reasonably expect to provide any income or compensation to any City

No. 1 Standard of this FDS? [1 Yes [x] No. elected official during the 12-month period following the date of this EDS? [] Yes If "yes" to either of the above, please identify below the name(s) of such City elected official(s) and describe such income or compensation: Does any City elected official or, to the best of the Disclosing Party's knowledge after reasonable inquiry, any City elected official's spouse or domestic partner, have a financial interest (as defined in Chapter 2-156 of the Municipal Code of Chicago ("MCC")) in the Disclosing Party? [x] No [] Yes If "yes," please identify below the name(s) of such City elected official(s) and/or spouse(s)/domestic partner(s) and describe the financial interest(s).

limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none,

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
	A WARRANT OF STREET STREET, ST.	- A	
(Add sheets if necessary)			
[x] Check here if the Disc	closing Party	has not retained, nor expects to ret	ain, any such persons or entities.
SECTION V CERTIF	CATIONS	S	
A. COURT-ORDERED	CHILD SUF	PPORT COMPLIANCE	
		ntial owners of business entities the support obligations throughout the	
		ctly owns 10% or more of the Disc tions by any Illinois court of compe	
[] Yes [x] No []]	No person di	irectly or indirectly owns 10% or m	ore of the Disclosing Party.
If "Yes," has the person er is the person in compliance		court-approved agreement for payagreement?	ment of all support owed and
[]·Yes [] No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] In Party nor any Affiliated E performance of any public inspector general, or integ	n the 5-year intity [see de contract, the crity complia	ne Matter is a contract being handle period preceding the date of this El stinition in (5) below] has engaged, he services of an integrity monitor, in unce consultant (i.e., an individual of esignated by a public agency to help	OS, neither the Disclosing in connection with the ndependent private sector rentity with legal, auditing,

tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee,

activity of specified agency vendors as well as help the vendors reform their business practices so they

can be considered for agency contracts in the future, or continue with a contract in progress).

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity:
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Ver.2018-1 Page 5 of 15

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1

11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with "N/A" or "none"). None
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient. None
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
 The Disclosing Party certifies that the Disclosing Party (check one) is [x] is not
a "financial institution" as defined in MCC Section 2-32-455(b).
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a predatory lender may result in the loss of the privilege of doing business with the City."

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to

believe has not provided or cannot provide truthful certifications.

MCC Section 2-32	-455(b)) is a predatory lender with	because it or any of its affiliates (as defined in in the meaning of MCC Chapter 2-32, explain
here (attach additio	onal pages if necessary):	
4.0	PARAMETER 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	The second secon
	the word "None," or no response med that the Disclosing Party certi	appears on the lines above, it will be fied to the above statements.
D. CERTIFICATI	ON REGARDING FINANCIAL I	NTEREST IN CITY BUSINESS
Any words or term	s defined in MCC Chapter 2-156 h	have the same meanings if used in this Part D.
after reasonable inc		he best of the Disclosing Party's knowledge se of the City have a financial interest in his or entity in the Matter?
[] Yes	[x] No	
	cked "Yes" to Item D(1), proceed tems D(2) and D(3) and proceed to	to Items D(2) and D(3). If you checked "No" o Part E.
official or employe other person or enti taxes or assessment "City Property Sale	e shall have a financial interest in ty in the purchase of any property s, or (iii) is sold by virtue of legal	idding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, ten pursuant to the City's eminent domain he meaning of this Part D.
Does the Matter inv	volve a City Property Sale?	,
[] Yes	[] No	
		mes and business addresses of the City officials fy the nature of the financial interest:
Name	Business Address	Nature of Financial Interest
A41. 1 A41. 11		
	Party further certifies that no prohy official or employee.	ibited financial interest in the Matter will be

Vcr.2018-1

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.
x_1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.
A. CERTIFICATION REGARDING LOBBYING
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on

2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Ver.2018-1

Page 9 of 15

behalf of the Disclosing Party with respect to the Matter.)

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the	Applicant?
[] Yes	[x] No
If "Yes," answer the three	questions below:
Have you developed as federal regulations? (See [] Yes	nd do you have on file affirmative action programs pursuant to applicable 41 CFR Part 60-2.) [] No
Compliance Programs, or applicable filing requirement	e Joint Reporting Committee, the Director of the Office of Federal Contracthe Equal Employment Opportunity Commission all reports due under the ents? [] No [] Reports not required
3. Have you participated equal opportunity clause? [] Yes	in any previous contracts or subcontracts subject to the [] No
If you checked "No" to que	estion (1) or (2) above, please provide an explanation:

Page 10 of 15

Ver.2018-1

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing PERMANENT INELIGIBILITY for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

Milo Investments, LLC
(Print or type exact-legal name of Disclosing Party)
By: (Sign here)
Benjamin D. Mottola
(Print or type name of person signing)
Manager
(Print or type title of person signing)
Signed and sworn to before me on (date) (b 21 - , at, (state).
Notary Public
Commission expires: 5 5 2028

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?

L] res			[x] r	NO												
which	1 such	person	is con		; (3) th	e name	e and	title o	f the	elected	city o	fficia	l or d	epart	tmen	entity to t head to onship.	
					,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,						,		~~				
. **													****				

Ver.2018-1 Page 13 of 15

r 137--

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		10, is the Applicant or any Owner identified as a building code to MCC Section 2-92-416?
[] Yes	[x] No	
		ablicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
[] Yes	[] No	[x] The Applicant is not publicly traded on any exchange.
	offlaw or problem	entify below the name of each person or legal entity identified in landlord and the address of each building or buildings to which

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

[] Yes
[] No
[x] N/A – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

BENJAMIN D. MOTTOLA LIVING TRUST

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing Party submitt	ing this EDS. Include d/b/a/ if applicable:
Benjamin D. Mottola Living Trust	
Check ONE of the following three boxes:	
Indicate whether the Disclosing Party submittin 1. [] the Applicant OR 2. [x] a legal entity currently holding, or a	ng this EDS is: nticipated to hold within six months after City action on
the contract, transaction or other undertaking to	o which this EDS pertains (referred to below as the of 7.5% in the Applicant. State the Applicant's legal
State the legal name of the entity in which the	
B. Business address of the Disclosing Party:	46 Pilgrim Road Marblehead, MA 01945
	Marblehead, MA 01945
	Email: bdm@stratfordcapitalgroup.com
D. Name of contact person: Ben Mottola	mannan - skus - 10
E. Federal Employer Identification No. (if you	have one):
property, if applicable): Application by Applicant for various forms of City financial assistant (LIHTCs and IAHTCs), approximately \$3,000,000 City Loan of N	EDS pertains. (Include project number and location of the comproximately \$5,000,000 of Tax Increment Financing ["TIF"]; Tax Credits tulti-Program Funds (probably HOME); purchase of City property at reduced party located at 43rd St. and S. Calumet Ave.).
G. Which City agency or department is request	ting this EDS? DPD and DOH
If the Matter is a contract being handled by the complete the following:	City's Department of Procurement Services, please
Specification #	and Contract #
Ver.2018-1 Pa	ge 1 of 15

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY	í
 Indicate the nature of the Disclosing Pa Person Publicly registered business corporation Privately held business corporation Sole proprietorship General partnership Limited partnership Trust 	rty: [] Limited liability company [] Limited liability partnership [] Joint venture [] Not-for-profit corporation (Is the not-for-profit corporation also a 501(c)(3))? [] Yes [] No [] Other (please specify)
2. For legal entities, the state (or foreign coun	atry) of incorporation or organization, if applicable:
Massachusetts	
3. For legal entities not organized in the State business in the State of Illinois as a foreign ent	of Illinois: Has the organization registered to do ity?
[] Yes [x] No	[] Organized in Illinois
B. IF THE DISCLOSING PARTY IS A LEGA	AL ENTITY:
the entity; (ii) for not-for-profit corporations are no such members, write "no members whice similar entities, the trustee, executor, administ limited partnerships, limited liability compared.	plicable, of: (i) all executive officers and all directors of a, all members, if any, which are legal entities (if there ch are legal entities"); (iii) for trusts, estates or other trator, or similarly situated party; (iv) for general or unies, limited liability partnerships or joint ventures, ager or any other person or legal entity that directly or of the Applicant.
NOTE: Each legal entity listed below must sub	omit an EDS on its own behalf.
Name Benjamin D. Mottola	Title Trustee
indirect, current or prospective (i.e. within 6 me ownership) in excess of 7.5% of the Applicant.	oncerning each person or legal entity having a direct or onths after City action) beneficial interest (including Examples of such an interest include shares in a p or joint venture, interest of a member or manager in a

NOTE: Each legal entity listed below may be required to submit an EDS on its own behalf. Name Business Address Percentage Interest in the Applicant 46 Pilgrim Road, Marblehead, MA 01945 Benjamin D. Mottola SECTION III -- INCOME OR COMPENSATION TO, OR OWNERSHIP BY, CITY ELECTED **OFFICIALS** Has the Disclosing Party provided any income or compensation to any City elected official during the [| Yes 12-month period preceding the date of this EDS? [x] No Does the Disclosing Party reasonably expect to provide any income or compensation to any City elected official during the 12-month period following the date of this EDS? [] Yes If "yes" to either of the above, please identify below the name(s) of such City elected official(s) and describe such income or compensation: Does any City elected official or, to the best of the Disclosing Party's knowledge after reasonable inquiry, any City elected official's spouse or domestic partner, have a financial interest (as defined in Chapter 2-156 of the Municipal Code of Chicago ("MCC")) in the Disclosing Party? [x | No [] Yes If "yes," please identify below the name(s) of such City elected official(s) and/or spouse(s)/domestic partner(s) and describe the financial interest(s).

limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none,

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

state "None."

retained or anticipated to be retained)	Address	(subcontractor, attorney, lobbyist, etc.)	paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
		A MAR	
[x] Check here if the Disc	closing Part	y has not retained, nor expects to re	tain, any such persons or entities.
SECTION V CERTII	FICATION	S	
A. COURT-ORDERED	CHILD SUI	PPORT COMPLIANCE	
		antial owners of business entities the support obligations throughout the	
		ectly owns 10% or more of the Disc tions by any Illinois court of compe	
[]Yes [x]No []	No person d	irectly or indirectly owns 10% or m	ore of the Disclosing Party.
If "Yes," has the person e is the person in compliant		a court-approved agreement for payagreement?	ment of all support owed and
[] Yes [] No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] I Party nor any Affiliated E performance of any publi- inspector general, or integ	n the 5-year Entity <u>[see</u> do c contract, the grity complis	the Matter is a contract being handled period preceding the date of this Elefinition in (5) below] has engaged, the services of an integrity monitor, ance consultant (i.e., an individual clesignated by a public agency to hele	DS, neither the Disclosing in connection with the independent private sector or entity with legal, auditing,

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

activity of specified agency vendors as well as help the vendors reform their business practices so they

can be considered for agency contracts in the future, or continue with a contract in progress).

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Ver.2018-1

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1 Page 6 of 15

	ontractor/subcontractor that does not provide such certifications or that the Applicant has reason to elieve has not provided or cannot provide truthful certifications.
	1. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further ertifications), the Disclosing Party must explain below:
	the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively resumed that the Disclosing Party certified to the above statements.
m of	2. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a emplete list of all current employees of the Disclosing Party who were, at any time during the 12-onth period preceding the date of this EDS, an employee, or elected or appointed official, of the City Chicago (if none, indicate with "N/A" or "none"). None
th of m th pc	B. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a emplete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed ficial, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything ade generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or one"). As to any gift listed below, please also list the name of the City recipient. None
C.	CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
1.	The Disclosing Party certifies that the Disclosing Party (check one) [] is [x] is not
	a "financial institution" as defined in MCC Section 2-32-455(b).
2.	If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
ple Me	We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further edge that none of our affiliates is, and none of them will become, a predatory lender as defined in CC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a edatory lender may result in the loss of the privilege of doing business with the City."

MCC Section 2-32		because it or any of its affiliates (as defined in in the meaning of MCC Chapter 2-32, explain
		The state of the s
	the word "None," or no response med that the Disclosing Party certi	appears on the lines above, it will be fied to the above statements.
D. CERTIFICATI	ON REGARDING FINANCIAL I	NTEREST IN CITY BUSINESS
Any words or term	ns defined in MCC Chapter 2-156 h	have the same meanings if used in this Part D.
after reasonable in		he best of the Disclosing Party's knowledge se of the City have a financial interest in his or entity in the Matter?
[]Yes	[x] No	
	ecked "Yes" to Item D(1), proceed Items D(2) and D(3) and proceed t	to Items D(2) and D(3). If you checked "No" o Part E.
official or employed other person or ent taxes or assessment "City Property Sale	ee shall have a financial interest in ity in the purchase of any property ts, or (iii) is sold by virtue of legal	bidding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, ten pursuant to the City's eminent domain the meaning of this Part D.
Does the Matter in	volve a City Property Sale?	
[]Yes	[] No	
		mes and business addresses of the City officials fy the nature of the financial interest:
Name	Business Address	Nature of Financial Interest
	Andrew Market Control of the Control	
	Party further certifies that no proh ty official or employee.	ibited financial interest in the Matter will be

Ver.2018-1

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.
\underline{x} 1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.
A. CERTIFICATION REGARDING LOBBYING
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)

2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Ver.2018-1

Page 9 of 15

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party [] Yes	
If "Yes," answer the thi	ree questions below:
Have you develope federal regulations? (S Yes	•
Compliance Programs, applicable filing require	the Joint Reporting Committee, the Director of the Office of Federal Contract or the Equal Employment Opportunity Commission all reports due under the ements? [] No [] Reports not required
Have you participat equal opportunity claus [] Yes	
If you checked "No" to	question (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. NOTE: With respect to Matters subject to MCC Chapter 1-23, Article I (imposing PERMANENT INELIGIBILITY for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

Benjamin D. Mottola Living Trust
(Print or type exact legal name of Disclosing Party)
By:
(Sign here)
Benjamin D. Mottola
(Print or type name of person signing)
Touten
Trustee
(Print or type title of person signing)
 , , , ,
Signed and sworn to before me on (date) $\frac{7/15/2024}{}$,
at Millier County, Mariachuse Hr (state).
HILL RICH RECKING
Notary Public
Commission expires: 1/1/2024
701/20E- 101/20E- 101
W. O. W. W.

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?

[]Yes	[x] No		
which such pe	rson is connected; (3) the na	ame and title of such person, (2) the name of the legal ename and title of the elected city official or department his ship, and (4) the precise nature of such familial relations	nead to
	- or this now wart is some state	110 - 100 A	
and the same of th	THE STATE OF THE S		*

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		10, is the Applicant or any Owner identified as a building code to MCC Section 2-92-416?
[] Yes	[x] No	
		ablicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
[]Yes	[] No	[x] The Applicant is not publicly traded on any exchange.
	cofflaw or problen	entify below the name of each person or legal entity identified n landlord and the address of each building or buildings to which

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

[] Yes
[] No
[x] N/A – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

NJ ENTERPRISES LIMITED PARTNERSHIP

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing Party submitting this EDS. Include d/b/a/ if applicable: NJ Enterprises Limited Partnership			
Check ONE of the following three boxe	s:		
the contract, transaction or other undertak. "Matter"), a direct or indirect interest in exname: OR 3. [] a legal entity with a direct or income.	or anticipated to hold within six months after City action on ing to which this EDS pertains (referred to below as the excess of 7.5% in the Applicant. State the Applicant's legal chase I, LP direct right of control of the Applicant (see Section II(B)(1)) at the Disclosing Party holds a right of control:		
	Peabody, MA 01960		
C. Telephone: (978) 535-5600 Fax:	Email: jmn@stratfordcapitalgroup.com		
D. Name of contact person: John M. Nelson	n		
E. Federal Employer Identification No. (i	f you have one)		
property, if applicable): Application by Applicant for various forms of City financial (LIHTCs and IAHTCs): approximately \$3,000,000 City L	ch this EDS pertains. (Include project number and location of assistance (approximately \$5,000,000 of Tax Increment Financing ["FIF"]: Tax Credits oan of Multi-Program Funds (probably HOME): purchase of City property at reduced ent of property located at 43rd St. and S. Calumet Ave.).		
G. Which City agency or department is re	equesting this EDS? DPD and DOH		
If the Matter is a contract being handled by complete the following:	y the City's Department of Procurement Services, please		
Specification #	and Contract #		
Ver.2018-1	Page 1 of 15		

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE	E DISCLOSING PARTY	Ý
1. Indicate the nat [] Person [] Publicly registered [] Privately held busi [] Sole proprietorship [] General partnership [X Limited partnership [] Trust	ness corporation p	rity: [] Limited liability company [] Limited liability partnership [] Joint venture [] Not-for-profit corporation (Is the not-for-profit corporation also a 501(c)(3))? [] Yes [] No [] Other (please specify)
2. For legal entities, t	he state (or foreign coun	ntry) of incorporation or organization, if applicable:
Washington, DC	The state of the s	Address of the control of the contro
•	ot organized in the State f Illinois as a foreign en	e of Illinois: Has the organization registered to do tity?
[] Yes	[×] No	[] Organized in Illinois
B. IF THE DISCLOS	ING PARTY IS A LEG	AL ENTITY:
the entity; (ii) for not- are no such members, similar entities, the tr limited partnerships, each general partner, n	for-profit corporations write "no members which ustee, executor, adminis limited liability compa	oplicable, of: (i) all executive officers and all directors of s, all members, if any, which are legal entities (if there ch are legal entities"); (iii) for trusts, estates or other trator, or similarly situated party; (iv) for general or anies, limited liability partnerships or joint ventures, ager or any other person or legal entity that directly or t of the Applicant.
NOTE: Each legal ent	ity listed below must su	bmit an EDS on its own behalf.
Name John M. Nelson IV		Title Class A General Partner
421 Corp.		Class B General Partner
indirect, current or pro ownership) in excess o	spective (i.e. within 6 m f 7.5% of the Applicant	oncerning each person or legal entity having a direct or onths after City action) beneficial interest (including . Examples of such an interest include shares in a ip or joint venture, interest of a member or manager in a

limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None." NOTE: Each legal entity listed below may be required to submit an EDS on its own behalf. Business Address Percentage Interest in the Applicant Name None SECTION III -- INCOME OR COMPENSATION TO, OR OWNERSHIP BY, CITY ELECTED **OFFICIALS** Has the Disclosing Party provided any income or compensation to any City elected official during the 12-month period preceding the date of this EDS? [] Yes [x] No Does the Disclosing Party reasonably expect to provide any income or compensation to any City elected official during the 12-month period following the date of this EDS? [] Yes If "yes" to either of the above, please identify below the name(s) of such City elected official(s) and describe such income or compensation: Does any City elected official or, to the best of the Disclosing Party's knowledge after reasonable inquiry, any City elected official's spouse or domestic partner, have a financial interest (as defined in Chapter 2-156 of the Municipal Code of Chicago ("MCC")) in the Disclosing Party? [x] No [] Yes If "yes," please identify below the name(s) of such City elected official(s) and/or spouse(s)/domestic partner(s) and describe the financial interest(s).

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Ver.2018-1 Page 3 of 15

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
Address of the second s		- , 1940-1916 distinct in	
(Add sheets if necessary)			
[x] Check here if the Disc	closing Party	has not retained, nor expects to ret	tain, any such persons or entities.
SECTION V CERTIF	CATION	S	
A. COURT-ORDERED	CHILD SUI	PPORT COMPLIANCE	
		antial owners of business entities the support obligations throughout the	
· •	•	ectly owns 10% or more of the Disc tions by any Illinois court of compo	-
[] Yes [x] No [] I	No person d	irectly or indirectly owns 10% or m	ore of the Disclosing Party.
If "Yes," has the person en is the person in compliance		acourt-approved agreement for payagreement?	ment of all support owed and
[] Yes [] No			
B. FURTHER CERTIFIC	CATIONS		ı
Procurement Services.] In Party nor any Affiliated E performance of any public inspector general, or integ investigative, or other sim	n the 5-year Entity [<u>see</u> dec contract, the grity compliantion of the complianties of	he Matter is a contract being handled period preceding the date of this Election in (5) below] has engaged, the services of an integrity monitor, is ance consultant (i.e., an individual of esignated by a public agency to help s well as help the vendors reform the	DS, neither the Disclosing in connection with the independent private sector or entity with legal, auditing, p the agency monitor the

can be considered for agency contracts in the future, or continue with a contract in progress).

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

Ver.2018-1 Page 4 of 15

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Ver.2018-1

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
 - 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to believe has not provided or cannot provide truthful certifications.	0			
11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:				
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusive presumed that the Disclosing Party certified to the above statements.	 /ely			
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the of Chicago (if none, indicate with "N/A" or "none"). None	City			
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time durin the 12-month period preceding the execution date of this EDS, to an employee, or elected or appoint official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anythin made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient. None	ited ng			
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION				
 The Disclosing Party certifies that the Disclosing Party (check one) [] is [x] is not 				
a "financial institution" as defined in MCC Section 2-32-455(b).				
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:				
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a predatory lender may result in the loss of the privilege of doing business with the City."				

If the Disclosing Party is unable to make this pledge because it or any of its affiliates (as defined in MCC Section 2-32-455(b)) is a predatory lender within the meaning of MCC Chapter 2-32, explain here (attach additional pages if necessary):				
If the letters "NA,"		appears on the lines above, it will be tified to the above statements.		
D. CERTIFICATI	ON REGARDING FINANCIAL	INTEREST IN CITY BUSINESS		
Any words or term	ns defined in MCC Chapter 2-156	have the same meanings if used in this Part D.		
after reasonable in		the best of the Disclosing Party's knowledge ee of the City have a financial interest in his or entity in the Matter?		
[] Yes	[x] No			
	ecked "Yes" to Item D(1), proceed Items D(2) and D(3) and proceed	to Items D(2) and D(3). If you checked "No" to Part E.		
official or employed other person or ent taxes or assessment "City Property Sale	ee shall have a financial interest in ity in the purchase of any property its, or (iii) is sold by virtue of lega	bidding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for I process at the suit of the City (collectively, ken pursuant to the City's eminent domain the meaning of this Part D.		
Does the Matter in	volve a City Property Sale?			
[]Yes	[] No			
		ames and business addresses of the City officials tify the nature of the financial interest:		
Name	Business Address	Nature of Financial Interest		
4. The Disclosing		hibited financial interest in the Matter will be		

Ver.2018-1

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

)

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.
x_1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.
A. CERTIFICATION REGARDING LOBBYING
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):

(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)

2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Ver.2018-1

Page 9 of 15

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the	
[] Yes	[X] No
If "Yes," answer the three of	uestions below:
Have you developed an federal regulations? (See 4 [] Yes	do you have on file affirmative action programs pursuant to applicable 1 CFR Part 60-2.) [] No
Compliance Programs, or tapplicable filing requirement	
[] Yes	[] No [] Reports not required
3. Have you participated i equal opportunity clause?	any previous contracts or subcontracts subject to the
[Yes	[] No
If you checked "No" to que	tion (1) or (2) above, please provide an explanation:
· · · · · · · · · · · · · · · · · · ·	The state of the s
TOWN TO A A STATE STATE OF THE	THE RESERVE OF SECTION OF SECTION AND SECTION ASSESSMENT AND SECTION ASSESSMENT ASSESSME

Page 10 of 15

Ver.2018-1

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

NJ Enterprises Limited Partnership	
(Print or type exact legal name of Disclosing Party)	
By:	
(Sign here)	
John M. Nelson, IV	
(Print or type name of person signing)	
General Partner	
(Print or type title of person signing)	
Signed and sworn to before me on (date) 6 16 2021,	
at FSSEX County, WA (state).	SALE TO
MMYanos Notary Public	Weston State
ONotary Public	
Commission expires: $5/5/2028$	ONWEAL THE

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?

[] Yes	[x] No				
which such person	entify below (1) the nais connected; (3) the nais a familial relations	ame and title of th	e elected city of	ficial or depar	tment head to
	Annual Marketin and the first of the second		· *		
As some respect to the	Apparation and the second section of the second second section of the second second section section sections as				

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

1. Pursuant to MCC Section 2-154-010, is the Applicant or any Owner identified as a building coscofflaw or problem landlord pursuant to MCC Section 2-92-416?			
	[] Yes	[x] No	
th	**		icly traded on any exchange, is any officer or director of de scofflaw or problem landlord pursuant to MCC Section
	[] Yes	[] No	[x] The Applicant is not publicly traded on any exchange.
as	. , , , ,	v or problem l	tify below the name of each person or legal entity identified andlord and the address of each building or buildings to which
,	17.1 III III III III III III III III III I		

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

[] Yes
[] No
[x] N/A – Lam not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.
·

421 CORP.

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION 1 -- GENERAL INFORMATION

A. Legal name of the Disclosing Party submitting this EDS. Include d/b/a/ if applicable:
421 Corp.
Check ONE of the following three boxes:
Indicate whether the Disclosing Party submitting this EDS is: 1. [] the Applicant OR 2. [] a legal entity currently holding, or anticipated to hold within six months after City action or the contract, transaction or other undertaking to which this EDS pertains (referred to below as the "Matter"), a direct or indirect interest in excess of 7.5% in the Applicant. State the Applicant's legal name: OR
3. [x] a legal entity with a direct or indirect right of control of the Applicant (see Section II(B)(1 State the legal name of the entity in which the Disclosing Party holds a right of control: Stratford 43 Green Investors Limited Partnership
B. Business address of the Disclosing Party: 100 Corporate Place, Suite 404 Peabody, MA 01960
C. Telephone: 978-535-5600 Fax:Email: jmn@stratfordcapitalgroup.com
D. Name of contact person: John M. Nelson
E. Federal Employer Identification No. (if you have one):
F. Brief description of the Matter to which this EDS pertains. (Include project number and location property, if applicable):
Application by Applicant for various forms of City financial assistance (approximately \$5,000,000 of Tax Increment Financing ["TIF ax Credits (LIHTCs and IAHTCs); approximately \$3,000,000 City Loan of Multi-Program Funds (probably HOME); purchase of Coroperty at reduced price) to facilitate development of property located at 43rd St. and S. Calumet Ave.
G. Which City agency or department is requesting this EDS? <u>DOH & DPD</u>
If the Matter is a contract being handled by the City-s Department of Procurement Services, please complete the following:
Specification # and Contract # Ver.2018-1 Page 1 of 15

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PART	Y
 Indicate the nature of the Disclosing Pa Person Publicly registered business corporation Privately held business corporation Sole proprietorship General partnership Limited partnership Trust 	arty: [] Limited liability company [] Limited liability partnership [] Joint venture [] Not-for-profit corporation (Is the not-for-profit corporation also a 501(c)(3))? [] Yes [] No [] Other (please specify)
2. For legal entities, the state (or foreign cour	ntry) of incorporation or organization, if applicable:
Virginia	
3. For legal entities not organized in the State business in the State of Illinois as a foreign en	e of Illinois: Has the organization registered to do tity?
[] Yes [X] No	[] Organized in Illinois
B. IF THE DISCLOSING PARTY IS A LEGAL ENTITY:	
the entity; (ii) for not-for-profit corporations are no such members, write "no members whice similar entities, the trustee, executor, administ limited partnerships, limited liability compared	oplicable, of: (i) all executive officers and all directors of s, all members, if any, which are legal entities (if there ch are legal entities"); (iii) for trusts, estates or other strator, or similarly situated party; (iv) for general or anies, limited liability partnerships or joint ventures, ager or any other person or legal entity that directly or t of the Applicant.
NOTE: Each legal entity listed below must su	bmit an EDS on its own behalf.
Name	Title
John M. Nelson, IV	President
indirect, current or prospective (i.e. within 6 m	oncerning each person or legal entity having a direct or nonths after City action) beneficial interest (including . Examples of such an interest include shares in a

corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

Ver.2018-1 Page 2 of 15

limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state ANone.@ **NOTE**: Each legal entity listed below may be required to submit an EDS on its own behalf. Name Business Address Percentage Interest in the Applicant None SECTION III -- INCOME OR COMPENSATION TO, OR OWNERSHIP BY, CITY ELECTED **OFFICIALS** Has the Disclosing Party provided any income or compensation to any City elected official during the 12-month period preceding the date of this EDS? [] Yes [x] No Does the Disclosing Party reasonably expect to provide any income or compensation to any City elected official during the 12-month period following the date of this EDS? [] Yes No If "yes" to either of the above, please identify below the name(s) of such City elected official(s) and describe such income or compensation: Does any City elected official or, to the best of the Disclosing Party-s knowledge after reasonable inquiry, any City elected official-s spouse or domestic partner, have a financial interest (as defined in Chapter 2-156 of the Municipal Code of Chicago (AMCC@)) in the Disclosing Party? [] Yes [x] No If "yes," please identify below the name(s) of such City elected official(s) and/or spouse(s)/domestic partner(s) and describe the financial interest(s). SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney. lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: Ahourly rate@ or At.b.d.@ is not an acceptable response.
(Add sheets if necessary))		
[x] Check here if the Dis	sclosing Part	y has not retained, nor expects to re	tain, any such persons or entities.
SECTION V CERTIF	FICATION	s	
A. COURT-ORDERED	CHILD SUI	PPORT COMPLIANCE	
		antial owners of business entities the support obligations throughout the	•
		ectly owns 10% or more of the Disc tions by any Illinois court of compe	
[]Yes [x]No[]]	No person di	rectly or indirectly owns 10% or mo	ore of the Disclosing Party.
If AYes,@ has the person e is the person in complian		court-approved agreement for payr agreement?	nent of all support owed and
[] Yes [] No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] In Party nor any Affiliated E performance of any publi inspector general, or integ investigative, or other sin	n the 5-year Entity <u>[see</u> de c contract, the grity complision	he Matter is a contract being handle period preceding the date of this EI efinition in (5) below] has engaged, he services of an integrity monitor, ance consultant (i.e., an individual of lesignated by a public agency to help well as help the vendors reform the	OS, neither the Disclosing in connection with the independent private sector or entity with legal, auditing, p the agency monitor the

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago. including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

can be considered for agency contracts in the future, or continue with a contract in progress).

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft: forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any AContractor@ (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, ADisclosure of Subcontractors and Other Retained Parties@);
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Ver.2018-1 Page 5 of 15

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any Acontrolling person@ [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud. forgery, perjury, dishonesty or deceit against an officer or employee of the City or any Asister agency@; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Articles permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management (ASAM®).
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

Ver.2018-1 Page 6 of 15

11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.
12. To the best of the Disclosing Party=s knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with AN/A@ or Anone@). None
13. To the best of the Disclosing Party-s knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed official, of the City of Chicago. For purposes of this statement, a Agift@ does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with AN/A@ or Anone@). As to any gift listed below, please also list the name of the City recipient. None
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
 The Disclosing Party certifies that the Disclosing Party (check one) is [x] is not
a "financial institution" as defined in MCC Section 2-32-455(b).
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a predatory lender may result in the loss of the privilege of doing business with the City."

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to

believe has not provided or cannot provide truthful certifications.

MCC Section 2-32-4		because it or any of its affiliates (as defined in nin the meaning of MCC Chapter 2-32, explain
	ne word "None," or no response ed that the Disclosing Party cert	appears on the lines above, it will be ified to the above statements.
D. CERTIFICATION	N REGARDING FINANCIAL	NTEREST IN CITY BUSINESS
Any words or terms of	defined in MCC Chapter 2-156	nave the same meanings if used in this Part D.
after reasonable inqu		he best of the Disclosing Party-s knowledge ee of the City have a financial interest in his or entity in the Matter?
[] Yes	[x] No	
-	ed "Yes" to Item D(1), proceed ms D(2) and D(3) and proceed t	to Items D(2) and D(3). If you checked "No" o Part E.
official or employees other person or entity taxes or assessments, "City Property Sale")	shall have a financial interest in in the purchase of any property or (iii) is sold by virtue of legal	bidding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, ten pursuant to the City's eminent domain the meaning of this Part D.
Does the Matter invo	lve a City Property Sale?	
[] Yes	[] No	
		mes and business addresses of the City officials ify the nature of the financial interest:
Name	Business Address	Nature of Financial Interest
4. The Disclosing Pacquired by any City	•	nibited financial interest in the Matter will be

Ver.2018-1 Page 8 of 15

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.
x1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not
federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.
A. CERTIFICATION REGARDING LOBBYING
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)
2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee

Page 9 of 15

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the	Applicant?	
[] Yes	[x] No	
If AYes,@ answer the three	questions belo	ow:
 Have you developed ar federal regulations? (See 4 Yes 		ve on file affirmative action programs pursuant to applicable (0-2.)
	the Equal Em	ting Committee, the Director of the Office of Federal Contract ployment Opportunity Commission all reports due under the
[] Yes	[] No	[] Reports not required
3. Have you participated i equal opportunity clause?	n any previou	us contracts or subcontracts subject to the
[] Yes	[] No	
If you checked ANo@ to que	estion (1) or (2	2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City-s Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. **NOTE:** With respect to Matters subject to MCC Chapter 1-23, Article I (imposing **PERMANENT INELIGIBILITY** for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

421 Corp.	
(Print or type exact legal name of Disclosing Party)	
Ву:	
(Sign here)	
John M. Nelson, IV	
(Print or type name of person signing)	
President	
(Print or type title of person signing)	
Signed and sworn to before me on (date) 7/15/2	2021
21 Millesen County, Massachuselts (state).	
-22	BALL RIC R ME
Notary Public	ONNOTARY
Commission expires: 11/1/2024	SLIC ONNESS OF MASS

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any AApplicable Party® or any Spouse or Domestic Partner thereof currently has a Afamilial relationship® with any elected city official or department head. A Afamilial relationship® exists if, as of the date this EDS is signed, the Disclosing Party or any AApplicable Party® or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

AApplicable Party@ means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. APrincipal officers@ means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any AApplicable Party@ or any Spouse or Domestic Partner thereof currently have a Afamilial relationship@ with an elected city official or department head?

[] Yes	[x] No				
which such person	dentify below (1) the is connected; (3) th has a familial relati	e name and title	of the electe	d city official or	department head to

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		10, is the Applicant or any Owner identified as a building code to MCC Section 2-92-416?		
[] Yes	[x] No			
2. If the Applicant is a legal entity publicly traded on any exchange, is any officer or director of the Applicant identified as a building code scofflaw or problem landlord pursuant to MCC Section 2-92-416?				
[] Yes	[] No	[x] The Applicant is not publicly traded on any exchange.		
3. If yes to (1) or (2) above, please identify below the name of each person or legal entity identified as a building code scofflaw or problem landlord and the address of each building or buildings to which the pertinent code violations apply.				

7 81

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

[] Yes
[] No
[x] N/A – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

WILSON VENTURES II LLC

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing Wilson Ventures II LLC	, Party submitti	ng this EDS. Include d/b/a/ if applicable:
Check ONE of the following th	ree boxes:	
Indicate whether the Disclosing	Party submittin	g this EDS is:
1. [] the Applicant		
the contract, transaction or other "Matter"), a direct or indirect int name: 43rd and Calumet Phase I OR	undertaking to erest in excess , LP	anticipated to hold within six months after City action on which this EDS pertains (referred to below as the of 7.5% in the Applicant. State the Applicant's legal
· . · · · ·		ight of control of the Applicant (see Section II(B)(1)) Disclosing Party holds a right of control:
B. Business address of the Discl	osing Party:	8245 Boone Blvd., Suite 640
		Vienna, VA 22180
C. Telephone: (703) 942-6610	Fax:	Email: spw@stratfordcapitalgroup.con
D. Name of contact person: Step	ohen P. Wilson	
E. Federal Employer Identificat	ion No. (if you	have one):
property, if applicable): Application by Applicant for various for ["TIF"]; Tax Credits (LIHTCs and IAHT)	orms of City financi TCs); approximatel	EDS pertains. (Include project number and location of all assistance (approximately \$5,000,000 of Tax Increment Financing by \$3,000,000 City Loan of Multi-Program Funds (probably HOME); velopment of property located at 43rd St. and S. Calumet Ave.).
G. Which City agency or departs	nent is request	ing this EDS? DPD and DOH
If the Matter is a contract being to complete the following:	nandled by the	City's Department of Procurement Services, please
Specification #		and Contract #
Ver.2018-1	Pag	ge 1 of 15

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PARTY 1. Indicate the nature of the Disclosing Party: [| Person [x] Limited liability company [] Publicly registered business corporation [] Limited liability partnership [] Joint venture Privately held business corporation [] Sole proprietorship [] Not-for-profit corporation [] General partnership (Is the not-for-profit corporation also a 501(c)(3))? [] Limited partnership []Yes [] No [] Trust [] Other (please specify) 2. For legal entities, the state (or foreign country) of incorporation or organization, if applicable: Delaware 3. For legal entities not organized in the State of Illinois: Has the organization registered to do business in the State of Illinois as a foreign entity? []Yes [x] No [] Organized in Illinois B. IF THE DISCLOSING PARTY IS A LEGAL ENTITY: List below the full names and titles, if applicable, of: (i) all executive officers and all directors of the entity; (ii) for not-for-profit corporations, all members, if any, which are legal entities (if there are no such members, write "no members which are legal entities"); (iii) for trusts, estates or other similar entities, the trustee, executor, administrator, or similarly situated party; (iv) for general or limited partnerships, limited liability companies, limited liability partnerships or joint ventures, each general partner, managing member, manager or any other person or legal entity that directly or indirectly controls the day-to-day management of the Applicant. NOTE: Each legal entity listed below must submit an EDS on its own behalf. Title Name Stephen P. Wilson Manager 2. Please provide the following information concerning each person or legal entity having a direct or indirect, current or prospective (i.e. within 6 months after City action) beneficial interest (including ownership) in excess of 7.5% of the Applicant. Examples of such an interest include shares in a

corporation, partnership interest in a partnership or joint venture, interest of a member or manager in a

limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none, state "None."

NOTE: Each legal entity listed below may be required to submit an EDS on its own behalf.

Name	Business Address	Percentage Interest in the Applicant
Stephen P Wilson	8245 Boone Blvd., Suite 640, Vienn	a, VA 22180
Rosemary Wilson	8245 Boone Blvd., Suite 640, Vienna	
SECTION III IN OFFICIALS	ICOME OR COMPENSATION	N TO, OR OWNERSHIP BY, CITY ELECTED
_	Party provided any income or conceeding the date of this EDS?	npensation to any City elected official during the [] Yes [x] No
	Party reasonably expect to proving the 12-month period following	de any income or compensation to any City the date of this EDS? [] Yes [x] No
If "yes" to either of the describe such income		he name(s) of such City elected official(s) and
	1	
inquiry, any City ele		isclosing Party's knowledge after reasonable partner, have a financial interest (as defined in ICC")) in the Disclosing Party?
	tify below the name(s) of such Cibe the financial interest(s).	ty elected official(s) and/or spouse(s)/domestic
THE CONTRACT A COMMENTAL WAS ASSUMED THE TAX TO SEE A COMMENTAL OF THE CONTRACT OF THE CONTRAC		

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Business Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Pecs (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
(Add sheets if necessary)			
[x] Check here if the Disc	closing Party	y has not retained, nor expects to re-	tain, any such persons or entities.
SECTION V CERTII	FICATION	S	
A. COURT-ORDERED	CHILD SUI	PPORT COMPLIANCE	
		antial owners of business entities the support obligations throughout the	
		ectly owns 10% or more of the Disc tions by any Illinois court of compo	
[] Yes [x] No [] ?	No person d	irectly or indirectly owns 10% or m	ore of the Disclosing Party.
If "Yes," has the person e is the person in compliance		a court-approved agreement for payagreement?	ment of all support owed and
[]Yes []No		,	
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] In Party nor any Affiliated Experiormance of any public inspector general, or integrative, or other simple simple simple simple.	n the 5-year Entity <u>[see</u> de c contract, th grity complianilar skills, d	he Matter is a contract being handle period preceding the date of this El efinition in (5) below has engaged, he services of an integrity monitor, is ance consultant (i.e., an individual of designated by a public agency to hel as well as help the vendors reform the	DS, neither the Disclosing in connection with the independent private sector or entity with legal, auditing, p the agency monitor the

can be considered for agency contracts in the future, or continue with a contract in progress).

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

Ver.2018-1 Page 4 of 15

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to believe has not provided or cannot provide truthful certifications.
11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with "N/A" or "none"). None
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient. None
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
 The Disclosing Party certifies that the Disclosing Party (check one) [] is [x] is not
a "financial institution" as defined in MCC Section 2-32-455(b).
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a predatory lender may result in the loss of the privilege of doing business with the City."

MCC Section 2-32-4		because it or any of its affiliates (as defined in in the meaning of MCC Chapter 2-32, explain
A ** concentration of the company of		
	ne word "None," or no response a ed that the Disclosing Party certi	appears on the lines above, it will be fied to the above statements.
D. CERTIFICATION	N REGARDING FINANCIAL I	NTEREST IN CITY BUSINESS
Any words or terms of	lefined in MCC Chapter 2-156 h	ave the same meanings if used in this Part D.
after reasonable inqui		ne best of the Disclosing Party's knowledge e of the City have a financial interest in his or ntity in the Matter?
[] Yes	[x] No	
•	red "Yes" to Item D(1), proceed ms D(2) and D(3) and proceed to	to Items D(2) and D(3). If you checked "No" o Part E.
official or employee so other person or entity taxes or assessments, "City Property Sale").	shall have a financial interest in hin the purchase of any property or (iii) is sold by virtue of legal	idding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, en pursuant to the City's eminent domain he meaning of this Part D.
Does the Matter invol	ve a City Property Sale?	
[] Yes	[] No	
		nes and business addresses of the City officials fy the nature of the financial interest:
Name	Business Address	Nature of Financial Interest
4. The Disclosing Pa acquired by any City of		ibited financial interest in the Matter will be

Ver 2018-1

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

Please check either (1) or (2) below. If the Disclosing Party checks (2), the Disclosing Party must disclose below or in an attachment to this EDS all information required by (2). Failure to comply with these disclosure requirements may make any contract entered into with the City in connection with the Matter voidable by the City.

connection with the Matter Voldable by the City.
x 1. The Disclosing Party verifies that the Disclosing Party has searched any and all records of the Disclosing Party and any and all predecessor entities regarding records of investments or profits from slavery or slaveholder insurance policies during the slavery era (including insurance policies issued to slaveholders that provided coverage for damage to or injury or death of their slaves), and the Disclosing Party has found no such records.
2. The Disclosing Party verifies that, as a result of conducting the search in step (1) above, the Disclosing Party has found records of investments or profits from slavery or slaveholder insurance policies. The Disclosing Party verifies that the following constitutes full disclosure of all such records, including the names of any and all slaves or slaveholders described in those records:
SECTION VI CERTIFICATIONS FOR FEDERALLY FUNDED MATTERS
NOTE: If the Matter is federally funded, complete this Section VI. If the Matter is not federally funded, proceed to Section VII. For purposes of this Section VI, tax credits allocated by the City and proceeds of debt obligations of the City are not federal funding.
A. CERTIFICATION REGARDING LOBBYING
1. List below the names of all persons or entities registered under the federal Lobbying Disclosure Act of 1995, as amended, who have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter: (Add sheets if necessary):
(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None"

appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)

2. The Disclosing Party has not spent and will not expend any federally appropriated funds to party.

2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Ver.2018-1

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs $\Lambda(1)$ through $\Lambda(4)$ above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

Is the Disclosing Party the Applicant?

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Ver.2018-1		Page 10 of 15
-		2) above, please provide an explanation:
equal opportunity clause? [] Yes	[] No	
	in any previou	us contracts or subcontracts subject to the
Compliance Programs, or applicable filing requirement	the Equal Emp ents?	ing Committee, the Director of the Office of Federal Contract ployment Opportunity Commission all reports due under the [] Reports not required
 Have you developed at federal regulations? (See Yes 	41 CFR Part 6	re on file affirmative action programs pursuant to applicable 60-2.)
If "Yes," answer the three	questions belo	ow:
[]Yes	[x] No	

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. NOTE: With respect to Matters subject to MCC Chapter 1-23, Article I (imposing PERMANENT INELIGIBILITY for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

Wilson Ventures II LLC
(Print or type exact legal name of Disclosing Party)
By: Allo
(Sign here)
Stephen P. Wilson
(Print or type name of person signing)
Manager
(Print or type title of person signing)
Signed and sworn to before me on (date) 6 17 2021,
at ESSEY County, WA (state).
Notary Public Notary Public
Commission expires: 5/5/2028

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?

] Yes	[x] No			
which such person	entify below (1) the na is connected; (3) the na has a familial relations	ame and title of the e	lected city official or o	lepartment head to
V				119 - 1 1 1 1
			a commercial of the second control of the se	
		**************************************	WILL WOM 50 1	

C 137

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

		10, is the Applicant or any Owner identified as a building code to MCC Section 2-92-416?
[] Yes	[x] No	
		blicly traded on any exchange, is any officer or director of code scofflaw or problem landlord pursuant to MCC Section
[J Yes	[] No	[x] The Applicant is not publicly traded on any exchange.
	offlaw or problem	entify below the name of each person or legal entity identified a landlord and the address of each building or buildings to which
		TO THE RESERVE THE STATE OF THE
	All Marketine All All All All All All All All All Al	

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

[] Yes
[] No
[x] N/A – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.

DOUBLEWIDE LLC

02021-3265

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT

SECTION I -- GENERAL INFORMATION

A. Legal name of the Disclosing Party	submitting this EDS. Inc	lude d/b/a/ if applicable:	
Doublewide LLC Check ONE of the following three boxes:			
1. [] the Applicant			
on the contract, transaction or other	undertaking to which this	I within six months after City action EDS pertains (referred to below as the ne Applicant. State the Applicant's	
OR —43rd and Calumet P	hase I Limited Partnership		
	_	f the Applicant (see Section II(B)(1)) olds a right of control:	
B. Business address of the Disclosing		te 404	
	Peabody. MA 01960	** ***********************************	
C. Telephone: (978) 535- 5600 F	ax:	Email: kf\w\aigstratfordcapitalgroup.com	
D. Name of contact person: Kyle Wolff			
E. Federal Employer Identification No	<i>7</i>		
F. Brief description of the Matter to w property, if applicable):	hich this EDS pertains. (In	nclude project number and location of	
Application by Applicant for various forms of City fir Credits (LIHTCs and IAHTCs); approximately \$3,000 at reduced price) to facilitate de		ids (probably HOMF), purchase of City property	
G. Which City agency or department is	s requesting this EDS? D	PD and DOH	
If the Matter is a contract being handle complete the following:	d by the City's Department	t of Procurement Services, please	
Specification #	and Contract #	· · · · · · · · · · · · · · · · · · ·	
Ver.2018-1	Page 1 of 15		

SECTION II -- DISCLOSURE OF OWNERSHIP INTERESTS

A. NATURE OF THE DISCLOSING PART	Y
 Indicate the nature of the Disclosing Paperson Publicly registered business corporation Privately held business corporation Sole proprietorship General partnership Limited partnership Trust 	[x] Limited liability company
2. For legal entities, the state (or foreign cour	ntry) of incorporation or organization, if applicable:
	e of Illinois: Has the organization registered to dontity?
[] Yes [x] No	[] Organized in Illinois
B. IF THE DISCLOSING PARTY IS A LEG	GAL ENTITY:
the entity; (ii) for not-for-profit corporation are no such members, write "no members whit similar entities, the trustee, executor, administimited partnerships, limited liability comp	pplicable, of: (i) all executive officers and all directors of is, all members, if any, which are legal entities (if there ich are legal entities"); (iii) for trusts, estates or other strator, or similarly situated party; (iv) for general or anies, limited liability partnerships or joint ventures, ager or any other person or legal entity that directly or at of the Applicant.
NOTE: Each legal entity listed below must su	abmit an EDS on its own behalf.
Name Kyle F Wolff	Title Sole Member, Manager
indirect, current or prospective (i.e. within 6 mownership) in excess of 7.5% of the Applicant	

Name	Business Address	Percentage Interest in the Applicant
Kyle F Wolff	100 Corporate Place Suite 404, Peabody MA 01960	
SECTION III OFFICIALS	INCOME OR COMPENSATION	TO, OR OWNERSHIP BY, CITY ELECTE
	sing Party provided any income or comp od preceding the date of this EDS?	ensation to any City elected official during the [] Yes [x] No
	osing Party reasonably expect to provide during the 12-month period following the	any income or compensation to any City ne date of this EDS? [] Yes [x] No
	er of the above, please identify below the ncome or compensation:	name(s) of such City elected official(s) and
inquiry, any C		losing Party's knowledge after reasonable artner, have a financial interest (as defined in C")) in the Disclosing Party?
	e identify below the name(s) of such City describe the financial interest(s).	elected official(s) and/or spouse(s)/domestic

limited liability company, or interest of a beneficiary of a trust, estate or other similar entity. If none,

SECTION IV -- DISCLOSURE OF SUBCONTRACTORS AND OTHER RETAINED PARTIES

The Disclosing Party must disclose the name and business address of each subcontractor, attorney, lobbyist (as defined in MCC Chapter 2-156), accountant, consultant and any other person or entity whom the Disclosing Party has retained or expects to retain in connection with the Matter, as well as the nature of the relationship, and the total amount of the fees paid or estimated to be paid. The Disclosing Party is not required to disclose employees who are paid solely through the Disclosing Party's regular payroll. If the Disclosing Party is uncertain whether a disclosure is required under this Section, the Disclosing Party must either ask the City whether disclosure is required or make the disclosure.

Name (indicate whether retained or anticipated to be retained)	Address	Relationship to Disclosing Party (subcontractor, attorney, lobbyist, etc.)	Fees (indicate whether paid or estimated.) NOTE: "hourly rate" or "t.b.d." is not an acceptable response.
[x] Check here if the Disc	closing Party	y has not retained, nor expects to ret	ain, any such persons or entities.
SECTION V CERTII	FICATION	S	
A. COURT-ORDERED	CHILD SUI	PPORT COMPLIANCE	
	·-	antial owners of business entities the support obligations throughout the	•
		ectly owns 10% or more of the Disc tions by any Illinois court of compe	
[] Yes [x] No []]	No person d	irectly or indirectly owns 10% or m	ore of the Disclosing Party.
If "Yes," has the person er is the person in compliance		a court-approved agreement for payaggreement?	ment of all support owed and
[] Yes [] No			
B. FURTHER CERTIFIC	CATIONS		
Procurement Services.] In Party nor any Affiliated E performance of any public	n the 5-year intity [see decontract, the	he Matter is a contract being handle period preceding the date of this El efinition in (5) below] has engaged, ne services of an integrity monitor, is ance consultant (i.e., an individual of	OS, neither the Disclosing in connection with the ndependent private sector

2. The Disclosing Party and its Affiliated Entities are not delinquent in the payment of any fine, fee, tax or other source of indebtedness owed to the City of Chicago, including, but not limited to, water and sewer charges, license fees, parking tickets, property taxes and sales taxes, nor is the Disclosing Party delinquent in the payment of any tax administered by the Illinois Department of Revenue.

investigative, or other similar skills, designated by a public agency to help the agency monitor the activity of specified agency vendors as well as help the vendors reform their business practices so they

can be considered for agency contracts in the future, or continue with a contract in progress).

- 3. The Disclosing Party and, if the Disclosing Party is a legal entity, all of those persons or entities identified in Section II(B)(1) of this EDS:
- a. are not presently debarred, suspended, proposed for debarment, declared ineligible or voluntarily excluded from any transactions by any federal, state or local unit of government;
- b. have not, during the 5 years before the date of this EDS, been convicted of a criminal offense, adjudged guilty, or had a civil judgment rendered against them in connection with: obtaining, attempting to obtain, or performing a public (federal, state or local) transaction or contract under a public transaction; a violation of federal or state antitrust statutes; fraud; embezzlement; theft; forgery; bribery; falsification or destruction of records; making false statements; or receiving stolen property;
- c. are not presently indicted for, or criminally or civilly charged by, a governmental entity (federal, state or local) with committing any of the offenses set forth in subparagraph (b) above;
- d. have not, during the 5 years before the date of this EDS, had one or more public transactions (federal, state or local) terminated for cause or default; and
- e. have not, during the 5 years before the date of this EDS, been convicted, adjudged guilty, or found liable in a civil proceeding, or in any criminal or civil action, including actions concerning environmental violations, instituted by the City or by the federal government, any state, or any other unit of local government.
- 4. The Disclosing Party understands and shall comply with the applicable requirements of MCC Chapters 2-56 (Inspector General) and 2-156 (Governmental Ethics).
- 5. Certifications (5), (6) and (7) concern:
 - the Disclosing Party;
 - any "Contractor" (meaning any contractor or subcontractor used by the Disclosing Party in connection with the Matter, including but not limited to all persons or legal entities disclosed under Section IV, "Disclosure of Subcontractors and Other Retained Parties");
 - any "Affiliated Entity" (meaning a person or entity that, directly or indirectly: controls the Disclosing Party, is controlled by the Disclosing Party, or is, with the Disclosing Party, under common control of another person or entity). Indicia of control include, without limitation: interlocking management or ownership; identity of interests among family members, shared facilities and equipment; common use of employees; or organization of a business entity following the ineligibility of a business entity to do business with federal or state or local government, including the City, using substantially the same management, ownership, or principals as the ineligible entity. With respect to Contractors, the term Affiliated Entity means a person or entity that directly or indirectly controls the Contractor, is controlled by it, or, with the Contractor, is under common control of another person or entity;
 - any responsible official of the Disclosing Party, any Contractor or any Affiliated Entity or any other official, agent or employee of the Disclosing Party, any Contractor or any Affiliated Entity, acting pursuant to the direction or authorization of a responsible official of the Disclosing Party, any Contractor or any Affiliated Entity (collectively "Agents").

Neither the Disclosing Party, nor any Contractor, nor any Affiliated Entity of either the Disclosing Party or any Contractor, nor any Agents have, during the 5 years before the date of this EDS, or, with respect to a Contractor, an Affiliated Entity, or an Affiliated Entity of a Contractor during the 5 years before the date of such Contractor's or Affiliated Entity's contract or engagement in connection with the Matter:

- a. bribed or attempted to bribe, or been convicted or adjudged guilty of bribery or attempting to bribe, a public officer or employee of the City, the State of Illinois, or any agency of the federal government or of any state or local government in the United States of America, in that officer's or employee's official capacity;
- b. agreed or colluded with other bidders or prospective bidders, or been a party to any such agreement, or been convicted or adjudged guilty of agreement or collusion among bidders or prospective bidders, in restraint of freedom of competition by agreement to bid a fixed price or otherwise; or
- c. made an admission of such conduct described in subparagraph (a) or (b) above that is a matter of record, but have not been prosecuted for such conduct; or
- d. violated the provisions referenced in MCC Subsection 2-92-320(a)(4)(Contracts Requiring a Base Wage); (a)(5)(Debarment Regulations); or (a)(6)(Minimum Wage Ordinance).
- 6. Neither the Disclosing Party, nor any Affiliated Entity or Contractor, or any of their employees, officials, agents or partners, is barred from contracting with any unit of state or local government as a result of engaging in or being convicted of (1) bid-rigging in violation of 720 ILCS 5/33E-3; (2) bid-rotating in violation of 720 ILCS 5/33E-4; or (3) any similar offense of any state or of the United States of America that contains the same elements as the offense of bid-rigging or bid-rotating.
- 7. Neither the Disclosing Party nor any Affiliated Entity is listed on a Sanctions List maintained by the United States Department of Commerce, State, or Treasury, or any successor federal agency.
- 8. [FOR APPLICANT ONLY] (i) Neither the Applicant nor any "controlling person" [see MCC Chapter 1-23, Article I for applicability and defined terms] of the Applicant is currently indicted or charged with, or has admitted guilt of, or has ever been convicted of, or placed under supervision for, any criminal offense involving actual, attempted, or conspiracy to commit bribery, theft, fraud, forgery, perjury, dishonesty or deceit against an officer or employee of the City or any "sister agency"; and (ii) the Applicant understands and acknowledges that compliance with Article I is a continuing requirement for doing business with the City. NOTE: If MCC Chapter 1-23, Article I applies to the Applicant, that Article's permanent compliance timeframe supersedes 5-year compliance timeframes in this Section V.
- 9. [FOR APPLICANT ONLY] The Applicant and its Affiliated Entities will not use, nor permit their subcontractors to use, any facility listed as having an active exclusion by the U.S. EPA on the federal System for Award Management ("SAM").
- 10. [FOR APPLICANT ONLY] The Applicant will obtain from any contractors/subcontractors hired or to be hired in connection with the Matter certifications equal in form and substance to those in Certifications (2) and (9) above and will not, without the prior written consent of the City, use any such

contractor/subcontractor that does not provide such certifications or that the Applicant has reason to believe has not provided or cannot provide truthful certifications.
11. If the Disclosing Party is unable to certify to any of the above statements in this Part B (Further Certifications), the Disclosing Party must explain below:
If the letters "NA," the word "None," or no response appears on the lines above, it will be conclusively presumed that the Disclosing Party certified to the above statements.
12. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all current employees of the Disclosing Party who were, at any time during the 12-month period preceding the date of this EDS, an employee, or elected or appointed official, of the City of Chicago (if none, indicate with "N/A" or "none"). None
13. To the best of the Disclosing Party's knowledge after reasonable inquiry, the following is a complete list of all gifts that the Disclosing Party has given or caused to be given, at any time during the 12-month period preceding the execution date of this EDS, to an employee, or elected or appointed official, of the City of Chicago. For purposes of this statement, a "gift" does not include: (i) anything made generally available to City employees or to the general public, or (ii) food or drink provided in the course of official City business and having a retail value of less than \$25 per recipient, or (iii) a political contribution otherwise duly reported as required by law (if none, indicate with "N/A" or "none"). As to any gift listed below, please also list the name of the City recipient.
None
C. CEDEUTYCA TYON OF STATES AS FINANCIAL DISTRICTION
C. CERTIFICATION OF STATUS AS FINANCIAL INSTITUTION
 The Disclosing Party certifies that the Disclosing Party (check one) is [x] is not
a "financial institution" as defined in MCC Section 2-32-455(b).
2. If the Disclosing Party IS a financial institution, then the Disclosing Party pledges:
"We are not and will not become a predatory lender as defined in MCC Chapter 2-32. We further pledge that none of our affiliates is, and none of them will become, a predatory lender as defined in MCC Chapter 2-32. We understand that becoming a predatory lender or becoming an affiliate of a predatory lender may result in the loss of the privilege of doing business with the City."

MCC Section 2-32	2-455(b)) is a predatory lender with onal pages if necessary):	because it or any of its affiliates (as defined in in the meaning of MCC Chapter 2-32, explain
	" the word "None," or no response a med that the Disclosing Party certi	appears on the lines above, it will be fied to the above statements.
D. CERTIFICAT	ION REGARDING FINANCIAL I	NTEREST IN CITY BUSINESS
Any words or term	ns defined in MCC Chapter 2-156 h	ave the same meanings if used in this Part D.
after reasonable in		he best of the Disclosing Party's knowledge se of the City have a financial interest in his or entity in the Matter?
[] Yes	[x] No	
	ecked "Yes" to Item $D(1)$, proceed Items $D(2)$ and $D(3)$ and proceed to	to Items D(2) and D(3). If you checked "No" o Part E.
official or employed other person or ent taxes or assessment "City Property Sal-	ee shall have a financial interest in latity in the purchase of any property its, or (iii) is sold by virtue of legal	idding, or otherwise permitted, no City elected his or her own name or in the name of any that (i) belongs to the City, or (ii) is sold for process at the suit of the City (collectively, ten pursuant to the City's eminent domain he meaning of this Part D.
Does the Matter in	volve a City Property Sale?	
[] Yes	[] No	
		mes and business addresses of the City officials fy the nature of the financial interest:
Name	Business Address	Nature of Financial Interest
	Party further certifies that no proh ty official or employee.	ibited financial interest in the Matter will be

Ver.2018-1 Page 8 of 15

E. CERTIFICATION REGARDING SLAVERY ERA BUSINESS

(If no explanation appears or begins on the lines above, or if the letters "NA" or if the word "None" appear, it will be conclusively presumed that the Disclosing Party means that NO persons or entities registered under the Lobbying Disclosure Act of 1995, as amended, have made lobbying contacts on behalf of the Disclosing Party with respect to the Matter.)

2. The Disclosing Party has not spent and will not expend any federally appropriated funds to pay any person or entity listed in paragraph A(1) above for his or her lobbying activities or to pay any person or entity to influence or attempt to influence an officer or employee of any agency, as defined by applicable federal law, a member of Congress, an officer or employee of Congress, or an employee Ver.2018-1

Page 9 of 15

of a member of Congress, in connection with the award of any federally funded contract, making any federally funded grant or loan, entering into any cooperative agreement, or to extend, continue, renew, amend, or modify any federally funded contract, grant, loan, or cooperative agreement.

- 3. The Disclosing Party will submit an updated certification at the end of each calendar quarter in which there occurs any event that materially affects the accuracy of the statements and information set forth in paragraphs A(1) and A(2) above.
- 4. The Disclosing Party certifies that either: (i) it is not an organization described in section 501(c)(4) of the Internal Revenue Code of 1986; or (ii) it is an organization described in section 501(c)(4) of the Internal Revenue Code of 1986 but has not engaged and will not engage in "Lobbying Activities," as that term is defined in the Lobbying Disclosure Act of 1995, as amended.
- 5. If the Disclosing Party is the Applicant, the Disclosing Party must obtain certifications equal in form and substance to paragraphs A(1) through A(4) above from all subcontractors before it awards any subcontract and the Disclosing Party must maintain all such subcontractors' certifications for the duration of the Matter and must make such certifications promptly available to the City upon request.

B. CERTIFICATION REGARDING EQUAL EMPLOYMENT OPPORTUNITY

If the Matter is federally funded, federal regulations require the Applicant and all proposed subcontractors to submit the following information with their bids or in writing at the outset of negotiations.

Is the Disclosing Party the	Applicant?
[] Yes	[x] No
If "Yes," answer the three of	questions below:
Have you developed an federal regulations? (See 4	,
-	
· •	n any previous contracts or subcontracts subject to the
If you checked "No" to que	stion (1) or (2) above, please provide an explanation:

SECTION VII -- FURTHER ACKNOWLEDGMENTS AND CERTIFICATION

The Disclosing Party understands and agrees that:

- A. The certifications, disclosures, and acknowledgments contained in this EDS will become part of any contract or other agreement between the Applicant and the City in connection with the Matter, whether procurement, City assistance, or other City action, and are material inducements to the City's execution of any contract or taking other action with respect to the Matter. The Disclosing Party understands that it must comply with all statutes, ordinances, and regulations on which this EDS is based.
- B. The City's Governmental Ethics Ordinance, MCC Chapter 2-156, imposes certain duties and obligations on persons or entities seeking City contracts, work, business, or transactions. The full text of this ordinance and a training program is available on line at www.cityofchicago.org/Ethics, and may also be obtained from the City's Board of Ethics, 740 N. Sedgwick St., Suite 500, Chicago, IL 60610, (312) 744-9660. The Disclosing Party must comply fully with this ordinance.
- C. If the City determines that any information provided in this EDS is false, incomplete or inaccurate, any contract or other agreement in connection with which it is submitted may be rescinded or be void or voidable, and the City may pursue any remedies under the contract or agreement (if not rescinded or void), at law, or in equity, including terminating the Disclosing Party's participation in the Matter and/or declining to allow the Disclosing Party to participate in other City transactions. Remedies at law for a false statement of material fact may include incarceration and an award to the City of treble damages.
- D. It is the City's policy to make this document available to the public on its Internet site and/or upon request. Some or all of the information provided in, and appended to, this EDS may be made publicly available on the Internet, in response to a Freedom of Information Act request, or otherwise. By completing and signing this EDS, the Disclosing Party waives and releases any possible rights or claims which it may have against the City in connection with the public release of information contained in this EDS and also authorizes the City to verify the accuracy of any information submitted in this EDS.
- E. The information provided in this EDS must be kept current. In the event of changes, the Disclosing Party must supplement this EDS up to the time the City takes action on the Matter. If the Matter is a contract being handled by the City's Department of Procurement Services, the Disclosing Party must update this EDS as the contract requires. NOTE: With respect to Matters subject to MCC Chapter 1-23, Article I (imposing PERMANENT INELIGIBILITY for certain specified offenses), the information provided herein regarding eligibility must be kept current for a longer period, as required by MCC Chapter 1-23 and Section 2-154-020.

CERTIFICATION

Under penalty of perjury, the person signing below: (1) warrants that he/she is authorized to execute this EDS, and all applicable Appendices, on behalf of the Disclosing Party, and (2) warrants that all certifications and statements contained in this EDS, and all applicable Appendices, are true, accurate and complete as of the date furnished to the City.

Doublewide LLC
(Print or type exact legal name of Disclosing Party)
By: Kt Way (Sign here)
Kyle Wolff
(Print or type name of person signing)
Sole Member
(Print or type title of person signing)
Signed and sworn to before me on (date) 6 16 2021.
at ESEX County. MA (state).
<u>URManos</u>
Nordry Public *
THE PART OF THE PA
Commission expires: 5 5 2028

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX A

FAMILIAL RELATIONSHIPS WITH ELECTED CITY OFFICIALS AND DEPARTMENT HEADS

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5%. It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

Under MCC Section 2-154-015, the Disclosing Party must disclose whether such Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently has a "familial relationship" with any elected city official or department head. A "familial relationship" exists if, as of the date this EDS is signed, the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof is related to the mayor, any alderman, the city clerk, the city treasurer or any city department head as spouse or domestic partner or as any of the following, whether by blood or adoption: parent, child, brother or sister, aunt or uncle, niece or nephew, grandparent, grandchild, father-in-law, mother-in-law, son-in-law, daughter-in-law, stepfather or stepmother, stepson or stepdaughter, stepbrother or stepsister or half-brother or half-sister.

"Applicable Party" means (1) all executive officers of the Disclosing Party listed in Section II.B.1.a., if the Disclosing Party is a corporation; all partners of the Disclosing Party, if the Disclosing Party is a general partnership; all general partners and limited partners of the Disclosing Party, if the Disclosing Party is a limited partnership; all managers, managing members and members of the Disclosing Party, if the Disclosing Party is a limited liability company; (2) all principal officers of the Disclosing Party; and (3) any person having more than a 7.5% ownership interest in the Disclosing Party. "Principal officers" means the president, chief operating officer, executive director, chief financial officer, treasurer or secretary of a legal entity or any person exercising similar authority.

Does the Disclosing Party or any "Applicable Party" or any Spouse or Domestic Partner thereof currently have a "familial relationship" with an elected city official or department head?

[] Yes	[x] No				
which such person	dentify below (1) the natis connected; (3) the nations has a familial relations	ame and title of	the elected city of	fficial or depa	artment head to
E	**************************************				
					age te
	***************************************	and the second section of the second section of the second section sec		······	

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX B

BUILDING CODE SCOFFLAW/PROBLEM LANDLORD CERTIFICATION

This Appendix is to be completed only by (a) the Applicant, and (b) any legal entity which has a direct ownership interest in the Applicant exceeding 7.5% (an "Owner"). It is not to be completed by any legal entity which has only an indirect ownership interest in the Applicant.

[] Yes				10, is the Applicant or any Owner identified as a building code to MCC Section 2-92-416?
the Applicant identified as a building code scofflaw or problem landlord pursuant to MCC Section 2-92-416? [] Yes [] No [x] The Applicant is not publicly traded on any exchange. 3. If yes to (1) or (2) above, please identify below the name of each person or legal entity identified as a building code scofflaw or problem landlord and the address of each building or buildings to which		[] Yes	[x] No	
3. If yes to (1) or (2) above, please identify below the name of each person or legal entity identified as a building code scofflaw or problem landlord and the address of each building or buildings to which	the	e Applicant identific		·
as a building code scofflaw or problem landlord and the address of each building or buildings to which		[] Yes	[] No	[x] The Applicant is not publicly traded on any exchange.
	as	a building code sco	fflaw or problen	i i

CITY OF CHICAGO ECONOMIC DISCLOSURE STATEMENT AND AFFIDAVIT APPENDIX C

PROHIBITION ON WAGE & SALARY HISTORY SCREENING - CERTIFICATION

This Appendix is to be completed only by an Applicant that is completing this EDS as a "contractor" as defined in MCC Section 2-92-385. That section, which should be consulted (www.amlegal.com), generally covers a party to any agreement pursuant to which they: (i) receive City of Chicago funds in consideration for services, work or goods provided (including for legal or other professional services), or (ii) pay the City money for a license, grant or concession allowing them to conduct a business on City premises.

On behalf of an Applicant that is a contractor pursuant to MCC Section 2-92-385, I hereby certify that the Applicant is in compliance with MCC Section 2-92-385(b)(1) and (2), which prohibit: (i) screening job applicants based on their wage or salary history, or (ii) seeking job applicants' wage or salary history from current or former employers. I also certify that the Applicant has adopted a policy that includes those prohibitions.

[] Yes
[] No
[x] N/A – I am not an Applicant that is a "contractor" as defined in MCC Section 2-92-385.
This certification shall serve as the affidavit required by MCC Section 2-92-385(c)(1).
If you checked "no" to the above, please explain.